EUROSTAT/C2/80031

EUROSTAT

Luxemburg, July 1980

Statistical Office of the European Communities Demographic and Social Statistics Wages and Incomes

> The structure of earnings in wholesale distribution, retail distribution, banking and insurance in 1974

> > A country by country synthesis

Study by: Dr.W.M. Lister Civil Service College London for: Statistical Office of the European Communities

Luxemburg, July 1980

Statistical Office of the European Communities Demographic and Social Statistics Wages and Incomes

> The structure of earnings in wholesale distribution, retail distribution, banking and insurance in 1974

> > A country by country synthesis

Study by: Dr.W.M. Lister Civil Service College London for: Statistical Office of the European Communities

The structure of earnings in wholesale distribution, retail distribution, banking and insurance in 1974

A country by country synthesis

CONTENTS

CHAPTER	1	:	GENERAL INTRODUCTION AND SOME OVERALL FEATURES	1
CHAPTER	2	:	WHOLESALE DISTRIBUTION	19
CHAPTER	3	:	RETAIL DISTRIBUTION	86
CHAPTER	4	:	BANKING	151
CHAPTER	5	:	INSURANCE	215
CHAPTER	б	:	SUMMARY AND CONCLUSIONS	271
CHAPTER	7	4.6 4	TECHNICAL ANNEX	278

CHAPTER 1: GENERAL INTRODUCTION AND SOME OVERALL FEATURES

1.1. The Eackground to the Report

In 1974 the Statistical Office of the European Communities (SOEC) carried out a survey into the Structure of Earnings in Wholesale and Retail Distribution, Banking and Insurance in the nine member countries of the European Community. The survey was carried out in all countries on the basis of similar methods and definitions and the results processed according to a standard plan drawn up by SOEC. A volume was produced for each country (Volumes 2-10) together with an introductory volume (Volume 1) giving the methodological characteristics and objectives of the survey, together with the plan for analysing the data.

This report gives a country-by-country synthesis of the main features of the data collected, together with some commentary and additional analysis.

For brevity the details of the conduct of the survey and methods used, which are given in Volume 1 as indicated above, will not be repeated in this report. However, it is worth noting that the scope and objectives of the survey were defined as follows:-

a. "the survey covers all undertakings employing 10 or more employees, in the sectors of wholesale and retail distribution, banking and insurance in the nine countries of the Community: the Federal Republic of Germany, France, Italy, the Netherlands, Belgium, Luxembourg, the United Kingdom, Ireland and Denmark.

b. it covers the earnings and individual characteristics of every category of employees in such undertakings ie manual and non-manual workers, supervisory and executive staff". Also:- "The returns related, for the most part, to the statistical survey unit, ie the employee, which means any person bound to the returning unit by a contract of employment on 31 October 1974 excluding articled apprentices, family employees, outworkers and in general all persons without a contract of employment."

The condition at a. means that the coverage of the sectors will be different from one country to another and may have very noticeable effects particularly in thé wholesale and retail distribution sectors. The interpretation of the conditions under b. in the separate countries may have been different. The effects of these conditions on the results will be discussed at a later stage.

1.2. The Structure of the Report

The structure of this report is as follows. It contains the following chapters:-

- 1. General Introduction and Some Overall Features
- 2. Wholesale Distribution

3. Retail Distribution

4. Banking

5. Insurance

6. Summary and Conclusions

7. Technical Annex

The purpose of the survey was to enable comparisons of the structure of earnings in Wholesale Distribution, Retail Distribution, Banking and Insurance to be made throughout the member states of the Europeen Communities. To facilitate the interpretation of the earnings data it is however useful to consider some of the general features of the labour force and some of these (eg number and percentage of part-time employees) are given in this introductory chapter.

It should be emphasised also that the purpose of the survey was to compare the structure of earnings, and not the level of earnings, and that in the subsequent chapters on the sectors, earnings figures will always be given relative to the overall averages of the group concerned (eg all male employees, all female employees etc). It is of interest, however, to have for reference purposes only, some of the absolute figures upon which these index rigures have been based. This general introductory chapter therefore contains tables of the

mean gross monthly earnings and mean annual earnings for the broadest groupings of employees (full-time male employees, full-time female employees, - part-time male employees, etc).

After this introductory chapter, there follow four chapters, each of which deals with one sector in the order given above.

Each chapter, in turn, follows a pattern. The first three sections deal with some of the overall features.

Section 1 gives the numbers of employees (part-time plus full-time together) and also these categories given separately; full-time female employees; and part-time female employees. Next there follows a section describing the composition of the sector (apart from the Insurance sector which has no subdivisions) and section 3 gives the distribution of all employees by the size of undertaking in which they are employed. Each section is accompanied by a table of figures with the reference to the source table in the country volumes given below it (where appropriate).

Section 4 deals with the graphs of the distribution of earnings of full-time male and female employees about the overall mean values. In the chapter on Wholesale

Distribution, an indication is given of the methodology used - and more details are given later in the Technical Annex - but this is not repeated in subsequent charters where only an outline and references are given.

Section 5 in each chapter considers the various indices of mean gross monthly earnings by the following factors of classification

5.1	sex
5.2	age
5.3	length of service
5.4	occupational qualification group
5.5	size of undertaking

Tables are given using the conventions regarding source tables, as in previous sections. The notation for unattributed values, cells rendered empty for reasons of confidentiality etc, is that given on page L at the beginning of this report and follows that used in the volumes for each country. Where appropriate, each subsection contains a table of the frequency distribution of male and female employees separately for each factor of classification, and tables of indices are given comparing groups of employees of the same sex and also male and female employees in the same groups.

Section 6 considers the indices of mean gross annual earnings by occupational qualification group both for employees of the same sex and between male and female employees in the same occupational qualification group.

Section 7 gives the frequency distribution of employees (male and female separately) by length of scheduled working week and section 8 considers the distribution of employees by percentage bonus payment.

The final section in each chapter (section9) relates to the characteristics and earnings of part-time employees. This section is further subdivided into 5 subsections as follows:-

9.2 length of scheduled Working Week

9.2 age

9.3 length of service

9.4 occupational qualification group and indices of earnings (mean gross monthly and mean gross annual)

9.5 size of undertaking.

Chapter 6 gives a summary of the main findings in report, bringing together in particular the main features of the indices calculated for the various groupings in the four sectors.

Chapter 7, the Technical Annex, contains details of any calculations involved, the methodology of the constructions of graphs for the distribution of earnings, and the definitions used in the various sectors.

1.3 -Some General Features of the Labour Force Covered by the Survey

In Table 1, the estimated population figures for full-time and part-time employees have been presented and in Table 2, these are expressed as percentages of the corresponding "Whole sector" totals.

TABLE	NUMBER	OF EMPLOYEES
		(a'000)

FUIL-TIME AND PART-TIME	GERMANY	FRANCE	ITAL.Y	NETH	BELG	INX	UK	IRELAND	DENMARK
WHOLESALE DISTRIBUTION	700.3	548.2	375.2	225.0	116.4	5.6	537.2	30.5	106.5
RETAIL DISTRIBUTION	799.1	471.8	247.2	201.0	124.6	.3.1	1466.7	26.7	67.0
BANKING	399.9	327.9	353. 3	88.8	65.3	5.4	395.8	15.3	39.9
INSURANCE	136.7	82.8	54.5	28.9	22.0	0.4	184.1	5.2	9.8
TOTAL	2036.0	1430.7	1030.7	543.7	328.3	14.5	2583.8	77.7	223.2

SOURCE TABLE AL COUNTRY VOLUMES

Perhaps the most interesting features arising from Tables 1 and 2 are the size of the Retail Distribution sector in the United Kingdom and the proportionate sizes of the Banking sector in both Italy and Luxembourg.

The Retail Distribution sector in the United Kingdom is almost twice as large as that in Germany which is next largest. It is also interesting to note that in Germany, Belgium and the United Kingdom the Retail Distribution sector is larger than the Wholesale Distribution sector with the position reversed in other countries.

The Banking sectors in Italy and Luxembourg account for 34.3% and 37.2% of the estimated total labour force covered by the survey in these countries. The percentages employed in Retail Distribution in these countries are correspondingly low (24% in Italy, 21.4% in Luxembourg) and these percentages may possibly reflect an effect of restricting from the scope of the survey undertakings with less than 10 employees. (Some additional comments are given on this in the Technical Annex).

In all countries the Insurance sector is consistently the smallest with that in the United Kingdom being largest both in absolute and proportionate terms.

.8

TABLE 2

GE	RMANY	FRANCE	ITALY	NETH	BELG	INX	UK .	IRELAND	DENMA RK
WHOLESALE DISTRIBUTION	34.4	. 38.3	36.4	41.4	35.4	38.6	20#8	39,2	47.7
RETAIL DISTRIBUTION	39.2	33.0	24.0	37.0	38.0	21.4	56.8	34.4	30.0
BANKING : INSURANCE	19.6 6.7	22.9 3.8	34.3 5.3	16.3 5.3	19.9 6.7	37.2 2.8	15.3 7.1	19.7	17.9 4.4

SECTORS AS % OF WHELE SERVICES SECTOR (FULL-TIME AND PART-TIME)

SOURCE TABLE AS COUNTRY VOLUMES

1.4 Female Employees - Full-time plus Part-time

Article 119 of the Treaty of Rome states that men and women should receive equal pay for equal work. This type of structure survey makes it possible - after eliminating the specific influence of the main structural factors on the overall averages - to pinpoint the disparities which still persist in this respect in the various countries. This is of particular interest in Wholesale and Retail Distribution, Banking and Insurance in which women represent a particularly large proportion of the total labour force. Table 3 gives the figures on this, where each entry is the percentage of female employees in each sector and in each country.

TABLE 3 I	FEMALE	EMPLOYEES	AS	A	%	ÛF	ALL	EMPLOYEES	IN	EACH	SECTO

					• • • • • • •			•	
	GERMANY	FRANCE	ITALY	NETHS.	BELG.	LUX.	UK	IRELAND	DENMARK
WHOLESALE	35.5	30.6	32.5	20.0	29.6	30.4	31.3	23.0	26.0
RETAIL	69.6	58.1	50.0	55.6	61.3	71.0	62.7	51.3	65,4
BANKING	54.9	48.3	15.1	42.9	32.1	44.5	54.5	50,3	55.4
INSURANCE	54.6	59.8	35.1	33.6	43.2	50.0	49.4	50.0	65,3
TOTAL% (ALI. SECTORS)	54.0	45.3	30.8	37.6	43.0	44.8	54.0	39.9	44.8

SOURCE THELE AS COUNTRY VOLUMES

In interpreting these figures it is very useful to remember the background against which they occur. In the Technical Annex Figure ⁵ which has been extracted from the 1975 Labour Force Sample Survey supplies this information. Each graph shows for women the percentage of the age group who are in employment. A variety of patterns is revealed. These patterns are well known and have been discussed elsewhere (cf for example Social Trends 10, 1980, p123) and we shall return to them when we consider the separate sectors.

Returning to Table 3 we find that, in general, the level of female participation is highest in the Retail Distribution sector for all countries and lowest in Wholesale Distribution with considerable

variations in the percentages observed. Banking and Insurance occupy, in general, an intermediate position though there are several noticeable exceptions. Perhaps the most noticeable is the very low percentage (15.1) of female employees in Banking in Italy and to a lesser extent the percentage (32.1) in Belgium. At the other extreme is Denmark with a figure of 65.4% in Insurance. These figures will be considered in more detail in subsequent chapters.

1.5 Part-time Employees

Another important characteristic of the labour force is the percentage of employees who work part-time. There may have been some slight difficulties in the application of the definition (cf Technical Annex) on a consistent basis over all countries but the figures are given below, expressed as a percentage of all employees in each sector and in each country.

	GERMANY	FRANCE	ITALY	NETH	BELG	LUX	UK	IRELAND	DENMARK
WHOLESALE DISTRIBUTION	15.5	4.4	1.9	7.7	3.4	5.4	12.3	1.3	12.9
RETAIL DISTRIBUTION	33.3	11.2	5.3	33.4	16.9	12.9	27.7	3.7	47.5
BANKING	13.5	2.1	0.6	11.9	2.6	1.8	5.7	- 1	23.1
INSURANCE	11.2	. 1.4	0.4	12.ŀ	2.3	-	10.4	-	23.5
ALL SECTORS	21.8	5.9	2.2	18.1	8.3	5.5	19.9	1.8	. 25.5

Table 4 PART-TIME EMPLOYEES AS A % OF ALL EMPLOYEES IN EACH SECTOR

SOURCE TABLE AL COUNTRY VOLUMES.

As might be expected there are considerable variations in the percentages observed from one sector to another and from one country to another.

Taken over all sectors Ireland and Italy have the smallest percentages of part-time employees (1.8% and 2.2% respectively). Denmark and Germany are at the other end of the scale (25.5% and 21.8% respectively) with Denmark particularly noticeable for its high percentages in Retail Distribution, Banking and Insurance.

Between sectors, Retail Distribution is consistently the highest as regards the percentage of part-time employees in the labour force with Banking and Insurance employing proportionately very few apart from in the countries previously mentioned.

Table5 shows the figures for part-time female employees expressed as a percentage of all employees.

Table 5

PART-TIME FEMALE EMPLOYEES AS A % OF ALL EMPLOYEES IN EACH SECTOR

	GERMANY	FRANCE	ITALY	NETHS	BELG	LUX	UK .	IRELAND	DENMARK
WHOLESALE	13.7	2.9	1.4	6.2	2.5	3.6	9.3	1.0	11.5
RETAIL	31.7	9.0	4.7	25.7	16.0	9.7	25.1	3.0	42.7
BANKING	12.9	1.6	0.5	10.5	2.5	1.9	4.8	-	20.6
INSURANCE	10.7	1.1	0.2	10.4	1.8	-	9.2	-	22.4

SOURCE TABLE AT COUNTRY VOLUMES

A comparison of the figures in Tables 4 and 5 illustrates that in all countries and sectors the majority of part-time employees are female. The figures also confirm the extent of part-time working in Denmark.

1.6 Earnings Data

As has been stated earlier, the purpose of this report was to enable comparisons of the structure of earnings in the various sections to be made. It was not intended that the information collected should be used to compare levels of earnings. In each of the chapters, therefore, the average earnings figures have been converted into index

form so that the mean gross monthly earnings, for example, of one sub group is expressed relative to the mean gross monthly earnings of the whole group. This facilitates comparison of the structure of earnings and avoids the wellknown problem of which rates of exchange, should be used, - and in any case these considerations are outwith the scope of this present survey.

It is, however, useful for workers who may wish to calculate further index figures to have available in summary form the values of the various averages in national currencies upon which these indices have been based. Tables 6, 7, 8, and 9 therefore give respectively the following information, where the averages are given in the respective national currencies:-

Table 6:	Mean Gross	Monthly Earnings) - Full-time Employees
Table 7:	Mean Gross	Monthly Earnings) - Part-time Employees
Table 8:	Mean Gross	Annual Earnings) - Full-time Employees
Table 9:	Mean Gross	Annual Earnings) - Part-time Employees

It should always be remembered, however, that these averages should not be used to compare <u>levels</u> of earnings for it was not for this purpose that they were originally collected.

		·	•		•		•	.•	
HALE	GERMANY DM	FRANCE	ITALY	NETHERLANDS FI	BELGIUM BF	LUXEMBOURG F.LUX	UNITED KINGDOM E	IRELAND E	DENMARK Kr
WHOLESALE	2180	3107	289406	2166	31615	29052	240	197	5957
RETAIL	2032	2816	270696	1843	28308	25423	226	181	5539
Danking	2656	3947	570441	2341	42025	36699	294	863	7185
INSURANCE	2784	3577	508713	2432	37057	40212	298	240	8173
FERMLE	GERHAWY DM	FRANCE	ITÀLY LIT	NETHERLANDS	BELGIUM BF	LUXENBOURG F.LUX	UNITED KINGDOM E	IRELAND E	DEMAAK Kr
WHOLESALE	1523	2068	228718	1324	20561	17064	126	111	4240
RETAIL	1349	1860	232387	1128	20129	14310	125	106	4023
BANKING	1883	2836	453696	1325	29908	22894	158	162	5033
INSURANCE	1997	2417	361655	1469	25350	21878	149	141	5160
1				-					
FERALE &	GERNANY DM	FRANCE FF	ITALY LIT	NETHERLANDS FI	BELGIUM	LUXEMBOURG F.LUX	UNITED KINGDOM	IRELAND £	DENMARK Kr
WHOLESALE	2011	2811	270776	2043	28483	25683	212	178	5669
RETAIL	1644	2300	254080	. 1527	23840	17318	174	143	4877
EARKING	2278	4323	·· 553545	1970	38384	30626	223	213	6202
IRSURANCE	2394	2888	458067	2181	32176	31690	231	192	6480
						-	-		

MEAN GROSS MONTHLY EARNINGS - IN NATIONAL CURRENCIES - FULL-TIME EMPLOYEES TABLE 6:

:

5

SOURCE :

TABLE C1

COUNTRY VOLUMES

JELLE GERMANK FRAME ITALIV REFREAME ITALIV REFREAME ITALIV REFREAME REAME ITALIV REMAIN REM										
Red ESALE B34 1452 198172 · 9766 · - 2996 GANL 776 1056 201977 - 8479 3234 69 - 1960 GANL 776 1056 201975 - 8479 3234 69 - 1960 GANL 776 1277# 415647 - 8479 3234 69 - 1960 GAURANCE - 2076 - 15410 - - 2016 GAURANCE - 2073 NETHERLANDS BELGUM NETHERLANDS BELGUM INTED KINADDM INTED KINADM INTED KINADDM INTED KINADDM INTED KINADDM INTED KINADDM	THE	GERMANY DM	FRANCE	ITALY LIT	NETHERLANDS FI	BELGIUM BF	LUXEMBOURG F.LUX	UNITED KINGDOM £	IRELAND £	denmark Kr
CIRL T76 1058 201975 - 8479 2234 69 * - 1360 CARLA 904 1287# 415647 - 8479 5334 69 * - 916 CARLA 904 1287# 415647 - 817 15410 - - 2016 CARLA 87 LTT NEMERAND BELGIUM NATKROUNG NATKROUNG NATKROUNG NATKROUNG 2933 76 - 2497 CARL 825 831 178405 - 13166 84 - 2318 CARLA 825 831 178405 - 13166 84 - 2497 CARLA 703 371235 - 13166 84 - 2318 CARLA 713 371235 - 13166 84 - 2318 CARLA 713 371235 - 13164 84 - 2316	HOLESALE	834	1452	198172	1	•	9766	•	1	2896
ACTING 904 1287 415647 - · 15410 · - 2016 ACTING 2073 2 - - - - - 2016 ACTING 2073 - - - - - - - 2016 ACTING 2073 - 1781 NETHERANDS BELGUN INTED KINGDOM IRELAND DENARK FEMALE 8851 178405 - 11017 9533 76 - 23437 ACTILL 703 371235 - 12642 8689 63 - 23437 ACTILL 703 371235 - 12642 8689 63 - 23437 ACTING 939 703 371235 - 12642 8689 63 - 23437 ACTING 939 703 371235 - 12642 8689 63 - 23437 ACTILL	ETAIL	776	1058	201975	. 1	8479	32,34	6 6 #		1360
INSTRATE 2073 # - <	CATKING	904	1287 #	415647	1	•	15410	•	1	2016
FEMALE GERMANN ENAME FRAME ITALY FT NETHERLANDS BELGEUM LUXEMBOURG UNITED KINGDOM IERAND IERAND ENAME GERMANN FF LIT T NETHERLANDS BF LUXEMBOURG INTED KINGDOM IERAND IERAND <td>LISURANCE</td> <td></td> <td>2073#</td> <td>•</td> <td>- · · · 1</td> <td>•</td> <td>9</td> <td>•</td> <td>1</td> <td>•</td>	LISURANCE		2073#	•	- · · · 1	•	9	•	1	•
FEMALE GERMANY FRANCE ITALY NETHERLANDS BELGETUM LUTTE LUT PELAUD DEWNARK GERMANY FFC LIT FT NET BF LUT FLUX ELUT PEC Kr GERMALE B25 B24 166703 - 11017 9533 76 - 2497 GENALE B25 B31 178405 - 12642 8689 63 - 2497 GENAL 723 3371235 - 12642 8689 63 - 2298 AMINING 939 703 371235 - 12610 - 74 - 2393 AMINING 939 76 - 74 - 2634 AMINING 890 # 175616 - 74 - 2835 AMINING 74 1366 84 - 2835 524 AMINING 8761 87 8619									•	••
MGLESKLE 82,4 166703 - 11017 9533 76 - 2497 SETALL 729 831 178405 - 12642 8689 63 - 2234 SETALL 729 831 178405 - 12642 8689 63 - 2234 MAILING 939 703 371235 - 13126 11966 84 - 2234 MAILING 939 703 371235 - 13126 144 208876 - 2134 - 2234 Statue 1113 890 # 308876 - 12610 - 74 - 2335 Statue ERMANY FRANC Infa ^T BF Infa ^T Infa ^T BF - 74 - 2335 Statue 74 174 BF BF FLUX BF ERAND ERAND ERAND BF BF BF BF BF	FEMALE	GERMANY DA	FRANCE	ITALY LIT	NETHERLANDS FI	BELGIUM	LUXEMBOURG F.LUX	UNITED KINGDOM E	Í IRELAND É	DENNARK Kr
ETML 729 861 178405 - 12642 8699 63 - 2218 AMMING 939 703 371235 - 13126 11966 84 - 2234 AMMING 939 703 371235 - 13126 11966 84 - 2234 AMMING 939 703 371235 - 12610 - 74 - 2334 AMMING 939 FRAMUE ITAL METHERLANDS BELGIUM LUXEMBOURG INTED KINCDOM IRELAND 2234 FEMMLE Base 1061 1775634 - 11508 9619 81 - 2529 MOLESALE 826 1061 1775634 - 11508 9619 81 - 2529 MOLESALE 832 379242 - 12370 12375 92 - 2136 MILLEMUL 738 0 13370 12375 92 <	HOLESALE	825	824	166703	1	11017	9533	76	1	2497
MAIKING 989 703 371235 - 13126 11966 84 - 2224 INSUBANCE 1113 890 tf 308876 - 12610 - 74 - 2835 INSUBANCE 1113 890 tf 308876 - 12610 - 74 - 2835 INLE & GENMANY FRANCE 11Å NETHERLANDS BELGIUM LUNEMBOURG UNITED KINGDOM IRELAND 2835 INLE & GENMANY FRANCE 1175654 - 11508 9619 81 - 2529 INLE & B36 1061 175654 - 11508 9619 81 - 2758 RETAIL 738' 914 181888 - 13370 12375 92.4f - 2136 SANNUNG 935 832 379242 - 13370 12375 92.4f - 2768 SANNUNG 11000 11814 353948	ETAIL	723	881	178405	1	12642	8689	63	t	2218
INSURANCE 1113 890 # 308876 - 12610 - 74 - 2835 INSURANCE 1113 890 # 308876 - 12610 - 74 - 2835 INSURANCE Inste & Demonstrate Inste & Inste & Inste & - 2835 INSTE & Demonstrate Def Inste & Inste & Inste & - 21569 1061 Inste & - 2759 MOLESALE 826 1061 177654 - 11508 9619 81 - 2759 METAIL 738' 914 181888 - 12472 8629 63 - 2136 SETAIL 738' 914 181888 - 12472 8629 63 - 2136 SETAIL 738' 832 3779242 - 13775 92 # - 2136 SANNUGS 1100 1181 # 353948 - 12703 - 216 2202 SOURGE: 1100 1181 #	SANKING	989	703	371235	1	13126	11966	84	ł	2224
IMALE & GERMANY FRANCE ITALY NETHERLANDS BELGIUM LUNEMBOURG IRELAND DENMARK FEMALE DM FRANCE ITALY NETHERLANDS BELGIUM LUXEMBOURG UNITED KINGDOM IRELAND DENMARK FEMALE DM FFAIL TS BELGIUM LUXEMBOURG WITED KINGDOM IRELAND DENMARK FEMALE B26 1061 175654 - 11508 9619 81 - 2529 MOLESALE 826 1061 175654 - 11508 9619 81 - 2739 MOLESALE 932 379242 - 13370 12375 92 - 2136 MINURAANCE 1100 1181 353948 - 12703 - 81 - 2768 SOURCE: INDO INDURE </td <td>TISURANCE</td> <td>1113</td> <td>890 #</td> <td>308876</td> <td></td> <td>12610</td> <td></td> <td>74</td> <td>8</td> <td>2835</td>	TISURANCE	1113	890 #	308876		12610		74	8	2835
ITALE & GERMANY FRANCE ITAL NETHERLANDS BELGIUM LJXEMBOURG UNITED KINGDOM IRELAND DENMARK FEMALE DM FF LIT FI BF LJXEMBOURG UNITED KINGDOM IRELAND DENMARK FEMALE DM FF LIT FI FI BF ELGUM LJXEMBOURG UNITED KINGDOM IRELAND DENMARK MOLESALE 826 1061 175654 - 11508 9619 B1 - 2529 MOLESALE 826 914 181888 - 12472 8629 63 - 2136 ARAILIG 935 832 379242 - 13370 12375 92 - 2136 MAKUIGE 1100 1181 353948 - 12703 - 81 - 2768 MAKUIGE 1100 1181 353948 - 12703 - 81 - 2768 MAKUIGE	16									
MOLESALE 826 1061 175654 - 11508 9619 81 - 2529 RETAIL 738' 914 181888 - 12472 8629 63 - 2136 SETAIL 738' 914 181888 - 12472 8629 63 - 2136 SANKING 935 832 379242 - 13370 12375 92 # - 2202 SANKING 935 832 379248 - 12703 12375 92 # - 2768 INSURANCE 1100 1181 # 353948 - 12703 - 81 # - 2768 SOURCE: 1100 1181 # 353948 - 12703 - 81 # - 2768 SOURCE: 1100 1181 # 353948 - 12703 - 81 # - 2768 SOURCE: 1100 1181 # 353948 - 12703 - 81 # - 2768	TALE & FENALE	GERMANY DM	FRANCE	LITALY	NETHERLANDS FI	BELGIUM	LUXEMBOURG F.LUX	UNITED KINGDOM	IRELAND E	DENMARK Kr
RETAIL 738' 914 181888 - 12472 8629 63 - 2136 SANKTHS 935 832 379242 - 13370 12375 92 th - 2202 SANKTHS 935 832 379242 - 13370 12375 92 th - 2202 INSURANCE 1100 1181 th 353948 - 12703 - 81 th - 2768 INSURANCE 1100 1181 th 353948 - 12703 - 81 th - 2768 SOURCE: TABLE G8 - 12703 - COUNTRY VOLUMES - 2768	HOLESALE	826	1061	175654	8	11508	9619	81		. 2529
DAXUNG 935 832 379242 - 13370 12375 92 # - 2202 INSURANCE 1100 1181 # 353948 - 12703 - 81 # - 2768 INSURANCE 1100 1181 # 353948 - 12703 - 81 # - 2768 SOURCE: TABLE G8 COUNTRY VOLUMES COUNTRY VOLUMES COUNTRY VOLUMES - - 2768	RETAIL	738	914	181888	8	12472	8629	63		2136
INSURANCE 1100 1181# 353948 - 12703 - 81# - 2768 SOURCE: TABLE C8 COUNTRY VOLUMES	BANKING	935	. 832	379242	1	13370	12375	92 #	8	2202
SOURCE: TABLE CB COUNTRY VOLUMES	INSURANCE	1100	1181#	353948	1	12703	1	# 18	1	2768
	SOURCE			. TABLE C	÷ õ		COUNTRY	VOLUMES		

:

. *

TABLE &	: MEAN (FROSS ANNUA	L EARNINGS -	IN NATIONAL	CURRENCI	ES - FULL-TI	ME EMPLOYEES		
HALE	GERMANY DM	FRANCE	ITALY	NETHERLANDS	BELGIUM BF	LUXENBOURG F.LUX	UNITED KINGDOM E	IRELAND	DENMARK Kr
SIDLESALE	25969	38056	3490930	26203	391255	346150	2637	2408	70307
RETALL	24436	34074	3159090	22398	342206	295499	2443	2205	62779.
BARKING	31635	46480	7058521	28238	507535	434148	· 3341	3350	82153
INSURANCE	85958	43195	6263083	29115	440263	485423	3418	2995	94475
								•.	
FEMALE	GERHANY DH	FRANCE	LIT LITALY	NETHERLANDS FI	BELGIUM BF	LUXEMBOURG F.LUX	UNITED KINGDOM	IRELAND	DENHARK Kr
HOLESALE	18179	24447	2642736	16136	252978	204025	1353	1367	48670
RETAIL .	16004	21110	2532693	13615	232699	165496	1332	1291	46552
BANKING	22259	32582	5498627	15876	356307	271855	1784	2021	55877
INSURANCE	23249	27499	4407099	17767	298438	251532	1656	1742	57478
17									
HALE &	GERMANY DM	FRANCE	ITALY LIT	NETHERLANDS FI	BELGIUM BF	LUXEMBOURG F.LUX	UNITED KINGDOM E	IRELAND £	DENMARK Kr
MHOLESALE	24036	34293	3243280	24940	352786	307713	2333	2189	66894
RETAIL	19682	27135	2907127	18782	283214	200784	1904	1760	57667
BATKING	. 27143	08665	. 6848202	24303	464407	365359	2568	2745	70416
INSURANCE	28185	33917	5640369	26459	383582	381120	2696	2420	74042
SOURCE :			TABLE C:	S		COUNTRY	VOLUMES	•	

و.'

ł

•

								•	
14TE	GERMANY DA	FRANCE	ITALY LIT	NETHERLANDS FI	BELGIUM BF	LUXEMBOURG F.LUX	UNITED KINGDOM E	IRELAND	DENMARK Kr
ALLESPLE	10271	20546	2123387	1	•	112674	•	1	32252
RETAL	9448	15790	2509222		92650	94032	821 柱	1	19317
JALAING	10623	•	4739352			•	•	t	26189
INSURANCE	•	27443#	•	-	•	1	•	1	
a Wala	GERMANY DN	FRANCE	ITALY LIT	NETHERLANDS	BELGIUM BF	LUXENBOURG F.LUK	UNITED KINGDOM E	IRELAND E	· DEWNARK Kr
WHOLESALE	9816	9778	1798967	1	132137	103929	832	1	29061
. TIVLIA	9297	10496	2059787	1	135591	97874	695	1	26750
BATKING	11628	8022	4932562	i	154301#	155865	910	I	25870
INSURANCE	13277	11106#	3777646	1	147841	i .	798	1	32208
18									
TALE & FERALE	GERIANY DM	FRANCE	ITALY ¢LIT	NETHERLANDS	BELGIUN BF	LUXENBOURG F.LUX	UNITED KINGDOM	IRELAND £	DENMARK Kr
HOLESALE	9864	13812	. 1891134	8	138232	106900	911 株	8	29320
RETAIL	9303	11421	2162817	1	133736	97378	LOL	1	26232
BAVKING	11584	9381	4896702	8	159128#	160785	1035 #		25899
ItisURATICE	13180	15832	4383997 #	1	146220	1	888 #		31711
SOURCE			TABLE C	0		COUNTRY	NOLUMES		

٠

ļ

I.

CHAPTER 2: WHOLESALE DISTRIBUTION NACE 61

2.1 <u>Some Overall Features</u>

Some important characteristics of the labour force in this sector are given in Table 10 below. These are the number of employees, full-time plus part-time and these categories given separately, the number of full-time female employees and the number of part-time female employees. These are also expressed as percentages of all employees in the sector in eac⁺

MARLE	10	FUELOYEES	00010
TRD1/12	10	ELDETELT EPAS	10UU * K

country.

Independent and the second statement of the second statement of the second statement of the second statement of						· ·			
u Talah kurake sama sa sa sa sa sa sa	CERMANY	FRANCE	IPALY	NETHER- LANDS	DELGIUM	LUXEN- POURG	UK	IRELAND	DEFEMALTS
FULLTIME AND FARTTIMP	700.3	548.2	375.2	225.2	116.4	5.6	537.2	30.5	106.5
FULL-TIME	591.8	524 .2	368.1	207.9	112.4	5.3	471.3	30.1	92.8
PART-TIKE	108.5	24.0	7.1	17.3	4.0	0.3	65.9	0.4	13.7
FULL-TI:ES FEMALE	152.6	151.9	116.6	31.0	31.5	1.5	118.2	6.7	15.4
PART-TIME FEMALE	95.9	14.7	5.4	14.0	2.9	0.2	50.1	0.3	12.3

FULL-FINE AND PART-TIME = 100

CONTRACTOR DATA AND A D	3-TEN ANY REPORT OF LOT OF LOT OF	College and the second second	the second state of the se	مرجع وسيعيدون المتعرف والمطلب أبأ					
	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- EOURG	UK	IRELAND	DENMARK
FULL-TIME AND PART-TIME	100 ·	100	100	100	100	100	100	100	100
FULL-TIME	84.5	95.6	98.1	92.3	96.6	94.6	87.7	<u>9</u> 8.7	87.1
PART-TINE	15.5	4.4	1.9	7.7	3.4	5.4	12.3	1.3	12.9
FULL-TTLE FEMALE	21.8	27.7	31.1	13.8	27.1	26.8	22.0	22.0	14.5
PART-TIME FEMALE	13.7	2.9	1.4	6.2	2.5	3.6	9.3	1.0	11.5
					and the set of the set				

SCURCE

TABLE A:

COUNTRY VOLUMES

The role of female employees is of particular interest and the percentages observed from country to country show interesting variations. Italy has the highest percentage of full-time female employees and apart from Ireland the lowest percentage of female employees who work part-time. France, Belgium and Luxembourg have of the order of 26-28% full-time female employees in the labour force whereas the Netherlands and Denmark have only of the order of 14% in this category. The majority of employees who work part-time are female and the percentages for this category (part-time female employees) vary from figures of the order of 1% (Ireland and Italy) to a figure of the order of 14% for Germany. The other countries fall into two groups with percentages between these two extremes :- France, Belgium, Luxembourg and the Netherlands with percentages of the order of 3-6% and the United Kingdom and Denmark with percentages of the order 9-12%.

2.2 The Composition of the Wholesale Distribution Sector

Throughout this report, results will be given at the sector level - ie Wholesale Distribution, Retail Distribution, Banking and Insurance. It is useful to remember, however, that results are available from the survey at a greater level of disaggregation and where sample size permits, a <u>cimilar</u> analysis may be carried out for the various subsectors. Also, the composition of the sector varies from one country to another and these different compositions may have had some effect on the results which are quoted only at sector level. For example, some subsectors, by the nature of the

the work involved may be more likely to have more female employees or perhaps more part-time employees than others. In Table 11, therefore, are given figures which illustrate the differences in the overall structure of the sector from one country to another

TABLE 11 COMPOSITION OF WHOLESALE SECTOR

(a)

NUMBERS OF EMPLOYEES	(FULL-TIME +	- PART-TIME)) 000 " s
----------------------	--------------	--------------	------------------

NACE CODE	GERMANY	FRANCE	ITALY	NETHS.	BELGIUM	LUX.	υκ	IRELAND	DEITMARIC
611	-:7•9	21.7	48.1	16.1	.2.3	0,1	32.0	6.3	8.7
612	82.5	69.4	45.3	18.0	14.9	. 0.4	26.6	2.9	9.1
613	80.1	• 47•5	35.6	25.4	· 9.3	¹ 0.7	66.4	5.1	20.1
614	120 . Ò	106.8	52.4	54•4	27.3	0.7	40.8	4.4	29.2
615	101.9	85.5	33.1	29.3	13.3	1.6	26.2	1.6	7.5
616	37.1	20.0	34•5	10.8	5.5	Ò.1	27.8	1.2	2.2
617	139•9	148.0	73.5	38.5	21.7	1.5	169.4	6.1	17.3
617.6	6.4	-	· •	-	0.3	-	8.5	0.1	0.4
618	42.6	30.3	31.9	12.4	13.0	0.2	19.0	0.5	4.8
618,1	28.5	. 15.0	21.5	-	7.5	0.2	13.9	0.3	· 1,8
619	48.3	-19.1	20.8	19.9	9.0	0.3	129.0	2.3	7•7
619.1) 619.2)	25.9	5.4	3.7	-	3.0	0.2	22.1	1. ;	3.0
Tetal	700.3	548.2	375-2	225.0	116.4	5.6	537-2	30.5	106.5

	1	١.	
Ł	b	3	

SUB-SECTORS AS % OF WHOLESECTOR WHOLESALE DISTRIBUTION NACE 61

CE CODE	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DENMARK
611 612 613 614 615 616 617 617.6 618 618.1 619 619.1) 619.2	6.8 11.8 11.4 17.1 14.6 5.3 20.0 0.9 6.1 4.1 6.9 3.7	4.0 12.7 19.5 15.6 27.0 - 5.5 2.7 3.5 1.0	12.8 12.1 9.5 14.0 8.8 9.2 19.6 - 8.5 5.7 5.5 1.0	7.2 8.0 11.3 24.2 13.0 4.8 17.1 5.5 - 8.8 -	2.0 12.8 8.0 23.4 11.4 4.7 18.6 0.3 11.2 6.4 7.7 2.6	2.0 7.1 12.5 12.5 28.6 2.0 26.8 - 3.6 3.6 5.4 3.6	6.0 5.0 12.4 7.6 5.2 31.5 5.2 3.5 2.6 24.0 4.1	20.55 14.29-03.50 56	8.2 8.5 18.9 27.4 7.0 2.1 16.2 0.4 4.5 1.7 7.2 2.8

SQURCE TABLE A1 COUNTRY VOLUMES

Some comments on the subsectors are as follows:-<u>Food, Drink and Tobacco (617)</u> is the largest sector in Germany, France, Italy and the United Kingdom. It is the second largest sector in Belgium, Ireland, Luxembourg and the Netherlands and the third largest in Denmark.

<u>Machinery, Equipment and Vehicles (614)</u> is the largest sector in Belgium, the Netherlands and Denmark and is the second largest sector in Cermany, France and Italy.

Furniture, Household Goods and Hardware (615) is the largest sector in Luxembourg. Agriculture, Primary Products and Textiles (611) is the largest sector in Ireland and third largest in Italy.

Other sectors which have not previously been mentioned and which occur in the top three for particular countries are:

<u>Timber and Building Materials (613)</u> which is the second largest in Donmark, third largest in the United Kingdom and Ireland, and equal third largest in Luxembourg.

Other Wholesale Distribution (619) which is the second largest in the United Kingdom.

Fuels, Ores, Metals and Industrial Chemicals (612) which is third largest in Belgium.

In all countries apart from Italy the three largest sectors employ in each case between 51% and 68% of all employees in the wholesale distribution sector.

2.3 The Sector by size of Undertaking

Another important feature which has an influence on the structure of earnings is the size of the undertaking in which the employee works. The percentages of employees (both male and female and full-time and part-time) by size of undertaking are given in Table 12.

In each country the smallest size of undertaking category contains the greatest concentration of employees, but the concentration varies from 27.1% in Ireland to 55 4% in Italy. The United Kingdom has a noticeably higher percentage of employees in the largest size category of undertaking (greater than or equal to 1000 employees). In Ireland, 40.4% of employees are in undertakings with between 100 and 499 employees - the next largest percentage in this category is 30.1% and is for Germany.

Later in this chapter we shall consider the distribution of full-time and part-time employees separately over undertakings of different sizes.

TABLE "IS SIZE OF UNDERTAKIES (FULL-TIME AND PART-TIME EMPLOYEES)

	·		· · · · · · · · · · · · · · · · · · ·				والمراجع والمتاجي		10
SIZE OF UNDERTAKING (EMPLOYEES)	WEST GERMANY	FRANCE	ITALY	NETH	BELG	TNX	UK .	IRELAND	DENMA RK
10-49	36.7	41.0	'58 .4	44.1	47.5	49.3	31.1	27.1	37.6
50-99	18.3	17.0	17.0	18.9	18.3	25.2	14.7	18.5	15.4
100-199	15.6	12.1	` 11 . 2	14.4	13.1	17.9	12.9	22,1	14.4
200-499	14.5	13.3	7.3	12. <u>9</u>	10.6	7.6	13.1	18.3	13.7
500-9 99	6.6	5.4	4.1	•	7.9	- }	8.8	9•7.	7.9
000.0 🌾	8.3	11.1	2.0	.}	5•6	_}	19.4	443	11.1
TOTAL NUMBERS	100.0 700.3	100.0 - 548.2	100.0 375-2	100.0 225.0	100.0 116.4	100.0 5.6	100.0 537.2	100 .0 30 . 5	· 100.0 106.5

0/

SOURCE TABLE AS COUNTRY VOLUMES

••

2.4 Distribution of the labour force according to gross monthly earnings

One of the main aims of the survey was to examine the structure of earnings in the labour force analysed by various characteristics. This section considers the distribution of employees according to gross monthly earnings.

25

۰.

The analysis is limited to full-time employees and to (gross) monthly earnings which has been adjusted to take account of bonuses and premiums not paid on a regular monthly basis but which would be included in annual earnings. This was done to ensure improved comparability between monthly and annual figures. It should also be remembered that certain countries did not record earnings which fell above certain ceiling values determined in each case by the national statistical institute. Some comments are given on these in the Technical Annex and details are given in Volume 1, section 8.14 and footnote, and sections 9.2.5 and 9.2.6.

In each of the volumes for the different countries, a frequency distribution of full-time employces containing 27 class intervals of gross monthly earnings (adjusted as mentioned above) has been given by activity, sex and occupational qualification group . The earnings were expressed in different national currencies and therefore presented a problem in comparing the distributions from one country to another. To convert each distribution into a common currency would have been of limited usefulness, in particular since the purpose of the survey was to to be comparisons of the "structure" of earnings to be made rather than the "level" of earnings from one country to another. In addition, such a conversion would have been subject to the well-known problems of deciding which rates of exchange to use.
This is why, in the present comparison, the distributions have not been considered as a function of the level of gross monthly earnings but rather as a function of the relative deviation of the earnings from the mean gross monthly earnings. This is a procedure which has been adopted in the analysis of similar data in previous structure surveys (cf of the special series 'Structure of earnings in industry' of 1966 and 1972). To gain a better appreciation of the comparison between the gross monthly earnings of male and female full-time employees the distributions have been also expressed in terms of deviations from the overall mean gross monthly earnings (for male and female full-time employees combined). The frequency distributions have been adjusted to take account of the new class interval sizes produced (which varied from country to country). The frequencies have also been scaled so that the area between each curve and the x-axis is proportional to the number of full-time employees, male or female, as appropriate. Z AN DUTLINE OF THE FORMULAE USED IS GIVEN ON PAGE 28 J \sim . NB For the purposes of this report the values corresponding to the open ended first and last intervals have not been plotted7

The resulting graphs are presented in Figure 1. The graphs illustrate very clearly the differing percentages of full-time female employees in the full-time labour force, which are given below:

<u>Graphs</u> These have been constructed from the frequency distribution: of gross monthly earnings for full-time male and female employees in each country.

Relative Deviation, $u = (x-\bar{x})/\bar{x}$

where x denotes earnings in the national currency $\bar{x}_{-=}$ mean gross monthly earnings (male and female employees combined) $\Delta x=$ class interval in x

 $\Delta u = \Delta x / g$

Scaled Frequency, $Y = f/_{\Delta u}$ where f is the percentage frequency. The values of Y have been scaled such that the area between the graph and the horizontal axis in each case is proportional to the number of full-time male/female employees.

 FEMALE
 MALE



29

RELATIVE DEVIATION

.

WHOLESALE GERMANY

•



. ____







Table 13

Germany

Full-time female employees as a % of all full-time employees:

25.8

29.0 France Italy 31.7 Netherlands 14.9 28.0 Belgium 28.3 Luxembourg United Kingdom 25.1 Ireland 22.3 Denmark 16.6 Source: Table Al Country Volumes In particular the figures for the Netherlands and Denmark are much lower than elsewhere.

It is also noticeable that the earnings of female employees are in all countries in the lower part of the earnings distribution with the highest earnings (more than 20% above the overall mean) being almost the exclusive preserve of male employees.

In general the graphs are unimodal (male employees in Inxembourg being the exception) and are postively skewed with the variation in the earnings of male employees being greater than that for female employees, as is shown in Table 14 below which gives the relevant coefficients of variation.

Table 14 Coefficients of Variation (%)

Gross Monthly Earnings of Full-time male and female employees

	Male Employees	Female Employees
Germany	37.8	36.3
France	•	51.4
Italy	44.0	32.9
Netherlands	54.2	41.1
Belgium	51.0	38.5
Inxembourg	45.7	44.8
United Kingdom	66.1	30.2
Ireland	50.8	38,3
Denmark	42.2	25,2

SOURCE TABLE C1 COUNTRY VOLUMES

2.5 <u>Gross Monthly Earnings as a function of the structural characteristics of the labour force - Full-time Employees</u> This section of the report considers the effects which the various characteristics of the labour force have had on the distribution of gross monthly earnings around the mean value in each case. The characteristics to be considered are as follows:-

a. sex

b. age

c. length of service

d. occupational qualification group

e. size of undertaking

35

As has been outlined in Volume 1, the information on earnings was collected and presented in units of the national currency of the various member countries. The results of this survey are intended to be used to facilitate comparisons of 'structure' of earnings rather than 'levels' of earnings and to make these comparisons easier the absolute values have been converted into indices. These indices are then directly comparable not only as been economic activities but also as between countries. Most of the results for any given group are therefore presented in the form of indices derived from the general mean value calculated for the group, and it is with these indices and the frequency distributions of employees that this section will be primarily concerned.

The analysis has been limited to what may be derived from the data which has been published in the volumes for each country and in fact further analysis (eg the fitting of regression lines) is also possible but has not been attempted in this report. In addition the raw data would seem to be amenable to a more sophisticated analysis using multivariate techniques.

2.5.1 Employees classified by Sex

The distribution and graphs shown in the previous section illustrated the position of the earnings of full-time male and female employees. Many factors will have to be considered to explain the differences eg age, length of service, occupational qualification group, size of undertaking, length of scheduled working week and system of payments - and this will be done in this section and subsequent ones. The global picture however is of interest and is shown below in Table 15.

Table 15

Summary of Female/Male Mean Gross Monthly Earnings. (Full-time employees, earnings of full-time male employees = 100)

Germany	69.8
France	66.6
Italy	.79.0
Netherlands	61.1
Belgium	65.0
Inxembourg	58.7
United Kingdom	52.3
Ireland	56.4
Denmark	71.2

SOURCE TABLE C2 COUNTRY VOLUMES

What is very noticeable from these figures is the position of Ttaly where the difference is smallest, and at the other extreme the position of the United Kingdom, Ireland and Luxembourg where the differences are much greater.

2.5.2 Employees classified by age

The average age in years is given below for full-time male and female employees:-.

Male Employees

Female Employees

Germany	36.0	30.8
France	35.5	30.8
Italy	35.4	28.C
Netherlands	34.5	23.9
Belgium	35.2	28.7
Luxembourg	33.4	24.5
United Kingdom	36.6	32.2
Ireland	34.7	25.6
Denmark	38.2	31.8

SOURCE TABLE A6 COUNTRY VOLUMES

It can be seen that whereas the average age for male employees is very similar in all countries the values for female employees vary considerably with the averages for the Netherlands, and to a slightly lesser extent those for Luxembourg and Ireland noticeably lower than elsewhere. At the other end of the scale the averages for Denmark and the United Kingdom for both male and female employees are consistently higher than elsewhere. The frequency distributions which are shown in Table 16 below serve to confirm these different age patterns and it is useful to remember that these reflect the underlying activity patterns which were discussed in Chapter 1.

TABLE 16: AGE DISTRIBUTION FULL-TIME EMPLOYEES

			147						
GE (YRS)	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DENMARK
<:21 21-29 30-44 45-54 ≥55	3.8 23.3 43.8 18.2 10.8	5.4 27.9 35.3 20.1 10.1	4.4 25.4 41.6 21.4 7.2	6.5 28.2 38.8 16.7 9.7	6.6 26.8 36.5 20.8 9.4	13.0 26.0 35.2 16.6 9.3	10.7 22.6 30.6 20.1 16.0	10.2. 26.5 33.0 17.6 12.0	2.3 23.4 37.4 20.4 16.6
moguesed								02	

L CULIN LICI

MALE

(%)___

(4)

AGE (YRS)	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DENHARK
21 21-29 30-44 45-54 	17.4 34.7 24.9 16.0 7.0	15.6 37.0 25.2 14.1; 7.0	17.0 43.3 26.5 11.3 1.8	36.4 42.6 11.8 6.0 3.2	18.4 38.9 27.4 12.2 3.2	40.3 32.7 16.1 8.3 2.7	22.8 25.4 22.6 19.9 9.2	31.480-29	8.6 42.2 25.2 15.6 8.3
UNDEX-LORED							I	11	

SOURCE TABLE AS COUNTRY VOLUMES

In particular the concentration of female employees into the youngest age group in Netherlands, Luxembourg and to a lesser extent in Ireland is most noticeable. Another interesting feature, again concerned with female employees is the contribution of women in the older age group to the labour force particularly in the United Kingdom and to a lesser extent in Denmark, Germany and France. Italy and Belgium hold a middle position in respect of both younger and older age groups.

For male employees the differences are less pronounced apart from the slightly higher percentages of older employees in the United Kingdom and Denmark.

Indices of Mean Gross Monthly Earnings - Employees classified by Age and Sex

Two sets of indices are given. Table 17 gives the "within sex" comparisons with the figures for male and female employees combined also given. Table 19 gives the "between sex" comparison with the mean gross monthly pay of male employees in the same age group taken equal to 100.

For male employees taken alone the index figures rise from values in the range 48-65 (Netherlands 48.8, Italy 64.5) to a peak in the range 108-119 (Germany 108.4, Luxembourg 118.3) occurring in the age group 45-54 years for all countries apart from Germany, Ireland and the United Kingdom where it occurs in the 30-44 years age old. In the oldest age group, 55 years of age and older the index takes values in the range 96-109 (United Kingdom 96.5, Italy 108.1).

For female employees alone the values of the indices tend to be higher than those for the corresponding age groups for males. This reflects the different age distribution patterns for male and female employees, with proportionately more women being concentrated in the younger age groups than men. In the under 21 years of age category the

TABLE 17 INDEX OF JEAN GROSS MONTHLY EARNINGS - BY AGE

-

MAE

AGE IN YEARS	GERMARY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	IJK	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54 > 55 (*)	56.7 86.8 108.4 106.8 96.7 75.0#	51.9 79.7 110.9 115.0 108.0 95.5	64.5 86.2 105.2 109.7 108.1	48.8 82.6 111.6 117.4 107.4 95.0 #	54.5 83.3 110.7 111.6 104.3	50.5 90.7 113.9 118.3 103.6	56.0 87.5 115.8 114.5 96.5	51.2 90.1 113.9 112.2 104.5 (4.2.0*	58.2 85.3 107.6 108.2 98.6
ALL	100	100.0	100	100.0	100.0	100.0	100.0	100.0	100

FEMALE

	AGE IN YEARS	CERMANY	FRANCE	ITALY	NETTER- LANDS	BELGUIM	LUXEM- ECURG	UK	IRELAND	DERMARK
And a second sec	< 21 21 - 29 30 - 44 45 - 54 > 55 (*)	70.6 100.2 114.4 106.7 103.6	72.6 92.4 112.6 114.2 119.3 101.5	83.3 96.9 108.7 114.5 108.2	70.4 107.7 133.6 138.5 137.2	64.9 95.5 116.1 121.7 119.3	73.4 105.0 125.7 138.4 147.6	82.8 107.4 105.5 103.0 100.2	75•7 103•0 125•1 121•5 125•0	73.0 96.4 107.2 106.7 108.7
	ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGUIM	LUXEM- BOURG	UK	IKELAND	DENMARK
< 21	50.5	55.2	69 .8	48.7	52.8	52.4	57•5	52.2	58.3
21 - 29	87.8	81.1	87.7	83.9	83.9	91.7	89.4	88.7	85.0
30 - 44	112.4	113.8	107.9	116.8	114.0	. 122•1	117.8	121.0	109-1
45 - 54	107.6	117.6	113.3	122.5	117.1	126.9	113.3	119.2	109.0
≥ 55	100.0	112.5	113.0	112.5	112.2	115.1	101.3	111.8	101.6
(*)	77.5#	97.3	-	98.i#			_		
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100	100.0	100.0
SOURCE	, Ωπιασβάζι , γαγατός Ωπιασβάζι , γαγατός	TABLE	DS 4	ć c ou	NTRY VOLUM	25			

SOURCE

TABLE DS

values range from 64.9 for Belgium to 83.3 for Italy. For the United Kingdom the peak value (107.4) comes in the 21-29 years group, for Germany and Ireland in the 30-44 years group, and for Italy, the Netherlands and Belgium in the 45-54 years group (values 114.5, 138.5 and 121.7 respectively). For France, Luxembourg and Denmark the peak comes in the 55 years and over age group with the value for Luxembourg (147.6) noticeably higher than elsewhere. The only other country where the index value is of comparable order is in the Netherlands (137.2) and both of these countries have a female labour force concentrated in the younger age groups.

Of particular interest are the indices for the youngest age-group. To illustrate the change between this group and the next youngest 21-29 years of age the following table has been constructed where the mean gross monthly pay for the 21-29 age group has been taken equal to 100.

Table 18

Mean gross monthly earnings of full-time employees under 21 years of age (Earnings of full-time employees aged 21-29 = 100, male and female employees considered separately).

	Male	Female
Germany	65.3	70.4
France	65.1	78.6
Italy	74.8	86.0
Netherlands	59.1	65.4
Belgium	65.4	68.0
Luxenbourg	55.7	69.9
United Kingdom	64.0	77.1
Ireland	56.8	73.5
Denmark	68.2	75.7

In all cases the change is greater for male employees than for females with the difference between the sexes least in the case of Belgium. Of all countries Italy shows the least difference between the age groups both for male and female employees. For male employees Luxembourg, Ireland and the Netherlands show the greatest differences between the age groups with a similar feature as regards femalo employees (though the order is interrupted by the figure for Belgium).

For comparisons between male and female employees when the effect of age is removed, (see Table 19 below) the value of the index for each of the subgroups observed to be greater than the ratio of the overall means. This reflects the different age distribution patterns for male and female suployees as has been mentioned previously.

TABLE 19

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES

(BY AGE) (MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY = 100)

AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEIS BOURG	UK	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54 $\gg 55$ (*)	86.9 80.6 73.7 69.8 74.8	93.0 77.2 67.6 66.1 73.5 70.7	102.1 88.8 81.6 82.5 79.1	88.1 79.7 73.2 72.1 78.1	77.5 74.5 68.2 70.9 74.4	85.4 67.9 64.8 68.7 83.7	77,4 64,2 47,7 47,1 54,3	83.5 64.5 62.0 61.1 67.5	89,3 80,4 70,9 70,2 78,5
ALL ALL	69.8	66.6	79.0	61.1	65.0	58.7	52,3	56.4	71.2

SCURCE

TABLE DS

COUNTRY VOLUMES

Of particular interest 's the figure for the under 21 years age group in Italy which exceeds 100 and also the figures for this age group in Belgium and the United Kingdom, which are noticeably lower than elsewhere. The figures for Italy are uniformly high while those for the United Kingdom are uniformly low (particularly in the 30-44 and 45-54 years age groups). The figure for the 55 years and older category in the United Kingdom is very noticeably lower than elsewhere and may reflect a "return to work" effect with older women in the United Kingdom returning to work in the lower paid jobs.

2. 5.3 Employees classified by length of service

There will obviously be a relationship between the length of service distribution and the age distribution in that younger employees will necessarily have served a shorter time than their older colleagues - except, perhaps, in jobs with high "turn-over" rates.

The figures are given below in Table 20

TABLE 20

.

.

LENGTH OF SERVICE DISTRIBUTION

(FULI-TIME EMPLOYEES)

	MA	LE		·				\$	
YEARS	GERMANY	FRANCE	ITALY	NETH.	BELG.	LUX	UK	IRELAND	DENNA RK
<2	26.7	28,0	27.1	28.6	30.4	33.6	30.6	13.0	20.0
2-4	23.9	24,1	25.1	24.2	24.2	24.3	23.7	2.4.8	25.1
5-9	20,2	20,2	20.6	19.6	19.2	16.0	17.5	22.5	20,9
10-19	19.8	17.3	20,2	17.1	16.9	17.2	15.6	21.3	18.5
<i>¥</i> 20	9.4	7.8	7.0	9.9	9.3	8.9	12,1	18,3	13.0
·	100	100	100	100	100.0	100.0	100	100	100
JUMBERS	439.1	372.0	251.4	176.7	80.9	3,8	353,1	23.4	77.2
000 3	FEMA	LE							%
YEARS	. GERMANY	FRANCE	ITALY	NETH.	BELG.	LUX.	UK	IRELAND	DEIIMARK
< 2	34.9	32.7	35.6	51.4	37.0	48,5	38.3	26.3	32.0
2-4	30.5	27.9	28.2	29.3	27.5	28.1	29.4	36-1	32.3
5-9	19.1	18.2	19.6	11.7	18,5	13.9	17.1	19.9	17.9
10-19	11.3	12.7	13.1	4.8	11.7	6.5	9,1	10.6	10.1
≥20	4.2	5.4	3.5	2.3	5.3	3.0	4.7	7.0	5.5
	100	· 100	100	100.0	100	100	100	100	100
Nillagers Dea's	152.6	151.9	116.6	31.0	31.5	1.5	118.2	6.7	15.4

Between 50 and 60 per cent of full-time male employees have served less than 5 years and between 25 and 30 per cent have served 10 years or more in all countries apart from Ireland and Denmark. In these countries the distribution is more equally belanced between longer and shorter legnths of service, particularly in Ireland where 37.8% have served less than 5 years and 39.6% have served 10 years or more. The different position of Ireland is shown even more clearly if only the longest, and shortest, length of service categories are considered.

For full-time female employees, there is evidence of a greater concentration into the shorter length c. service categories than for male employees with considerable variations from one country to another. Netherlands and Luxembourg show the most pronounced concentration into the shortest length of service categories (a reflection of the age structure in Table 16) and Ireland the least. As in the case of male employees in Ireland, it is there that the percentage of females in the longest length of service category is highest - though it is less noticeable than in the case of male employees.

In a more detailed analysis it would obviously be desirable to analyse the length of service distribution for narrower age bands as has been done in the 1966 Structure of Earnings Survey (eg male employees aged 30-44 years). However this information is not available from the published data and so we have confined our analysis to the global pictures.

Indices

As in the previous section, two sets of indices are considered -"within sex" and "between sex". - Tables 21 and 22 respectively.

For males there is an increase in the index with increasing length of service for all countries with the most noticeable increases for the longest service (20 years and over) for France (146.8) and Luxembourg (140.6) respectively.

For female employees there is a similar pattern. The entries in the cells are higher than the corresponding entries for male employer - a similar (and related) feature to that observed for the indices for various categories in the age distribution. The figures for France, Italy and Belgium are similar-particularly in the 20 years and over category (145.8, 245.3 and 145.4 respectively). The figures for Germany and Ireland are similar for this category (140.6 and 139.9) and those for the United Kingdom and Denmark comparable but much lower (127.5 and 124.5) respectively).

SB TABLE 21

INDEX OF MEAN GROSS MONTHLY EARNINGS - BY LENGTH OF SERVICE

MALE

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
. < 2 2 - 4	88.8 94.9	78.2 90.6	84•9 94•8	81.9 95.3	79•7 95•8	75-9 94-4	83.3 86.8	77•7 86•3	82 . 8 90.8
5 - 9	102.6	102.7	102.4	104.5	108.4	108.6	106.7	99.7	101.1
10 - 19	109.4	119.3	112.7	115.3	116.0	120.6	119.3	112.5	112.5
≥ 20	117.3	146.8	130.1	126.8	121.8	140.5	130.9#	118.4	122.9
(*)	-	103.4		126.0#	- 1	~~	*	ند	105.0
ALL	100	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

	LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	IJK	IRELAND	DENMARK
	< 2	90.2	84.6	85.7	88.0	81.7	79.8	93.0	81.1	89.7
	2 - 4	94.1	94.2	97•4	99.8	98.3	99•7	96.2	93.4	98.5
	5 - 9	106.2	104.1	106.4	120.5	108.2	117.7	105.7	110.8	105.4
	10 - 1 9	118.5	121.6	120.8	143.9	122.2	144.8	114.4	121.0	113.9
	> 20	140.6	145.8	145.3	171.1	145.4	194.6	127 • 5	139.9	124.5
;	(*)	•	99.0	-	97.7#	-	-	٤		<u>93.7</u>
	ALL	100	100	100	100	100	100.0	100.0	100.0	100.0

MALE AND FEMALE

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2	. 87.5	78.7	83.1	79.8	78.6	74.1	83.0	72.7	82.2
2 - 4	93•2	90.4	95.0	94.8	95 .5	93.7	86.4	84.5	91.0
5 - 9	103.6	103-7	103.8	107.7	108.8	111.3	106.4	102.2	102.1
10 - 19	113.8	122.2	116.3	120.8	120.0	131.1	124.8	118.5	114 . 9
≥ 20	124.3	149.9	136.0	133.5	129.5	155.3	140.2	126.9	126,2
(*)		101.1	-	125.9 [#]	-		5	5	104.0
ALL	100	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100,0

SOURCE

TABLE DIO

COUNTRY VOLUMES 48

Perhaps the most noticeable values are those for Netherlands and Luxembourg where the values attained in the longest length of service category are considerably higher than elsewhere (171.1 and 194.6) respectively). A possible explanation lies in the age and length of service distributions. In these countries, proportionately more are younger and have served a shorter length of time than in other countries. This concentration tends to depress the overall average earnings and the mean earnings of these longer serving employees therefore appears proportionately much greater than those for other countries. In the comparison between male and female employees, when the effect of length of service is removed, see Table 22 below, the values of the indices for mest of the subgroups increase but apart from Netherlands and Luxembourg less noticeably than in the case of the age distribution.

TABLE 22	INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIKE FEMALE EMPLOTEES - BY LENGTH OF SERVICE	•
	(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY - 100)	

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 \geq 20 (*)	70.9 69.2 72.3 75.6 83.8	71.9 69.2 67.4 67.8 66.1 63.7	79.8 81.2 82.1 84.8 88.3	65.7 64.0 70.5 76.3 82.5 47.4#	66.6 66.8 64.9 68.6 77.7	61.7 62.1 63.6 70.5 81.3	58.4 58.0 51.8 50.2 51.0#	58,9 61.0 62.7 60.7 66.7	77.0 77.2 74.2 72.0 72.1 63.5
ALL	69.8	66.6	79.0	61.1	65.0	58.7	52.3	56.4	71.2

SOURCE

COUNTRY VOLUSES

Once again the figures for Italy are consistently higher than elsewhere while those for the United Kingdom consistently lower. In Germany, Italy, Netherlands, Belgium and Luxembourg there is evidence of an incroase in the value of the ratio with increasing length of service. In France, Denmark and the United Kingdom the reverse pattern is observed.

2,5.4 Employees by Occupational Qualification Group

So far we have discussed the full-time labour force analysed by sex, age and length of service. All these factors are inter-related and also with occupational qualification group which is the factor which is discussed in this section.

Classification of employees according to qualifications differ considerably between economic activities and between countries, and it was therefore left to the national statistical institutes to classify employees in broad groups on the basis of qualifications which were defined by common agreement for the separate sectors of wholesale and retail distribution, banking and insurance.

For wholesale distribution every employee was assigned to one of the five qualification groups given below depending on the tasks performed, the degree of difficulty involved, the inherent responsibilities and the level of training or experience required for doing the job.

The groupings are:

- 1. Management personnel
- 2. Executives and managerial staff
- 3. Highly qualified junior personnel
- 4. Skilled junior personnel
- 5. Unskilled junior personnel

A fuller description of these categories is given in the Technical Annex.

Groups 1, 3 and 4 were subdivided into 2 further categories but in the main tables only the subdivision of category 1 was given (Annex).

With these definitions the figures given in Table 23 were obtained.

TABLE 23 GRADE STRUCTURE (% OF FULL-TIME EMPLOYEES IN EACH CATEGORY)

_	MALE	_	<u>`</u>					Terror Inc.	6
BRADE	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5	0.9 2.5 14.5 44.8 26.7 10.6	4.3 10.0 17.3 15.9 34.7 17.8	1.4 2.6 6.7 13.6 38.0 37.6	3.7 8.3 17.1 32.9 31.5 6.5	2.8 6.2 16.5 29.1 27.3 18.0	3.0 3.9 11.5 29.0 36.1 16.6	3.1 3.2 15.9 16.8 48.5 12.5	5.6 8.3 18.6 48.1 19.3	3.1 3.4 11.9 22.4 32.0 27.1
JMBERS JOO's)	439•1 <u>-</u>	372.0	251.4	176.7	80.9	3.8	353.1	23,4	77.2

FEMALE

91

-									10
IRADE	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5	0.3 0.5 4.0 35.6 40.9 18.7	0.7 2.8 5.6 16.1 47.5 27.3	0.1 0.4 2.2 9.8 45.2 42.2	0.5 2.9 9.7 30.8 51.0 5.0	0.2 1.2 7.0 34.1 39.2 18.3	0.8 0.9 8.6 27.5 50.1 12.2	3.4 7.2 71.8 16.9		3.2 15.4 52.3 28.7
UMBERS	152.6	151.9	116.6	31.0	31.5	1.5	: 118.2	6.7	15.4

MALE AND FEMALE

%

JRADE	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	IRELAND	DEMIARK
1A 111 2 3 4	0.7 2.0 11.8 42.4 30.4	3.3 7.9 13.9 15.9 38.4	1.0 1.9 5.3 12.4 40.3	3.2 7.5 16.0 32.6 34.5	2.0 4.7 13.5 30.4 30.3	2.4 3.0 10.7 28.6 40.0	2.3 2.6 12.7 14.4 54.3	4.7	2.6 2.9 10.5 21.2 35.4
	504 P	20.7	29.1	0.3	19.1	15+3	15.0	20.1	02.6

There are considerable variations in the percentages for all the categories for the different countries. The figures would seem to indicate very different structures though perhaps some of the effect might be due to problems in the interpretation of the definitions, in particular those for category 3 (highly qualified junior personnel) and category 4 (skilled junior personnel).

Some of the main differences may be seen by considering the percentages for male and female employees combined. For example, in Italy 79.4% of full-time employees fall into either category 4 or category 5, (skilled junior personnel and unskilled junior personnel respectively). For Ireland, the United Kingdom and Denmark the corresponding percentages are 72.9, 67.9 and 62.8. Germany and Netherlands present a different picture with 43.1% and 40.7% respectively. Germany has proportionately the largest category 3 (highly qualified junior personnel) with 42.4% of all full-time employees in that category. This contrasts with the percentages for Ireland, the UK and Italy at 15.9, 14.4 and 12.4 respectively. Management Personnel (category 1) accounts for 2.7% of full-time employees in Germany (and 2.9% in Italy) whereas in France the responding percentage is 11.2 and in the Netherlands, 10.7.

When the figures for men and women are considered separately these differences in structure are confirmed. Female employees tend to be more concentrated into categories 4 and 5 than do male employees with very few female employees in the top two categories, particularly in the United Kingdom, Ireland and Denmark.

Indices

There are two main considerations to be noted when interpreting these particular indices and are as follows:-

- 1. Several of the entries in the tables, particularly for female employees in category 1A have restrictions put on them and are marked either 'data for which the standard error of the estimate of the mean lies between 5% and 10%' or 'data is non-existent or not collected' or 'data not published, or confidential, or relating to an undersize sample (less than 10 employees), or for which the standard error of the estimate of the mean equals or exceeds 10%'.
- 2. Most member countries did not record the exact remuneration of employees whose earnings exceeded a certain ceiling (cf. footnote to section 8.14 of Volume 1 of the survey and also the Technical Annex). These ceilings varied from country to country and

and in the volume for each country there is a preliminary table which gives the percentages in each occupational group exceeding the ceiling. As a percentage of all full-time employees in Wholesale Distribution the numbers exceeding the ceiling in each country are very small, and in every case is less than 1 per cent. However, most of these 'high earners' tend to be concentrated in either category 1A or 1B and when taken as a percentage of all employees in these categories the percentages are somewhat larger. Volume 1 indicates that these earners were excluded from the calculation of the mean. The effect of this on the overall mean might be expected to be small but on the means of these particular categories much greater. This would have the effect of depressing some of the indices obtained for these topmost categories. A fuller discussion, with examples is given in the Technical Annex.

The figures which have been obtained are given in Table 24. With the above comments in mind it is interesting to note that for males the most variation by occupational qualification appears in France with employees in categories 2, 3, 4 and 5 faring proportionately worse than their counterparts in

INDEX OF MEAN GROSS MONTHLY EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 24

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	Be lciu m	LUXEN- BOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	198.3 184.6 139.8 99.5 80.0 71.4	2.83.6 179.4 121.0 84.7 69,1 57.3	254.6 184.7 152.0 122.6 92.1 76.0	226.2 156.8 127.4 91.6 75.5 65.8	2405 (730 (25.1 93.6 78.1 675 -	207,3 181,2 130,1 100,6 84,7 70,4	276.5# 188.2# 125.8 101.5 79.3 70.4	193,6 131,6 116.0 86,3 74,8	201.7 177.8 138.1 106.0 85.6 74.9
ALL	100	100	100	100	100	100	100	100	100

FEMALE

**************************************	GERMANY	FRANCE	ITALY	NETHER- LANUS	BELCIUM	Luxes- Bourg	UK	IRELAID	DERMARK
1A 1E 2 3 4 5 (*)	231.9# 229.7 159.1 117.5 87.8 73.7	3498 212.6 156.5 116.9 89.6 74.3	282.7# 214.2 174.0 137.5 101.2 83.0	2738 155.0 133,2 107.5 85.6 80.6	264.8# 195.6 147.9 108.6 89.9 76.4	275.0 137.3 112.A 88.0 69.3	130.5# 128.1 98.1 85.2	185.9# 133.4 97.6 83.3	147,5 117.0 98.9 84.1
ALL	(00	100	100	100	100	100	100	100	

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGTUM	LUXEM- BOURG	UK	IRELAND	DENLARK
1A 1B 2 3 4 5 (*)	210,5 198,3 148,9 103,8 79,7 69,1	309.7 194.1 131.7 91.4 72.7 59.9	270.6 196.3 160.5 127.4 94.0 77.5	238.2 162.4 125.9 93.3 74.7 67.8	265.2 188.5 134.5 95.4 78.7 69.0	2264 20011 1345 1031 829 716	313,7# 207.7# 138.4 110.4 79.2 71.0	213.1 144.5 124.7 84.3 78.5	211.7 186.0 143.3 108.5 86.0 76.0
	100	100	100	Construct of Construction of Fichage of Lange Statements of Constructions	100	100	100	100	100

other countries. On the other hand males in these categories in Italy fare proportionately better than elsewhere.

For female employees, the percentages tend to be higher than in the corresponding categories for men. This may be traced to the different distribution of employees over the occupational qualifications with the concentration towards the lower categories - a feature seen earlier with similar indices for age and length of service. Proportionately fewer reach the higher categories - in particular 1A and 1B, but relative to all full-time female employees, those who do, fare better. Of particular interest in this connection are the figures for categories 2 and 3 for Italy and Ireland which are noticeably higher than for the same categories in other countries.

When the effect of the occupational qualification structure is removed (from the comparisons between male and female employees (see Table 25 below) the values taken by the indices in each of the individual categories are higher than the overall index figure in each case.

TABLE 25

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-ATME FEMALE ENPLOYEES - BY OCCUPATIONAL QUALIFICATION GROUP

(MEAN GROSS MONTHLY EARNINGS OF FULL-TTHE MALS EMPLOYEES IN SAME CATEGORY = 100)

						the second second second second			
	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
1 2 3 4 5 6 (*)	81.7# 86.9 79.4 82.4 76.6 72.1	82.1 78.9 86.0 9!.9 86.3 86.3	87.7# 91.6 90.5 88.6 86.8 86.8 86.3	74.0* 60.4 66.5 71.7 69.3 74.9	71.6# 73.5 76.9 75.4 74.9 73.6	- 72.9 62.0 65.9 61.1 57.8 -	- 54,3# 66.0 64.7 63.3	- 79.7# 64.9 63.9 62.8 -	7610 7816 82.2 79.9
all the same of the second second	69.8	66.6	79.0	61.1	65.0	58.7	52.3	56.4	71.2

SOURCE

TABLE C2

COMPARY VOULAES

This is an effect which has been noted for the other indices and may be traced to the properties of the occupational qualification distributions for male and female employees.

It is interesting to note that the figures for Italy for each category are consistently higher than elsewhere (apart from category 3 in France) and that the figures for Luxembourg, the United Kingdom and Ireland (apart from category 2 - where the reliability is suspect in any case) are low. In Germany and Luxembourg (and to a much lesser extent in Italy) the ratios for the higher grades (1A, 1B and 2) are naticeably higher than those for the lower (4 and 5). This pattern is reversed in France, with the United Kingdom, Denmark and Belgium showing similar if less pronounced features, and with Belgium showing the least variation apross occupational qualifications.

2.5.5 Employees classified by size of undertaking

The final set of indices which we shall consider in this section is that where the factor of classification is the size of the undertaking to which the employee belongs. In an earlier section (SECTION 2.3) we considered the distribution of all employees over the different sizes of undertakings. In Table 26 the figures are shown for fulltime employees only together with the distributions for male and female employees separately are also given.

As full-time employees predominate in the labour force the features mentioned earlier are observed for these figures (eg the concentration in all countries into the smaller sizes of undertakings with the United Kingdom the main exception in the case of the largest size category of undertaking. The figures for male and female employees are also very similar with some evidence of a greater concentration of male employees (than fewale employees) in the smallest size of undertaking category in Germany, France, the Netherlands, Luxembourg and the United Kingdom with the opposite effect observed in Belgium, Ireland and Denmark.

TABLE 26 FULL-TIME EMPLOYEES BY SIZE OF UNDERTAKING

M	Al	.E

							·		10
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	37.4 18.6 15.2 14.3 6.2 8.3	41.1 16.9 11.9 13.2 5.5 11.4	58.1 16.8 11.2 7.8 4.2 2.0	43.6 18.8 14.3 13.2	46.8 18.4 13.1 10.4 8.9 2.4	50.1 27.2 17.2 5.5	31.4 15.2 13.1 12.6 8.1 19.7	26.7 18.9 22.2 18.6 10.1 3.4	37.3 15.1 14.0 13.0 8.8 11.7
ALL	100.0	100	100	100	100	100	100	100	100

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	33.2 18.1 16.9 16.0 7.7 8.1	37.1 18.2 13.4 14.6 5.5 11.3	58.1 17.9 11.1 6.8 4.1 2.0	42.7 20.3 15.3 13.0	48:3 18:3 13:4 12:0 6:3 1.7	46.1 22.0 21.0 10.9	26.5 :5.3 12.7 14:9 11.4 19.2	28:5 17:4 22.0 16:8 7.9 7.3	38,4 14,40 15,1 16,1 5,10,7
ALL	100.0	100.0	100	100	100	100	100	100	100

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	36.3 185 15.7 14.7 6.6 8.3	39,9 17,3 12,3 13,6 5,5 11,4	58.1 17.1 11.2 7.5 4.1 2.0	43.4 19.0 14.5 13.2	47.2 (8.4 13.2 10.9 8.1 2.2	49.0 25.7 18:2 7.0	30.2 15:2 13.0 13.1 8.9 19.6	27.1 18.6 22.2 18.2 9.6 4.2	37,5 15,0 14,2 13:5 8,3 11,5
ALL	100.0	<u>i</u> @0,	100	(00	100	100	(00	100	(00

SOURCE

TABLE #3

COUNTRY VOLUMES

ð

60

Indices

The values obtained are given in Table 27 and 28. Table 27 gives the comparisons between employees of the same sex. Table 28 compares male and female employees in the same size of undertaking category.

Of particular interest in Table 27 are the differing patterns which are in evidence for male employees. Apart from Luxembourg and the United Kingdom, there is evidence of an increase in the index with increasing size of undert-king. This feature is most noticeable for Italy and to a slightly lesser extent for Iraland. On the contrary the United Kingdom exhibits the reverse pattern with a decrease in index as the size of undertaking increases.

For female employees there are several patterns. There is a noticeable increase in the index with increasing size of undertaking in Italy and Ireland. France and Germany show increases in the index between the smallest size category and the largest but pass through lower values in between. The Netherlands and Denmark show increases with size, though the increase is not a smooth and progressive one. Belgium
TABLE 27 INDEX OF MEAN OF MEAN GROSS MONTHLY EARNINGS BY SIZE OF UNDERTAKING

MALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49	96.0	95.4	91.2	98.5	94.7	101.6	107.4	92.5	95.9
50-99	101.4	99.2	105.8	98.2	100.4	96.0	100.1	92.4	100.4
100-199	99.2	102.2	108.0	99-1	105.3	102.5	100.5	99.6	101.7
200499	102.2	101.9	124.4	105.0	104.7	96,8	96.2	105.0	98.6
500-999	103.0	103.8	132.5	-	112.1	-	95.7	129.6	105.3
≥1000 (*)	110.4 -	110.8 -		-	103.5	-	92.0	•	107 . 1
ALL	100.0	100	100	100	100.0	100.0	100.0	100.0	100.0

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49	100.9	100.0	95 . 4	98.2	93.8	105.8	102.6	92.0	97.5
50-99	100.2	99 •7	97.8	99.5	100.5	91.5	9 9 ₌ 4	97.7	100.6
100-199	98.7	96.2	101.6	99.3	107.9	97.1	92.6	98.5	101.0
200-499	99.4	95.8	119.2	103.9	103.4	97.3	99•5	110.5	100.0
500-999	96.7	103.5	121.3	-	117.1	- 1	101.3	131.8	102.2
≥1000	102.7	108.3	- .	-	114.7	-	101.7		105-1
(*)	-		-	-	-	-	-	-	-
ALL	100.0	100.0	100.0	100.0	100.0	100	100	100	100.0

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARX
10-49	97.6	97.0	92.3	98.5	94.3	102.8	108.9	91.9	96.0
50-99	:01.3	٥8.8	103.1	97.8	100.3	96.5	99•3	93•7	100.6
100-199	98.4	100.0	105.5	98.8	105.6	98.7	99-3	99•5	101.2
200-499	100.8	99.8	124.3	105.2	103.0	95.3	94.8	106.8	97.8
500999	100.2	103.8	128.4		116.0	-	93.5	132.1	107.7
≥1000	109.2	110.4	-	-	107.6	in a second	93.9		107.2
(*)	-	-	-	-	_	- 1	-	-	
ALL	100.0	100 . C	100.0	100.0	100.0	100.0	100.0	100.C	100

SOURCE

.

62

COUNTRY VOLUMES

.

also shows increases with size with the peak in the 500-999 size category. For the United Kingdom the index decreases uniformly to its lowest value corresponding to the 100-199 size category, from which it increases uniformly to a value, for the greater than or equal to 1000 employee category, which is less than that for the 10-49 employee category. Luxembourg shows a similar pattern with a high (105.8) in the smallest size category, a low in the 50-99 size category and rising to 97.3 in the 200-499 size category. A more detailed analysis of the undertaking sizes in the subsectors would probably throw more light on these variations in pattern.

When the effect of size of undertaking is taken out of the comparison between male and female employees as is shown below in Table 23 it is seen that the indices for the individual categories vary less in general from the overall ratios than was the case when a similar procedure was adopted for age, length of service and occupational qualification proup.

TAPIS LO

THDER OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES - BY SIZE OF UNDER-TAKING

(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY=100)

Table 2	20								
NO OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENMARK
10 - 49	73.4	69.8	82.9	60.9	64.5	61.1	50.0	56.2	72.4
50 - 99	69.0	66.9	76.7	61.9	65.1	56.0	51.9	59.7	71.3
100 - 199	69.5	62.7	74.8	61.3	66.6	55.6	48.2	55,8	70.7
200 - 499	68.0	62.5	72.8	60.5	64.3	59.0	54.1	59.4	72.2
500 - 999	65.6	66.3	72.1	-	67.9	-	55.4	57.4	68.5
≫ 1000 (*)	_64.9	65.1	78.4	-	72.1	-	57.8	52.5	67.8
ALL	69.8	66.6	79.0	61.1	65.0	58.7	52.3	56.4	71.2

SOURCE

COUTTRY VOLUMES

TABLE D6

This is due in part to the fact that the distributions for male and female employees by size of undertaking are similar (which was not the case in the other distributions).

As in the case of the other indices the figures for Italy are higher than elsewhere and those for the United Kingdom, Luxembourg and the United Kingdom low. There is some slight evidence of a decline in the index with increasing size in Germany (and France) and of the reverse effect in Belgium and the United Kingdom. In the Netherlands the ratio remains fairly constant (but no figures are disclosed for the categories with undertakings of sizes above 499 employees). In Ireland there is an overall decline in the index from smallest to largest size category but higher values are reached in between. In the remaining countries a "U" shaped pattern is exhibited with the ratic for the largest size category less than that for the smallest.

2.6 Gross Annual Earnings

The analysis so far has looked at gross monthly earnings for full-time employees, classified by various factors. A similar analysis is possible for gross annual earnings for which a frequency distribution containing 27 class intervals was drawn up for each country.

Indices were calculated for mean gross annual earnings in a manner similar to that for mean gross monthly earnings and are available for all full-time male employees, all full-time female employees and for these groups classified by occupational qualification group. Indices are also available to compare male and female earnings for each of these categories and the global figures are given in the summary table below.

Table 29

Summary of Female/Male Mean Gross Annual Earnings (Earnings of full-time male employees o 100)

Germany	70.0	(69.8)
France	64.2	(66.6)
Ttaly	75.7	(79.0)
Netherlands	61.6	(61.1)
Belgium	64.7	(65.0)
Iuxembourg	58.9	(58.7)
United Kingdom	51.3	(52.3)
Ireland	56.8	(56.4)
Denmark	09.2	(71.2)

65

Figures in brackets are the monthly ratios

SOURCE TABLE C4 COUNTRY VOLUMES

INDEX OF MEAN GROSS ANNUAL EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 30

.

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXE1- LOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	196.5 183.4 139.1 97.9 78:4 70.1	29.3.9 170.4 116.8 79.8 65.2 53.7	295.2 218.0 153.6 120.9 87.9 72.0	220.9 155.0 118.8 90.5 74.0 66.3	231.3 •1645 120.6 90.2 73.6 64.5	200.6 (75.4 126.2 97.1 81.2 67.1	277.(# 91.3# 19.2 97.0 77.6 - 70.6	195,5 129,7 113,2 85,1 75,0	202.0 173.4 136.9 105.5 83.5 73.3
ALL	100.0	100.0	100.0	100.0	100.0	(00.0	100.0	100.0	(00.0

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	DK	IRELAND	Denmark
14	225.2件	361.2	301.8#	258.3#	267.9#	0	-	e	
1.B	224.8	212.6	237.3	159.9	195.8	213.6	e "	~	•
2	155.8	153.9	178.3	133.7	143.5	12.9.8	133,4#	182.4#	146.8
3	115.9	13.3	135,8	106.4	106.5	110.5	126.0	131.6	111.5
4	87.1	87.0	98,7	83.5	88.3	86.9	17.4	9/.1	97.0
(*)	72.1	71.3	82.3	74,8	75:7	66.4	81.7	82.5	85.8
ALL	100.0	100:0	100.0	100.0	100.0	100.0	100.0	(00,0	100.0

.

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LATTO	BELGIUM	LUXEM- BOURG	UK	IRELAND .	Demlárk
1A 1B 2 3 4 5 (*)	207.3 196.0 147.5 102.0 78.2. 67.5	321.5 185.3 127.3 86.3 68.7 56.2	314.9 231.8 162.7 125.4 89.9 74.2	230,3 159,8 121,7 91,9 73,2 67,6	255,2 179,8 129,7 92,7 74,4 66,4	217.9 192.6 130.0 99.6 79.8 68.2	313,3# 210,2# 130,6 105,1 77,5 69,4	213,4 - 141,4 121,1 83,3 78,5	212.3 181.5 142.0 107.8 83.7 74.4
Notesta and some second second	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

For most countries as might be expected, the agreement between the values for annual ratios and monthly ratios is good, with the annual ratios tending to be lower in most cases than the monthly ones. The most noticeable difference between annual and monthly ratios is for Italy (and may possibly be related to the different structure of bonus payments, see section 2.8)

Table 30 gives the various indices of mean gross annual earnings by occupational qualification group. The comments made earlier regarding gross monthly earnings apply to the annual figures (cf section).5.4). The figures are generally in good agreement with the monthly ones in Table 24 The differentials between the mean earnings of the higher as opposed to the lower occupational qualification groups appear to be slightly narrower in the case of the annual figures than was observed for the monthly ones, in Germany, Belgium, the Netherlands and Denmark with a widening in the remaining countries with the effect most pronounced in Italy and France.

The figures giving the comparisons between female and male mean gross annual carnings are given below in

Table 31

TABLE 31 INDEX OF MEAN GROSS ANNUAL EARNINGS OF FULL-TIME FEMALE EMPLOYEES BY OCCUPATIONAL QUALIFICATION GROUP

(MEAN GROSS ANNUAL EA	ARNINGS OF	FULL-TIME	MALE	EMPLOYEES	IN SHME	CHIEGERATION
-----------------------	------------	-----------	------	-----------	---------	--------------

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	IUXEM BOURG	UK	IRELAND	DENMARK
1Λ 1B 2 3 4 5 (*)	80.2# 85.8 78.4 82.8 77.7 72.0	79.0 80.1 84.7 91.2 85.8 85.2	77.4# 82.4 87.1 85.0 85.0 86.6	72.0# 63.5 69.3 72.4 69.5 69.5	74.9# 77.0 76.9 76.4 77.5 75.9	 60:6 67:1 63.1 58:3	• 57.4# 66.6 64.4 59.5	79.8# 66.0 64.8 62.5	74.2 77.0 80:4 79.2
 And a state of the state of the	70.0	64.2	75,1	61.6	64.7	58.9	51.3	56,8	69.2

SOURCE

TABLE C4

COUNTRY VOLUMES

As in the figures in the previous table, these are also, in general, in good agreement with the corresponding monthly ones given in Table 25. Perhaps the most noticeable differences are that the values in categories 1A and 1B in Italy are much lower than the corresponding

entries for the monthly figures (87.7 and 91.6 respectively). Similarly figures noticeably lower than the monthly ones are found in category 5 in the Netherlands and the United Kingdom.

2.7. Length of Scheduled Working Week

An important factor which has a bearing on all the earnings data and indices considered so far is the length of scheduled working week. As is set out in section 8.11 of Volume 1, the figure entered was the number of hours which the employee was required to work each week under the relevant contract, regulations or system operated in the undertaking, regardless of the rate of payment for these hours.

If, in the undertaking, the duration of work varied from week to week, the figure entered was the average weekly duration in the reference period.

The results obtained are shown in Table 32 below. It is interesting to note the variations which the figures reveal. It is noticeable that for both men and (and Ireland) women the United Kingdom has much higher percentages working shorter hours (38 hours or less) than in other countries. For Germany and Italy the concentration of employees, both male and female, having a scheduled working week of more than 38 hours but less than or

but for males two categories, greater than 38 hours and less than or equal to 40 hours, and greater than 42 hours but less than or equal to 44 hours together account for 88.8% of employees. It is also interesting to note that France has the highest percentages both for men and women working a scheduled week of more than 42 hours.

MALE				·					*
HOURS	GERMANY	FRANCE	ITALY	NETIIS.	BELOIUN	wx,	UK	TRELAND	DENITARK
€ 36		0,0	1.0	0.2	. 0.7		12.6	11.6	1.5
>36- 38	0.1	0,1	0.4	3.6	4.1	-	16.2	(3,9	12.5
>38- 40	80,8	27.9	93.2	42.5	54.7	60.9	51.2	59,4	35.7
>40- · 42	11.6	14.4	0.8	49.4	39 . 2	7.6	4.3	5.0	47-9
>42 44	1.9	19.4	2.2	1.3	0,8	27.9	2,8	4.2	0.5
>44- 46	3.0	16.2	1.0	2.6	0.4	0.9	.1.9	2.4	0.5
746- 48	1.3	12.2	1.2	0.2		i.4	0.8	1.2	•
> 48		9.3	0.1	0.2		1.2	1.0	2.3	0.4
% UNDISCLOSED	1.3	0.4	0.1				9.2		0,8
FEMALE		×							ħ
HOURS	GERMANY	FRANCE	ITALY	NETHS.	Belcium	LUX.	UK	TRELAND	DERIMARK
≰ 36	0,1	0,1	1.3	0.5	1.1		39.4	21.9	5.3
> 36 38	0.2	0.4	0,2	7.9	5.9		33.6	26.4	20.7
>38- 40	86.9	49.4	91.1	52.4	52.5	89.3	24.4	46.2	43.0
7 40- 42	10.0	21.7	0.8	37.5	39.3	1.7		2.9	29.9
7 42- 44	1.2	15.7	2.7	0.5	0.7	8.3	•	1.7	•
744- 46	1.1	8,6	0.6	.1;2	0.4		•	· • ·	-
>46 48		2.6	2.9		-	-			-
	0.3	£40							•
> 48	-	1.2	0.1	42			•	•	,• == ;

SOURCE TABLE AS COUNTRY VOLUMES

2.8 BONUS PAYMENTS

The sections in Volume 1 which relate to bonus payments are 8.17 and 9.2.5. (p85 in English translation) and the interpretation of the definition of "bonuses and premiums not paid on a monthly basis in 1974" may have created some problems and have tended to accentuate differences in the structure of bonus payments across the countries.

The figures are given in Table 33 and perhaps the most striking feature is the position of the United Kingdom France and Denmark on the one hand with high percentages in the "O percent" bonus category and the position of Italy on the other hand with the 15 to 20 per cent bonus category being the largest. Similar patterns hold for both male and female employees with the pattern for female employees, if anything, slightly more concentrated. The other countries fall between these two extremes. Germany, France and Luxembourg are towards the lower end (over 75 per cent of male employees in these countries have bonuses of less than 10 per cent). In Netherlands over 75 per cent of the male employees have bonuses in the range 5-20 per cent. In Belguim the main concentration covers the 5-15 per cent bonus range. It is interesting to note that the UK has a relatively much higher percentage figure (3.2) for males in the 50 per cent or more bonus category.

TABLE 33

BONUS PAYMENTS - WHOLESALE DISTRIBUTION

MALE

(FULL-TIME EMPLOYEES)

· · · · · · · · · · · · · · · · · · ·					•				-
DRUMARK	TRELAND	UK	LUX.	BELG.	NETHS.	ITALY	' FRANCE	GERMANY	% BONUS
84.4	76.6	65.5	25.5	11.9	5.3	-	30.0	4.5	0
: 7.0	14.8	16.9	25.4	16.5	5.1	3.8	20,1	39.3	>0-5
2.3	3.8	6.4	32.1	24.9	39.3	9.8	31.3	34.6	≥ 5-10
2.0	1.5	2.5	8.2	27.7	22.6	19.5	9.6	13.8	710-15
0.9	1.1	1.4	5.3	8.3	14.8	49.1	3.6	3.6	715-20
1.0	1.0	2.3	2,1	8.0	7.4	13.6	2.2	2.5	220-30
1.0	0.8	1.9	0.9	2,0	3.3	3.7	1.2	1.1	730-50
0.9	0.4	3.2	0,6	0.7	2.2	0.5	0,8	0.6	> 50
DT.IMARK	ireland	UK	LUX.	BELG.	NETHS.	ITALY	FRANCE	GERIANY	emale % bonus
87.5	78.9	70.6	34.5	13.4	8,7	-	26.3	4.5	0
8,7	14.5	17.9	15.6	13.8	7.0	2.9	16.4	32.1	<u>ک م-</u> 5
1.7	4.5	6.5	33.8	23,1	38.9	10.0	36.3	38.5	25-10
•	0.8	2.3	8.1	30.2	22.8	14.4	13.0	17.9	7/10-15
•	•		5.3	9.9	15.6	52.8.	4.2	4.4 .	7,15-20
•			2.3	8.4	5.2	12.8	1.6	1.9	1/20-30
•				1.1	1.0	6.0	0,6	0.6	730-50
	•		·	0.1	0.8	1.1	0.4	0.2	7 50
								1	Selection of the American States

SOURCE TABLE B1 COUNTRY VOLUMES

72.

The final section of this chapter is concerned with parttime employees. From section 8.10 of Volume 1 of the series we have the note that 'the term 'part-time' was taken to denote a weekly duration of work that was systematically shorter than the normal working week'. Some summary statistics were given about part-time employees in Table 10. These are given again in aslightly rearranged form in Table 34 below.

Table 34 Part-time Employees

	n (*	umbers 000*s)	AS A PERCENTAGE OF ALL EMPLOYEES				
	All part-time	Part-time Female	All part-time	Part-time female			
Germany	108.5	95.9	15.5	13.7			
France	24.0	14.7	4•4	2.9			
Italy	7.1	5.4	1.9	1.4			
Wetherlands	17.3	14.0	7 .7	6,2			
Relation	4.0	2.9	3.4	2.5			
Invenhaure	0.3	0.2	5.4	3.6			
United Kingdom	65.9	50.1	12,3	9.3			
Int for writewow	0.4	0.3	1.3	1.0			
Denmark	13.7	12.3	12.9	11.5			

As can be seen from these figures, and as was mentioned earlier part-time employees play 3 role of varying importance from country to country within all cases, the majority of part-time employees being female.

These results are based on samples and in many cases the cell sizes became very small and the resulting percentages, averages etc subject to large standard errors. There are therefore many occasions on which the results were considered to be too unreliable for publication. In particular no values of the index for mean gross monthly earnings or mean gross annual earnings have been quoted for Ireland or the Netherlands. However it is useful to consider the results which are available.

2.9.1 Length of Scheduled Working Week

It is perhaps of interest to see how this varies from country to country. Table 32 gave the figures for full-time employees. Table 35 now gives the figures for part-time employees.

In all countries, the concentration for both male and female employees is in the less than or equal to 24 hours category with the degree of concentration varying from one country to another. For male employees it is interesting to note that the only figures quoted for Belgium, Luxembourg, the United Kingdom and Ireland (for reasons of small sample size) are in the shortest time categories (with 26.3 of employees not disclosing figures at all in the United Kingdom). Other countries show sizeable percentages working more than 30 hours in particular the Netherlands (36.7) and Italy (25.9).

For female employees the concentration into the shortest time category is more extreme than for male employees in France, Netherlands (very slightly) and Denmark, with the reverse pattern observed in the other countries, the most extreme difference being observed in Belgium.

· · · ·

. . . ?

	•								
	MALE		1.	I	j.				0/
ters	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	Lux	u.K	TRELA	DERMINEK
s24	61.8	61.8	45.8	65.1	90.5	83:6	6416	96.3	56.6
724-26	15.5	6.7	4.3	4.0	7.1				10.3
726-28	4.3	2.6	3,0	,	•	0			
723-30	8.5	7.7		8.9	•		•		8.5
230-37	10.3	8.8	13.1	16.2	•	•	,		12.4
>32-34	3.1	25	10.8	16.6	U	3	••••••••••••••••••••••••••••••••••••••	-	
734-36	4.6	7.9	2.0	3.9		-			•
236	1.2	1	,	-		-	-		
undescroses	FEMAL	= (9,4)	(28.7)				26.3		15:0
1	1	-							°/a
KRS.	GERMANY	FRANCE	IGHLY	NETHS	BELGIUM	LUX	U.K	IRELAND	DENMARK
\$24	57.2	66.1	45.1	66.3	67.6	61.9	56,5	8512	61.7
>24-26	17.8	6.8	1.8	5.3	13.0		12.1	•	9.8
726-28	4.5	2.8	3.2	3.5	. 9,3	25,9	6.4	\$	6.2
7.28-30	8.6	9.7	6.4	7.0	3.8	6:1	12.5	-	7.7
7/30-32	12.0	11.0	8.4	14.1	8.8	8.1	125	-	12.2
>.32-34	5.6	5,1	3,4	10.7		•	-	-	5.5
734-36	3.6	61	3.0	. 6.2	5-4-1	U			2.1
7.36	1.1	,	0.9	-			-	•	•
MUDUSCLOED	(014)	(2.7)	(36.2)				12,5		4,5
t	MALE	AND P	EMALE				· ·		01
:							•		10
fees.	GERMINY	FRANCE	ITALY	NETHS.	BELGUM	LUX	U.K.	TRELAND	DENMICK
524	57.8	64,4	45.3	66.1	74.1	69.7		88,7	61.2
7.24-26	17,5	6.8	214	5,1	11.3	3.6		•	9.8
226-28	46	2,8	3.1	3:0	6.8	17.9		•	5.9
> 28-30	8,6	8,9	5.7	7.3	6.7	5.9		-	7.8
7.30-32	11.8	10.2	9.5	14.5	6.8	7,5		-	12.2
7:32-34	5.3	4.1	511	11.8	•	•		-	5.4
7.34-36	3.7	6.8	2:8	5,8	-	•	1	-	2.0
. 7.36	1.1	0.5							•
UNA SCLOSED	(0,4)	(5.3)	(34,4)				,	-	5.5

SOURCE TABLE AS COUNTRY VOLUMES

75

The Netherlands, as in the case of male employees, has proportionately more serving more than 30 hours than elsewhere.

2,9.2 Age

The average ages are given below:

Mold d

	1100-00-00	t Guidto	
Germany	52.4	37.9	
France	43.9	41.6	
Italy	39.3	35.3	
Netherlands	37.5	34.0	
Belgium	45.1	37.4	
Luxembourg	43.2	35.7	
United Kingdom	59.1	42.1	
Ireland		• - /	Not auoted
Denmark	42.9	36.9	•

As might have been expected, the average age for part-time employees, both male and female, is higher in all cases quoted than it was for full-time employees. However, whereas in the case of full-time male employees, the average ages were very much of the same order, those for part-time employees are not with over twenty years difference between that for the Netherlands (37.5 years) and the United Kingdom (59.1 years). Part-time female employees are younger on average than their male counterparts and show variations from country to country but. less marked than for male part-time employees.

As was the case for full-time female employees, part-time employees in the Netherlands tend to be younger than elsewhere.

The frequency distributions given in Table 36 confirm these differences.

MALE		1	•	•				*	
AGE	GERMANY	FRANCE	JTALY	NETHERLANDS	BELGIUM	LUXENBOURG	UK	IRELAND	DENMARK
21 21-29 30-44 45-54	2.9 8.5 16.7 10.5 . 60.1	4.9 13.3 27.5 20.8 31.1	7.3 15.4 38.3 17.6 21.4	18.3 17.6 19.1 21.1 23.4	3.5 13.7 20.9 22.5 39.4	15•5 31•8 17•3 30•9	76.8	58.8	15.4 15.7 18.9 10.7 39.4
	100.0	100.0	100.0	100.0	100.0	100.0			100.0
WINGERS DOD'S	⁵⁶⁰ (1.3) 12.6	(2.4) 9.2	1.7	3,2	1,2	0.1	15,8	0,1	1.3

. TABLE 36 AGE DISTRIBUTION PART-TIME EMPLOYEES

FEMALE									
AGE	GERMANY	FRANCE	ITALY	NETHERLANDS	BELGIUM	LUXEMBOURG	UK	TRELAND	DEIMARK
 ≥11 21-29 30-44 45-54 ≥55 	1.7 17.2 47.0 23.5 10.4	4.7 13.0 33.4 25.3 21.4	6.3 25.9 37.2 22.6 8.0	7.7 27.6 39.0 17.4 8.1	3.5 24.4 35.0 24.4 12.7	26.9 41.6 20.3 8.1	8.9 42.9 26.8 19.8	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	2.4 23.9 42.0 19.7 11.9
	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.ð	100.0
WALSCLOSE NUMBERS 000's	(0.2) 95.9	(2.2) 4.7	5.4	14.0	2.9	0.2	50,1	0,3	12.3

SOURCE TABLE A3 COUNTRY VOLUMES

9.3 Length of Service

The percentages obtained are shown in Table 37 and a comparison between these and those for full-time employees (Table 20) reveals some interesting features.

For male employees, the concentration into the shortest length of service category for part-time employees is more pronounced in all countries apart from the United Kingdom than for full-time male employees. In the United Kingdom, over the two longest length of service categories the percentages are higher than elsewhere. Germany has a similar if less pronounced characteristic, with Denmark most noticeably higher for the 20 years or more category.

For female employees, there are both some noticeable differences and agreements between the percentages for full-time and part-time employees. For Germany, the Netherlands, the United Kingdom, Luxembourg and to a lesser extent in Belgium the patterns are similar. In Italy and Ireland the concentration into the shortest length of service category is much more pronounced than for full-time employees with a similar, if less pronounced effect in France. Denmark exhibits a different

SB TABLE 37

LENGTH DF SERVICE

MALE

									10
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BE'.GIUM	LUXEM- BOURG	, UK	IRELAND	DEMMARK
< 2 2 - 4 5 - 9 10 - 19 ≥ 20 (*)	35.3 25.2 17.9 10.8 10.8	35.9 25.4 15.9 11.3 6.7 4.7	40.2 31.1 11.4 13.7 3.6	52.8 22.4 11.6 7.1 5.6	44.7 26.0 10.6 9.8 8.9	42.7 29,1 11.8 9,1	25,3 28,3 16,2 15,2 13,1	61.3	40.1 23.6 11.2 14.9
ALL	100	100	100	100	100	100 .	100		

01

9

FEMALE

	I LINALE									· %
	LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DETIMARK
	< 2.	34.5	39,1	59.0	48.5	41.4	51.3	38.7	45.0	25,9
	2 - 4	31.6	26.8	19.7	30.5	30.1	23.9	25.9	15.7	33.3
	5 - 9	20.5	17.2	13.1	12.9	14.0	15.7	17.6	20.8	22.5
	10 - 19	10.8	9.7	7.6	6.1	8.0	6.6	11.2	•	12.1
	> 20	2.6	3.7	0	1.4	6.6	Ð	5.1		3.6
,	(*)		3.5		0.5					2.7
	ALL	100	100	100	100	(00)	100	100	100	100

MALE AND FEMALE

		•			•				10
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 ≥ 20 (*)	34.6 30.9 20.2 10.8 3.5	37.9 26.3 16.7 10.3 4.9 3.9	54.5 22.4 12.7 9.1 1.3	49.3 29.0 12.7 6.3 2.2 0.5	42.3 28.9 13.0 8.5 7.2	48,2 25.7 14,3 7,5 4,2	35:4 26:5 17:2 12:1 7:0	50,1 16,0 .18,9	27.3 32.3 21.4 11.6 4.7 2.7
ALL	100	100	100	(00)	100	[00]	100	100	100
SOURCE	TAE	ILE A4		COUNȚRY	VOLUMES		79-		

TABLE 38

DECUPATIONAL QUALIFICATION GROUP

•• •••

MALE	
111111	

	GERMANY	FRANCE	ITALY	NETHER- Lands	DELCIUN	LUXEN- BOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	- 4.4 21.8 31.0 42.9	8.2 4.6 13.8 7.0 24.3 42.1	2.1 5.0 8.5 35.5 47.5	1.8 3.2 5.4 16.0 51.8 21.8	4,7 13.7 20.9 53.4	, 15,5 38,2 35,5	11.1 31.3 41.4	• 	8.8 36.5 47.8
ALL	100	100	100	100	100	100	100		

FEMALE

	GERMANY	FRANCE	ITALY	MATHER LAMDS	BFLGIUM	LUXEN- BOURG	UK	IRELAND	DESMARX
1A 1B 2 3 4 5 (*)	- 1.6 22.9 39.9 35.6	0,4 0,9 2,1 7,3 22,3 67.0	1.3 10.4 20.5 67.5	2.3 4.1 24.9 54.8 13.6	2.6 30.0 21.1 45.4	9.6 51.3 32.5	- 3.5 62.6 31.0	- 45,0 55,0	4.5 54.5 35.4
ALL	100	100.0	100	100	100	100	100	100	(00

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	netuer- Lands	Belgium	LUXES- BOURG	UK	IRELAND	DENELARK
1A 1B 2 3 4 5 (*)	- 1,9 22,7 38,9 36,4	3.4 2.3 6.6 7.2 23:1 57.4	• 2,2 9,9 24,1 62,7	0.6 2.4 4.3 23.3 54.2 15.1	3.2 26.8 21.1 4.7.7	• 4.2 11.7 46.6 33.6	3,4 3,6 3,9 55,1 33,5	- - 41:2 57.6	914 5218 3616
anten internet and and a state of the state	100	100	100	100	100	100	100	100	100

%

pattern with a shift towards the longer length of service categories.

7.9.4

Occupational Qualification Group

The distribution of part-time employees over the occupational groups is given in Table 33. In all countries the concentration of part-time employees is into the lower occupational qualification groups with the concentration greater than in the case of full-time employees. One feature which is particularly noticeable for the United Kingdom (and to a less extent in France) is the percentage falling into category 1A. (An overall feature of this table is the number of cells which have no entries - because of problems with small sample size7.

Indices

Eccause of the problems of small sample size the main interest lies in the indices for female employees in categories 3, 4 and 5 since for most of the other categories, data has either not been collected or is subject to large standard errors. This applies both to the figures for mean gross monthly earnings and mean gross annual earnings (of Tables 24 and 40).

TABLE 37 INDEX OF MEAN GROSS MONTHLY EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP

MALE

:

	GERMANY	FRANCE	ITALY	Nether- Lands	BELCIUM	LUXEN- BOURG	UK	IRELAND	ITEVMARK
1Å 1B 2 3 4 5 (*)	 179.9# 119.8 94.9 84.7	300,5# 232,2# 135;6# 120,7# 59,1 47,6	130,8# 95.6 84.5		99.5# 82.0# 75.6	126.6 85.2 70.9	63.2# 56.2#		90.1
ALL	100	100	100 -			100	•		100

FEMALE

: LUXEM-NETHER-TRALATO DARMARX GERHANY FRANCE ITALY DELOTIN ПK. BOURG LANDS 18 ۰ ٠ ٠ . . 307.9# . • Ó **1**B 132.4# 129.0# 5 ø 213,3# à 158.2 2 127.9 109.9 152.0 . 151.6 125,3 127.2 3 101.4 100.8 94.9 91.4 105.8 123.5 102.9 102.0 4 79.4 72.4 81.4 77.4 91.1 77.2. (*) 100 100 100 ». * 100 100 100 ALL 100

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUN	LUXEM- BOURG	UK	IRELAND	DETTIALK
1A 1B 2 3 4 5 (*)	- 164,3 126.4 101.3 78,2	399.8# 300.3# 181.7# 134.1 89.9 61.5	153.2# 123.1 101.9 88.5		119,4# !06,1 90.7 86,1	168.9 142.2 94.8 71.8	• • • • • • • • • • • • • • • • • • •		127.5 104.3 80.2
· · · · · · · · · · · · · · · · · · ·	100	100	100	and construction of the states	100	100	100	N.	100

INDEX OF MEAN GROSS ANNUAL EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 40

.i

MALE

	OEHMANY	FRANCE	ITALY	NETHER- LANDS	DELGIUM	LUXEN- BOURG	UK	IRELAND	DENMARK
1Å 1B 2 3 4 5 (*)	- 180,2# 120.2 94,3 80.6	276.1# 197.2# 127.9# 106.9# 49.2 38.9	94.4 86.3		95,1# 70,7# 74,2#	• • 73,2 73,7	.61.4# 48.7#		- 91.3
ALL	100	100	100	-	•	100	•		100

FEMALE

	OEKMANT	FRANCÉ	ITALY	nether- Lands	BELGIUM	LUXEN- BOURG	118	TRELATIN	DEMALRY
1A 1B 2 3 4 5 (*)	- 164.9 127.5 101.4 75.3	322.3# 242.1# 151.6 126.6 72.9	• 119.1 99.1 92.6		137.3# 108.7 98.6 88.1	130,3 106,0 75,0	- 99,3 75.0		128,5 105.7 80.2
ALL	100	100	100		103.0	100.	100		100

MALE AND FEMALE

.

,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	- 171.0 126.7 100.7 76.0	397.3# 280.4# 187.0# 126.6 83.5 53.1	148,3# 116,9 99,5 90,0		129.9# 104.6 91.2 84.0	161.8 127.2 95.6 74.6	• 89.0 66.5		128.3 104,5 78.7
	100	100	100	art 20 De 7. jan andri frederen v	100	100	100#		100

The most variation over categories 3, 4 and 5 is shown in Luxembourg and France, the least in Belgium, with Germany, Italy and Denmark having similar patterns.

2.9.5 Size of Undertaking

The distribution of employees over differently sized undertakings shown in Table . In general part-time employees are concentrated into the smallest size of undertaking category with the degree of concentration varying from one country to country. However for female employees in Ireland there is a much more even distribution (where figures have been disclosed) over the different sizes of undertaking. In the United Kingdom the percentages in the largest size of undertaking category for both male and female employees is noticeably higher than elsewhere (as was noted for full-time employees).

MALE									. %
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	CENMARK
10-49 50-99 100-199	46.4 16.5 13.6	62.6 13.4 6.9	81.4 5.7 8.2	59.0 15:5 11.6 9.2	61.5 10.5 9.1	67.3 20.0 1010	49,5	•	50.4 20.1 8.7 17.1
200-499 500-999 ≥1000 (*)	5.9 628	5:4 5:5	2.6	4 + 2	17,5		10.1	-	-
ALL	100	100	100	100	100	100	100	100	100

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	37.6 17.7 15.3 13.7 7.1 8.5	62.7 12.4 9.1 6.5 3.5 5.8	72.8 9.4 12.3 0.6 1.2 3.8	51.0 17.9 13.7 9.5	52.7 20.0 9.5 3.9 2.2 11.3	48:2 14:7 12:2 24:9	34.2 11.5 11.8 13.1 8.6 20.8	24.0 20,8 20,6	36,5 17,2 16,8 14,5 6,0 9,0
ALI,	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

501	IRCE		TA	BLE A?		C	DUNTRY VO	LUMES	
ALL	100	100	100	100	100	100	100	(00)	(00
EMPS 10-49 50-99 100-199 200-499 500-999 >1000 (*)	38.6 17.6 15.1 13.4 7.0 8.3	62.7 12.8 8.2 6.4 4.2 5.6	749 8.5 11.3 0.8 1.0 3.5	52.5 17.4 13.3 9.5	55.2 17:2 9.4 2.9 1.8 13.4	55:0 16:6 11:4 16.9	37.9 10.9 11.9 13.1 8.0 18.2	24.5 10.7 20.3 21.0 14.2	37.9 17.5 16.0 14.7 5.4 8.5
NO. OF	GERMANY	FRANCE	ITALY	NETHER-	BELGIUM	LUXEM- BOURG	_ UK	IRELAND	DENMARK

SOURCE

TABLE A2

....

Chapter 3: Retail Distribution NACE 64/653-656

This chapter covers the structure of earnings in the Retail Distribution sector and follows the same pattern as that for the Wholesale Distribution sector which was discussed in the previous chapter. The terms and definitions used are those which are to be found in Chapter 1.

3.1 Some Overall Features

Some important characteristics of the labour force are given in Table 42 - number of employees, (also full-time and part-time separately) number of part-time employees number of full-time employees and number of part-time female employees. These are also given as percentages of all employees.

				•			
		•		N	- "	***	
TABLE	42	EMPLOY	TEES	000	13		
							. •

andele Ballander Baller er 2000er af 1944	GERMANY	FRANCE	ITALY	NETHER- •LANDS	BELGIUN	LUXEM- BOURG	UK	TRELAND	DIMMARK
FULL-TIME AND FART-TIME	799.1	471.8	247.2	201.0	124.6	3.1	1466.7	26.7	67.0
PULL-TIE	533.2	418.6	234.1	133.8	103.4	2.7	1059.7	25.6	35.1
PART-TINS	265.8	52.9	13.2	67.2	21.1	0.4	407.0	1.0	31.8
TJLL-TINE PEMALE	302.8	231.7	112.0	60.1	56.4	1.9	551.8	12.9	15.2
PART-TIME FEMALE	253.6	42.6	11.6	51.7	20.0	0.3	368.3	0.8	28.6

86

ţ

FULL-TIME AND PART-TIME = 100

	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXINA- BOURG	UK	IRELAND	DEITLAIK
JULL-TIME AND PART-TIME	100	100	100	100	100	100	100	100	100
FULL-TIME	66.7	88.8	94.7	66.6	83.1	87.1	72.3	96.3	52.5
PART-TIME	33.3	11.2	5.3	33.4	16.9	12.9	27.7	3.7	47.5
FULL-TIME FEMALE	37.9	49.1	45•3	29.9	45.3	61.3	37.6	48.3	22.7
PART-TINE FEMALE	31.7	9.0	4.7	25.7	16.0	9.7	25.1	3.0	42.7

CURCE

TABLE A1

COUNTRY VOLUMES

Perhaps the most interesting feature in the number of employees is the size of retail distribution in the United Kingdom which is almost double the next largest which arises in Germany.

There are also noticeable variations in the percentages of part-time employees. At one extreme are Ireland and Italy with percentages of 3.7 and 5.3 respectively in this category and at the other extreme is Denmark with a percentage of 47.5. The other countries fall into two groups - Germany, Netherlands and the United Kingdom with percentages in the range 27-34 and France, Belgium and Luxembourg with percentages in the range 11-17. The percentages for full-time female employees also show considerable variations from 22.7 for Denmark to 61.3 for Luxembourg (though this latter percentage is based on very small numbers). The figure for Denmark is of particular interest indicating that the size of the full-time female workforce is just over half that of the part time female worktorce. In all countries apart from Netherlands with a percentage of 29.9, the percentages full-time female employees fall in the range 37-50%.

When full-time and part-time employees are considered together the female workforce constitutes between 50% (Italy) and 72% (Luxembourg) of the total workforce in each country.

3.2 <u>Composition of the Retail Distribution Sector</u>

As was the case in the Wholesale Distribution sector part of the variation in the number of female employees and part-time employees in the labour force from one country to another may be related to variations in the composition of the sector.

The figures are given in Table 43(a) and (b). In all countries apart from Luxembourg, the largest subsector is either Food, Drink and Tobacco (641/642) or <u>Retail</u> <u>Distribution of a wide range of goods with non-food</u> <u>goods predominating</u> (656). <u>Food, Drinks and Tobacco</u> (641/642) is the largest subsector in Germany, France, Netherlands, Belgium and Denmark. In France, Netherlands and Belgium it is considerably larger than the next largest subsector which is 656, mentioned above. In Italy, sector 641/642 is the third largest and is proportionately much smaller than elsewhere.

Retail Distribution of a wide range of goods with non-food goods predominating (656) is the largest subsector in Italy and Ireland and in the United Kingdom (though there is very little difference between the size and the next largest (641/642) in this case).

<u>Clothing</u> (645) is the largest subsector in Luxembourg. The only other subsectors not previously mentioned which appear in the top three are:-

Household equipment, fittings and appliances, hardware and ironmongery (648/649)

which is second largest in Italy and third largest in France, Belgium, Luxembourg and the United Kingdom.

Dispensing Chemists (643) which is third largest in Denmark. In all cases the three largest subsectors in each country employ at least 70% of all persons in the retail distribution sector.



al		NUMB	ERS 000						
NACE CODE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
641/642 643 644 645 646 647 648/649 653 654/655 656	228.2 11.9 10.9 136.6 36.4 12.0 84.7 29.1 27.4 222.0	220.1 4.3 7.7 43.1 10.7 3.8 61.5 16.7 16.9 87.0	44.0 2.8 3.7 29.9 5.9 0.8 54.5 6.6 13.9 85.2	74.4 6.6 2.1 33.7 6.9 3.0 24.6 4.1 6.9 38.5	71.5 2.6 1.1 10.2 2.0 0.4 10.5 4.8 2.3 19.1	0.9 0.0 0.1 1.3 0.1 0.0 0.5 0.1 0.1 -	463.8 89.0 122.2 44.0 8.0 149.9 59.2 64.8 465.8	8.7 0.2 2.7 0.6 1.9 0.5 1.2 10.1	19.8 8.8 0.5 7.5 1.2 1.1 6.5 1.6 1.9 18.1
64/653-656	799.2	471.8	247.2	201.0	124.6	3.1	1466.7	26.7	67.0

TABLE 43: COMPOSITION OF RETAIL DISTRIBUTION SECTOR

(6)

SUB-SECTORS AS % OF WHOLE SECTOR

NACE CODE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMAPK
641/642 643 644 645 646 647 648/649 653 654/655 656	28.6 1.5 1.4 17.1 4.6 1.5 10.6 3.6 3.4 27.8	46.7 0.9 1.6 9.1 2.3 0.8 13.0 3.5 3.5 3.5 18.4	17.8 1.1 1.5 12.1 2.4 0.3 22.0 2.7 5.6 34.5	37.0 3.3 1.0 16.8 3.4 1.5 12.2 2.0 3.4 1 9.2	57.4 2.1 0.9 d.2 1.6 0.3 8.4 3.8 1.8 15.3	29.0 0.0 3.2 41.9 3.2 0.0 16.1 3.2 3.2 -	31.6 6.1 8.3 3.0 0.5 10.2 4.0 4.4 31.8	32.6 0.7 10.1 2.6 2.2 7.1 1.9 4.5 37.8	29.6 13.1 0.7 11.2 1.8 1.6 9.7 2.4 2.8 27.0
64/ 653-656	100	100	100	100	100	100	100	100	100

SOURCE TABLE A1 COUNTRY VOLUMES'

3.3 Size of Undertaking

Another important point to be considered is the size of undertaking in which the employee is employed. The distribution of employees by size of undertaking is given below in Table 44

ď

TABLE 44; EMPLOYEES BY SIZE OF UNDERTAKING (FULL-TIME AND PART-TIME EMPLOYEES)

					•				~~~~
SIZE OF UNDELTAKIK (EMPLOYEES)	W GERMANY	FRANCE	ITÅLY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
10-49	31.8	28.1	46.5	33.4	22.8	57•3	13.8	25.2	40.3
50-99	7-4	9•5	7.1	8.9	6.8	14.5	4.3	10.4	9.6
100-199	7.8	6.9	6.3	6.3	3.5	17.3	4-1	11.4	5.6
200-499	8.4	10.2	5.5	6.8	7.1	11.0	7.9	13.5	3.2
500-999	6.5	6.6	1.9	4.1	3.7	-	7.0	2.2.6	6.4
≥ 1000	38.1	38.7	32.8	40.4	56.2	-	62.9	17.0	34.9
	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
NUMBERS	799-2	471.8	247.2	201.0	124.6	3.1	1466.7	- 2.6.7	67.0

SOURCE TABLE A2 COUNTRY VOLUMES

. ..

-91

Apart from Ireland and Luxembourg, all other countries show patterns where between 65 and 80 per cent of employees are distributed between the two size categories, 10-49 employees and 1,000 employees or more. The distribution of employees between these two categories varies from one country to another but the United Kingdom exhibits the pattern with the greatest concentration in the largest undertakings, and the least concentration in the smallest. In Luxembourg, where the sector is numerically small and has different characteristics in its composition, the concentration of employees is in the smallest size category. In Ireland, just over a quarter of employees are in the smallest size of undertaking category and just under 40 per cent are distributed between the two largest size categories, somewhat as in wholesale distribution. However, the very large undertaking is everywhere much more common in retail distribution: 33-63% of employees across all countries (apart from Ireland and Luxembourg), compared with 2%-20% in wholesale distribution.

3.4 <u>Distribution of the labour force according to gross</u> monthly pay

As has been mentioned in the previous chapter, one of the main aims of the survey was to examine the structure of earnings in the labour force analysed by various characteristics. This section considers the distribution of employees in the Retail Distribution sector according to gross monthly pay and uses the same notation and methodology as was outlined in the previous chapter.

The graphs which have been constructed from the data are given in Figure 2. The table below gives the percentages of full-time female employees in the full-time labour force and the graphs reflect these percentages, which are higher than the corresponding entries for the Wholesale Distribution sector.

Full-time female employees as a % of all full-time employees:

Germany	56.8
France	55.4
Italy	47.9
Netherlands	44.9
Belgium	54.5
Luxemboura	73.14
United Kingdom	52.1
Ireland	50.4
Denmark	43.3

SOURCE TABLE A1 COUNTRY VOLUMES

*[based on a figure of 700 male employees, 1900 female employees, and a total of 2600 employees (Vol 2 gives a rounded figure of 2700)]. As an aside, perhaps one of the interesting features which this table reveals is how similar the percentages are in five of the countries - Germany, France, Belgium, the United Kingdom and Ireland. Luxembourg is noticeably high (and is based on small numbers) and Denmark, the Netherlands and Italy noticeably lower than elsewhere. In Denmark, however, it should be remembered that the percentage of part-time female employees is particularly high and that in the Netherlands and in Italy the underlying activity patterns (- and possibly in Italy, the coverage of the survey) are different.

When we return to the graphs, we notice that in general the graphs are unimodal (female employees in the Netherlands being an exception) and that they tend to be positively skewed. The variation in earnings which is shown in the graphs and (perhaps more clearly) in the table of coefficients of variation given below is in all cases apart from Belgium greater for male employees than for female employees.

Table 46 Coefficients of Variation (%)

Gross Monthly	Earnings	OT FULL	-time male	e and	temale	employees	

	Male Employees	Female Employees
Germany	37.3	35.0
France	64.8	45.5
Italy	42.7	30.2
Netherlands	93.5	42.3
Belgium	46.5	50.3
Luxembourg	49.4	38.5
United Kingdom	73.5	35.2
Ireland	49.7	42.9
Denmark	41.5	27.7

[For male employees the figures for the Netherlands and the United Kingdom are noticeably higher than elsewhere - and a closer examination of Table Cl in the volumes for each country reveals extremely high coefficients of variation in subsector 653 in the Netherlands, and subsector 654/655 in the United Kingdom]. In general the coefficients of variation are very similar to those in the Wholesale Distribution sector with the most noticeable difference (apart from those mentioned above) occurring for female employees in Belgium (coefficient of variation in Wholesale Distribution = 38.5%)

As was also noted in the Wholesale Distribution sector, it is noticeable that the earnings of female employees are in all countries in the lower part of the earnings distribution.

Graphs These have been constructed from the frequency distributions of gross monthly earnings for full-time male and female . employees in each country.

<u>Relative Deviation</u>, $u = (x-\bar{x})/\bar{x}$

where x denotes earnings in the national currency \bar{x}_{z} mean gross monthly earnings (male and female employees combined) $\Delta x = class$ interval in x Δu= Δx/g

Scaled Frequency, $Y = t/\Delta u$ where f is the percentage frequency. The values of Y have been scaled such that the area between the graph and the horizontal axis in each case is proportional to the number of full-time male/female employees.

FEMALE MALE

RETAIL GERMANY.



RETAIL FRANCE



1.5

2.0

-

.0




.







2.



.

- 101 -

2.0

3.5 Gross Monthly Earnings as a function of the structural characteristics of the labour force full-time employees

> This section of the report considers the effects which the various characteristics of the labour force have had on the distribution of gross monthly earnings around the mean value in each case. The characteristics to be considered are as follows:-

- a. sex
- b. age
- c. length of service
- d. occupational qualification group
- e. size of undertaking

The notation, and definitions of the various indices to be considered are the same as for Wholesale Distribution.

3.5.1 Employees classified by sex

The graphs and tables of the previous section illustrated the position of the earnings of full-time male and female employees. Table 47 below gives a summary of these characteristics.

> ينون ويوني روز بالجمعة

Table 4

Summary of Pemale/Male Mean Gross Monthly Earnings; (Full-time, employees, earnings of full-time male employees = 100)

Germany	66.4
France	66.1
Italy	85.8
Netherlands	61.2
Belgium	71.1
Inxembourg	56.3
United Kingdom	55.2
Treland	58.5
Denmark	72.6

As was the case in Wholesale Distribution, it is in Italy that the difference is smallest and in Ireland, Luxembourg and the United Kingdom is largest. Also, in general, the figures for both the Wholesale and Retail sectors are very similar with the more noticeable differences occurring in Italy and Belgium. These are global figures, however, and we shall in the following sections consider the effects which the various characteristics of the labour force have upon them.

3.5.2 Employees classified by age

The average age in years is given below for full-time male and female employees:-

	Male Employees	Female Employees
Germany	34.7	31.8
France	34.2	32.2
Italy	33.9	26.7
Netherlands	31.3	23.5
Balgium	36.0	31.2
Luxembourg	32.2	26.1
United Kingdom	35.6	32.5
Treland	33.8	27.5
Denmark	35.2	31.0

SOURCE TABLE A6 COUNTRY VOLUMES

The figures reveal an effect similar to that observed in the Wholesale Distribution sector. The averages for male employees are very much of the same order (when rounded - all apart from Luxembourg and the Netherlands falling in the range 34-36 years) whereas the averages for female employees show greater variation, with that for the Netherlands lower than elsewhere. Luxembourg, Italy and Ireland also form a group of "younger" averages. The frequency distributions which are shown below in Table 48 serve to confirm these different age patterns.

MALE			· · ·		· · · · · · · · · · · · · · · · · · ·				70
AGE (YRS)	GERMANY	FRANCE	ITALY	NETHS	BELGIUN	LUX	UK	IRELAND	DENMARK
< 21 21-29 30-44 45-54 ≥55	7.0 26.8 38.8 15.9 11.3	8.7 30.8 31.6 18.3 9.8	5.4 28.8 41.6 18.5 5.8	15.8 33.2 30.1 12.0 8.7	8.2 25.2 32.3 22.2 12.0	19.0 25.7 30.3 15.1 9.9	14.3 24.4 26.1 17.6 17.6	15.2 25.7 29.7 16.4 11.6	5•6 32•3 32•3 15•3 14•6
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
Undisclosed	[<u></u>	I	(0.3)	L	L	 	1.3	.

TABLE 42 : AGE DISTRIBUTION - FUIL-TIME EMPLOYEES

FEMALE % UK IRELAND **DENHARK** LUX NETHS BELGIUM AGE (YRS) GERMANY FRANCE TTALY 33.1 26.8 13.8 38.1-47.0 17-7 21.6 16.0 13.6 < 21 33.3 42.6 29.2 20.3 28.6 51.7 31.0 27.8 31.2 21-29 14.2 19.3 21.2 19.1 11.1 31.0 30-44. 26.7 27.2 23-1 9.4 12.6 8.6 21.3 7.2. 18.9 45-54 17.4 17.5 6.9 11.7 3.8 4.9 10.5 3.4 0.9 10.1 7.9. 2 55 (100.0)(100.0) (100.0) (100.0) (100.0) (100.) (100.0)(100.0)(100.0)

Undisclosed

(0.3)

SOURCE TABLE AS COUNTRY VOLUMES

-1044 1

1.5

For full-time male employees, there are various patterns. In Italy the youngest and oldest age categories are proportionately smaller than in other countries with a concentration in the 30-44 years of age category. Denmark also has a proportionately small under 21 years of age category but a proportionately large (14.6%) oldest age category. In Luxembourg and the Netherlands the youngest age category is proportionately larger than elsewhere and in the United Kingdom the oldest age category occupies the same position.

Perhaps the most interesting features occur in the age distributions for female employees. In Iuxembourg and the Netherlands there is a pronounced concentration in the under 21 years of age category (particularly in the Netherlands) and correspondingly few in the older age groups. Ireland has a similar if slightly less extreme distribution. In Italy the concentration is in the 21-29 years category with correspondingly low percentages in the youngest and oldest (in particular) The United Kingdom is interesting age groups. in the fact that over 30% of female employees are aged over 45 with the position in Germany somewhat similar.

AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXFM- BOURG	UX	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54 ≥ 55 (*)	56.7 90.9 112.2 105.7 95.9 79.1 #	56.6 85.5 114.7 112.4 104.4 97.4	68.1 86.2 107.3 111.1 105.5 	48.6 89.2 121.4 129.5 111.9 83.8 #	59.0 87.8 108.1 112.2 100.5	45.3 97.1 122.4 119.7 112.9	56.5 95.9 116.5 117.7 97.4	53.1 91.9 118.0 116.1 107.1 97.9	58.5 87.2 118.2 110.0 94.0
ALL	100	100	100	100	100	100	100	100	100

TEMALE

MALE

AGE IN YEARS	GERMANY	FRATCE	ITALY	NETHER- LANDS	BELCUIN ·	LUXEM- EOURG	UK	IRELAND	DEIJMAHK
< 21 21 - 29 30 - 44 43 - 54 > 55 (*)	71.2 98.8 115.8 110.7 108.7 99.7 #	77.8 93.7 109.9 107.6 111.9 103.2	83.2 96.6 109.7 116.1 115.6	71.5 114.4 139.9 139.1 144.0 89.0#	61.1 91.4 112.3 120.9 121.0	74.1 105.0 121.6 132.0 129.1	83.3 107.4 107.5 104.9 102.6	75.1 105.2 118.1 123.0 115.6 120.1	79.2 99.6 105.9 109.3 105.2
AJL	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

.

								And the state of the
GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGUIM	LUXEM - BOURG	UK	IRELAND)	DERMARK
60.6	64.8	75.0	54.5	5 6. 1	62.0	64.3	59-1	65,7
94.3	89.0	89.9	97 •7	88.6	100.7	102.2	94.4	· 90•5
119.6	114.7	109.6	136.8	110.4	128,8	117.7	129.0	119.2
107.2	110.7	115.5	139.3	. 117.6	136.8	108.9	126.6	111.3
102.6	109.5	111.5	128.3	114.6	131.4	105.9	113.9	98.8
92.2#	99.0		85.7#				106.0	and a state of the
100	100	100	100	100	100	100	100.0	100
	GERMANY 60.6 94.3 119.6 107.2 102.6 92.2 #	GERMANY FRANCE 60.6 64.8 94.3 89.0 119.6 114.7 107.2 110.7 102.6 109.5 92.2 [#] 99.0 100 100	GERMANY FRANCE ITALY 60.6 64.8 75.0 94.3 89.0 89.9 119.6 114.7 109.6 107.2 110.7 115.5 102.6 109.5 111.5 92.2# 99.0 - 100 100 100	GERMANYFRANCEITALYMETHER-LANDS60.664.875.054.594.389.089.997.7119.6114.7109.6136.8107.2110.7115.5139.3102.6109.5111.5128.392.2#99.0-85.7#100100100100	GERMANYFRANCEITALYNETHER-LANDSBELGUIN60.664.875.054.556.194.389.089.997.788.6119.6114.7109.6136.8110.4107.2110.7115.5139.3117.6102.6109.5111.5128.3114.692.2#99.0-85.7#-100100100100100	GEPMANTFRANCEITALYNETHER-LANDSBELGUIMLUXEM-BOURG60.664.875.054.556.162.094.389.089.997.788.6100.7119.6114.7109.6136.8110.4128.8107.2110.7115.5139.3117.6136.8102.6109.5111.5128.3114.6131.492.2#99.0-85.7#100100100100100100	GERMANTFRANCEITALYNETHER-LANDSBELGUIMLUXEM - BOURGUK60.664.875.054.556.162.064.394.389.089.997.788.6100.7102.2119.6114.7109.6136.8110.4128.8117.7107.2110.7115.5139.3117.6136.8108.9102.6109.5111.5128.3114.6131.4105.992.2#99.0-85.7#100100100100100100100	GERMANYFRANCEITALYNETHER-LANDSBELGUINLUXEM- BOURGUKIRELAND60.664.875.054.556.162.064.359.194.389.089.997.788.6100.7102.294.4119.6114.7109.6136.8110.4128.8117.7129.0107.2110.7115.5139.3117.6136.3108.9126.6102.6109.5111.5128.3114.6131.4105.9113.992.2#99.0-85.7#100100100100100100100100100.0

SOURCE

COUNTRY VOLUMES

....

These patterns reflect the underlying activity patterns for male and female employees as was discussed in the two previous chapters.

Indices of Mean Gross Monthly Earnings -Employees classified by Age and Sex

Three sets of indices are considered. Table 49 considers comparisons between groups of the same sex; Table 50 considers the position of employees under 21 relative to those aged between 21 and 30 years; and Table 51 considers the position between female and male employees.

For male employees the index figures rise from values in the range 45 to 69 (Luxembourg 45.3, Italy 68.1) for the under 21 age group to peak values in the range 111-130. For Germany, France, Luxembourg, Ireland and Denmark the peak values occur in the 30-44 years age group, whereas for Italy, Netherlands, Belgium and the United Kingdom the Peak comes in the 45-54 years age group.

For women the index figures take values in the range 61 to 84 (Belgium 61.1, UK 83.3) for the under 21 years age group. The peak values for Germany and the United Kingdom comes in the 30-44 years age group (115.8 and 107.5 respectively). Those for Italy, Iuxembourg, Ireland and Denmark come in the 45-54 years age group and those for France, Netherlands and Belgium in the 55 years and older group, with the figures for Netherlands noticeably higher than elsewhere.

In general, the values taken by the indices are very similar to those for the same groups in the Wholesale sector though there are noticeable difference. for the oldest age groups for female employees in Luxembourg and Ireland (where the corresponding entries are 147.6 and 125.0 respectively).

Table 50

Mean gross monthly earnings of full-time employees aged under 21 years (Earnings of full-time employees aged 21-29 = 100, male and female employees considered separately.

	Male	Female
Germany	62-4	72.1
France	66.2	83.0
Italy	79.0	86.1
Netherlands	54.5	62.5
Belgium	67.2	66.8
Luxembourg	46.6	70.6
United Kingdom	58.9	77.6
Ireland	57.8	71.4
Denmark	67.1	79.5

The figures reveal a pattern similar to that observed in the Wholesale sector. In all cases apart from Belgium (where there is very little difference at all between the figures quoted for male, and female employees) the change is greater for male employees than for female employees. Once again, Italy shows the least change both for male and female employees and Luxembourg most noticeably the greatest change for male employees. In general (apart from male employees in Luxembourg) the figures are of a very similar order to those in the Wholesale sector. When the effect of age is removed (see Table 51 below) the values of the indices for each of the subgroups are observed to be greater than the ratio of the overall means. This effect, also observed in Wholesale Distribution, reflects the different age distribution patterns for male and female employees.

TABLE 51

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES (BY ACE) (MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY - 100)

AGE IN YEARS	GERMANY	FRANCE	ІТАІ. Т	NETHER- LANDS	RELGIUM	LUXEM- BCURG	UK ·	IRELAND	DENHARK
< 21 21 - 29 30 - 44 45 - 54 ≥ 35 (*)	83.3 72.1 68.5 69.5 75.2 83.7#	90:8 72.4 63.3 63.2 70:8 70:0	104.9 96.2 87.8 89.7 94.1	90.0 78.5 70.5 65.8 78.8 65.0#	73.6 74.0 73.9 76.6 85.6	92.2 60.8 55.9 62.0 6 4.4	81.4 61.8 51.0 49.2 58.2	82.7 66.9 58.5 61.9 63.1 71.7	98,3 82.9 65.1 72.2 81.3
ALL	66.4	66.1	85.8	61.2	71.1	56.3	55.2	58.5	72.6

SOURCE

TABLE D9

COUNTRY VOLUMES

The figure for Italy in the under 21 years of age group is noticeable in that it exceeds 100. By comparison the figure for Belgium for the same age group is low. Italy is consistently high for groups while the United Kingdom is correspondingly low in particular for the 45-54 years of age group. This figure and that for the 55 years and over age group may relfect a "return to work effect in the United Kingdom, as was mentioned in the previous chapter.

Employees classified by length of service 3.53

It is to be expected that there will be a relationship between the length of service distribution and the age distribution as has been discussed earlier.

The figures are given below in Table 52

TABLE 5: LENGTH OF SERVICE DISTRIBUTION - FULL-TIME EMPLOYEES

MALE

Undisclosed		(3.1)		(10.9))		(10.9) 1.4	(2.0)
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
<2 YRS 2-4 5-9 10-19 20	29•5 26•9 19•9 16•4 7•3	31.8 22.8 17.9 16.5 7.9	24.8 27.8 20.9 20.2 6.3	34.1 24.6 17.9 14.0 8.4	26.9 21.5 18.9 20.0 12.6	40.8 27.7 15.4 10.7 5.4	31.2 22.7 16.6 14.8 13.9	16.0 19.8 21.1 20.8 20.9	28.7 26.3 17.7 16.3 8.9
YEARS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK

FEMALE

		2.8		0.7			1.1	1.4	1.3
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
 2 YRS 2- 4 5- 9 10-19 20 	31.2 30.4 20.7 13.0 4.8	32.0 26.1 18.4 14.8 5.9	26.3 29.0 25.2 17.0 2.5	48.0 30.2 13.3 5.5 2.3	26.3 25.1 19.4 18.8 10.4	40.6 32.6 15.1 8.9 2.8	37.6 27.7 18.2 10.5 4.7	27.3 31.2 21.6 11.4 7.1	35.2 33.1 18.4 8.2 3.9
YEARS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	TAX .	UK	IRELAND	DENMARK

Indisclosed

SOURCE TABLE A4 COUNTRY VOLUMES . 444

For male employees, perhaps the most interesting feature is the differing pattern which occurs for Ireland where it is very noticeable that the percentage in the longest length of service category is much higher than elsewhere. In other countries the weight of the distribution is towards the shorter length of service categories the most extreme case occurring in Luxembourg.

For female employees, the high percentages occurring in the shortest length of service categories in Netherlands and Luxembourg are most noticeable and are consistent with the patterns for age distribution observed in the previous section. At the other end of the scale the figure for Belgium in the 20 years or longer length of service category is noticeably higher than elsewhere and in fact 29.2% of female employees have served 10 years or longer there (which is more than 8% higher than the next highest figure recorded for this category (France: 20.7%)).

Indices

Two sets of indices are considered - Table 53 gives comparisons of mean gross monthly earnings of employees of the same sex; Table 54 gives comparisons between male and female employees.

For male employees alone there is a smooth increase in the values taken by the index with increasing length of service in all countries apart from the United Kingdom,

112

кі ti. where the values for the two "longest serving" countries are of very similar order. The value for Luxembourg in the longest length of service category is noticeably higher than elsewhere.

The values for the indices for female employees alone show some interesting variations. Perhaps the most noticeable features are the values which correspond to the longest length of service category in Netherlands and Luxembourg. They are markedly higher than elsewhere and reflect the different underlying age, length of service and activity patterns in these countries. For this category in France and the United Kingdom the figures recorded are lower than elsewhere.

As was noted in the Wholesale Distribution sector, the values taken by the indices for female employees tend to be higher than the corresponding entries for male employees a reflection of the different distributions of length of service between male and female employees.

Apart from the exceptions mentioned above (and a low figure for Belgium corresponding to the shortest length of are service category) the entries of similar order of magnitude.

TABLE 53

INDEX OF MEAN GROSS MONTHLY EARNINGS - BY LENGTH OF SERVICE

MALE

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
< ?.	87.0	78.7	83.8	· 77•4	78.7	75.6	80.3	71.4	83.6
2 - 4	95.9	93.0	92.2	93.9	93.4	97.1	90.7	85.9	92.4
5 - 9	104.7	105.4	106.3	108.3	104.5	123.6	107.2	101.9	106.2
10 - 19	114.2	119.9	115.2	124.0	111.5	131.3	124.4	114.2	121.4
≥ 20 (*)	118.9 —	134.0 112.6	124.0	134.9	123.0	150.4	123.3	116.5 111.3#	130.2 92.8
ALI.	100	100	100	100	100 .	100	100	100.0	100

FEMALE

****	LENGTH OF	OFDHANY	EDTHOE	TTALN	NETHER-		LUXEM-	111/	IDELAND	DENHAD
	(YEARS)	GERMANT	r KF.IJLE	TIALI	LANDS	DELGIUM	BOURG	UN		
	< 2	-88.0	87.0	87.1	84.2	70.5	81.0	88.8	81.9	91.7
:	2 - 4	93-2	95.0	95.2	98.1	90.4	97•3	9 9.0	92.3	97•9
	5 - 9	104.9	102.8	104.6	124.4	106.2	117.0	107.8	108.8	104.4
•	10 - 19	120.5	116.0	116.8	147.1	118.0	137.2	114.1	117.4	114.3
:	> 20	138.2	129.8	138.9	169.8	140.0	177.5	130.6	135.3	143.7
	(*)	•	103.8		176.0	_		90,5	141.7#	20.8
· ·	ALL	100	100	100	100	100	100	100	100	100

HALE AND F	emale		•			·.			
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMAR
< 2	87.0	82.3	84.8	76.9	75.1	78.7	81.7	71.4	85.8
2 - 4	93.3	92.7	93.1	93.1	90.7	95.6	91.2	83.6	93. 0
5 - 9	104.4	104.1	105.1	116.4	105.3	120,0	105.8	103.9	105 •1
10 - 19	119.5	119.5	116.6	139.7	114.9	137.8	127.3	123.8	125.5
≥ 20	131.4	136.1	130.9	155.9	132.1	178.2	142.5	135.0	139.7
(*)	•	110.0	· •		-	-	87.5#	1244#	93.8
ALL	100	100	100	100	100	100	100	100	100

SOURCE

114

COUNTRY VOLUMES

SB

In the comparison between male and female employees, when the effect of length of service is removed, see Table ⁵⁴ below, the values of the indices for most of the subgroups are greater than the value given for the ratio of overall means in each country. Where there is a rise above the overall ratio it tends to correspond to the longer length of service categories. In particular, the effect is to be observed for the longest length of service category in all countries apart from France, though the amount of the increase shows considerable variations - with Netherlands the greatest and the United Kingdom the least.

There is an increase in the ratio for the longest length of service category over the shortest in all countries apart from France and the United Kingdom though only in Belgium is the increase a smooth one. As was observed in the case of Wholesale Distribution the figures for Italy are higher than for elsewhere (particularly in the longest length of service category) whereas those for Luxembourg and the United Kingdom lower than elsewhere.

TABLE 55 .: GRADE STRUCTURE (OCCUPATIONAL QUALIFICATION) -- FULL-TIME EMPLOYEES

MA	LE								. *
GRADE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	TYIX	UK	IRELAND	DENMARK
1A 1B 2 3 4 5	0.5 1.9 14.3 47.7 26.8 8.8	3.6 7.8 18.9 14.4 36.3 19.0	1.1 2.6 5.8 10.9 47.0 32.7	3.5 5.9 12.7 34.6 43.3 -	2.4 4.2 10.4 27.9 40.2 14.9	4.8 9.6 13.1 24.4 30.8 17.3	2.0 3.7 19.9 18.3 47.1 8.9	4.5 10.9 22.9 47.5 14.1	3.5 6.1 14.0 26.0 34.5 15.9
	(100.0)	(100.0)	(100.0)	(100.0)	(100.0)	(100.0)	(100 .0)	(100.0)	(100,0)
NUMBERS	230.4	186.9	122.0	73.8	47.0	0.7	507.8	12.7	19.8

TEMATE

T 11	1.11 F 1.113								
GRADE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	TUX	UK	IRELAND	DENHARK
1A 1B 2 3 4 5	0.1 0.2 3.5 23.8 60.3 12.2	0.5 2.3 8.2 13.5 42.1 33.4	0.1 0.3 1.2 6.6 67.8 24.1	0.4 1.3 4.5 22.5 71.3 -	0.1 1.1 3.8 32.4 49.6 12.9	0.9 4.6 26.2 52.9 15.1	5.1 9.2 74.7 10.6	- 2.1 6.7 80.2 10.6	2.3 2.4 21.8 58.1 15.2
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
NUMBERS	302.8	231.7	112.0	60.1	56.4	1.9	551.8	12.9	15.2

MALE & FEMALE

GRADE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
1A 1B 2 3 4 5	0.2 1.0 8.2 34.1 45.8 10.7	1.9 4.8 13.0 13.9 39.5 27.0	0.6 1.5 3.6 8.8 56.9 28.6	2.1 3.8 9.0 29.2 55.9	1.1 2.5 6.8 30.4 15.3 13.8	1.6 * 3.3 7.0 25.7 *16.8 15.7	1.1 1.9 12.2 13.6 61.5 9.8	2.4 6.5 14.8 64.0 12.4	2.1 4.5 8.9 24.2 44.8 15.6
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
NUMBERS 000's	533.2	418.6	234.1	133.8	103.4	2.7	1059:7	25.6	35.1

%

%

When the figures for male and female employees are considered separately the concentration of female employees into category 4 is most noticeable with proportionately very few reaching the higher job categories. A similar, though less extreme effect, is observed for male employees, with category 4 once again the largest one in all countries apart from Germany where category 3 is in this position.

Indices

The considerations to be noted when interpreting the corresponding set of indices for Wholesale Distribution (cf page 28) apply to these indices also. Table 56 gives comparisons between groups of employees of the same sex - Table 57 the comparisons between male and female employees.

Some of considerations mentioned on page 28 may help to explain the situations in Germany and Denmark for males where the index figure for category 1A is lower than that for 1B. A similar situation occurs for females in these categories in Germany and Belgium.

For the remaining categories (2-5) the values of the indices for women tend to be higher than those for their male counterparts. As was discussed in the case of Wholesale Distribution this may probably be traced to the different distribution of women over the categories with proportionately more in the lower ones and proportionately fewer in the higher ones. For males, the figures for Italy are consistently higher (categories 2-5) than in the corresponding categories for any other country and in France (apart from category 2) consistently lower. The occupational qualification structure (cf Table 55) exhibits different patterns for these countries with

-119-

TABLE 56

MALE

	GERMANY	FRANCE	ITALY	NETHER LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	Denmark
1A 1B 2 3 4 5 (*)	185.5 195.1 136.1 101.1 79.9 69.5	264.5 171.7 121.3 90.8 76.8 61.1	2.14.7 188.6 .162.5 122.4 94.6 75.7	2365 1647 1323 971 735	251.0 (85.0 132.4 97.6 84.8 67.8	196,3 143,6 122,1 100 85,2 56,2	213.7# 114.9 96.6 82.9 72.6	183,3 136,6 108,8 85,6 75,7 -	166.2 169.4 130.6 100.6 81.7 72.7
ALL	100	100	100	(00	(00)	100	100	100	100

FEMALE

	GERMANY	FRANCE	ITALY	NETTHER	BELGIUM	LUXIM- BOUNG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	240.7 248.4 176.5 121.7 89.9 79.9	319.1 211.1 146.5 110.8 91.7 80.4	228,5# 192.6 171.0 132.7 100.8 81.8	253.3# 184.7 157.7 118.1 87.8	212,51 236.0 154.9 100.0 97.5 75.2	244.2 155.0 116.5 91.3 71.4	• 140:6 131.6 93.9 89.1	 205,4# 134,0 95,9 81,4	208.9 146.5 110.9 93.5 83.7
ALL	100	100	100	100	(00	100	100	100	100

MALE AND FEMALE

**

•

	GERMANY	FRANCE	ITALY	NETHER- LAND:3	BELGIUM	LUXEM- BOURG	UK	IRELAND	Denmark
1A 1B 2 3 4 5 (*)	272,4 236.2 162.5 115.0 80.0 72.6	314:1 199.7 138.3 99.8 82:1 68.1	227,5 198,8 170,7 127,3 96,3 78,5	275,3 189,8 150,0 106,9 75,3	294.2 214.8 149.1 97.8 89.7 71.5	268,5 209,1 154,5 109,2 84,2 65,9	2725# 138.9 114.6 82.3 77.1	223.8 169.3 128.5 84.7 81.4	190.0 187.8 145.3 105.3 84.0 76.8
ALL	100	100	100	100	100	100	.100	100	100
SOURCE	n - 1998 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 19 1	TADL	E 02,	- 120 -	COUNTRY	VOUNCES			

SOURCE ,

TABLE 02.

7 J.U

large proportional differences in the highest and lowest categories which may help explain these differences.

When the effect of the occupational qualification structure is removed see Table 57 below, the values taken by the index for each of the individual categories is higher in all cases (apart from category 1A in Belgium, where the figure quoted is of limited reliability) than the value for the ratio of overall means.

This is an effect which has been noted in most cases for other indices and also in the Wholesale Distribution sector and may be traced to the different distributions of male and female employees over the occupational qualification groups.

TABLE 57

INDEX OF MEAN GROSS MONTHLY FARMINGS OF FULL-TIME FIMALE EMPLOYEES - BY OCCUPATIONAL QUALIFICATION GROUP

(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY = 100)

مراجعة محافظة محافظة من محافظة المراجعة المراجعة المراجعة المراجعة الماسة المسترية

· · ·	GERMANY	FRANCE	ітліу	NETHER- LANDS	eslgium	LUXEM- BOURG	UK	IRELAND	Denmark
1★ 18 2 3 4 5 (*)	86.1 84.5 86.1 79.9 74.7 76.4	79.7 81.2 79.8 80.6 79.6 86.9	91,4# 87,7 90,4 93,0 91,5 92,7	65.6# 68.7 73.0 74.5 73.1 -	60.2# 90.7 83.2 72.8 81.8 78.4	95.7 71.4 65.6 60.3 71.5	67,5 75,2 62,5 67.8	- 87.9# 72.0 65.4 65.2	89.6 81.4 80.1 83.1 83.6
	66,4	66.1	85.8	61.2	71.1	56.3	55.2	58.5	72.6

SOURCE

TABLE C2

COUNTRY VOIDATS

À 2A

Once again the values for Italy are higher (apart from category 1B) than elsewhere for all categories. In Germany, Luxembourg and Denmark there is a decrease in the ratio over groups 1 to 4 with an upturn in 5. For Ireland, for groups 2-5 there is a decrease and for France, the Netherlands and Italy (to a lesser extent) there is an increase but the increase is not a smooth one. Belgium and the United Kingdom exhibit irregular patterns.

3.5.5 Employees classified by size of undertaking

The final set of indices which we shall consider in this section is that where the factor of classification is the size of undertaking. We have already considered the distribution of all employees over different sizes of undertakings (cf Section 3.3). Table58, gives the figures for full-time employees alone.

These patterns follow those given in Table 44. The most noticeable difference occurs in the case of Italy with the concentration of male employees into the smallest size of undertaking category accentuated and that for female employees split almost equally between the largest and smallest size categories.

TABLE 58

FULL-TIME EMPLOYEES BY SIZE OF UNDERTAKING

MALE

		•		-					
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	32.6 7.9 7.2 8.6 7.0 36.8	29.9 9,9 5:8 8,9 6,3 39.2	56.5 7.5 6.7 6.1 1.7 21.4	36.5 10,3 7,3 9,2 3,7 33,1	25,4 8,5 4.8 6.9 5.2 49.1	66.9 12.9 15.0 5.2	17,5 4,7 8,2 7,7 57,1	27.9 11.5 12.5 13.1 17.9 17.2	40,9 10,2 7,4 6,1 6,4 29,0
ALL	(00.0	100.0	100.0	100.0	(00.0	100.0.	100.0	100.0	100.0

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	31.7 7.1 7.9 8.2 6.1 39.0	26.0 9.5 7.4 11.6 6.8 28.7	38.7 7.0 6.4 5.3 2.1 40.5	38.7 9.6 7.3 7.0 4.9 32.5	25.9 6.6 3.5 7.7 3.2 53.1	51.7 15.2 19.8 13.3	10:8 4:1 4:0 8:0 6:9 66:3	22.7 9.1 10.3 13.9 26.2 17.7	41.8 10.8 5.5 2.8 6.9 32.1
ALL	100.0	100,0	100.0	100.0	100.0	(00.0	100.0	100.0	(00.0

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANLE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-495 500-999 ≥1000 (*)	32.1 7.5 7.6 8.3 6.5 38.0	27,7 9,7 6,7 10,4 6,6 38,9	48.0 7.3 6.5 5.7 1.9 30.6	37.5 10.0 7.3 8.2 4.2 32.8	25.6 7.5 4.1 7.4 4.1 51.3	55.9 14.6 18.4 11.1 -	14.0 4.5 8.1 7:3 61.9	25.3 10.3 11.4 13.5 22.1 (7.4	41.3 10.5 6.6 4.7 6.6 30.4
···ALL	100.0	(00.0	100.0	10.0.0	100.0	100.0	100.0	100.0	100.0
SOL	JRCE		AT	BLE A2	1	Ċ	OUNTRY VOL	LUMES	

Indices

Two sets of indices are considered:- Table ⁵⁹ gives comparisons between groups of employees of the same sex; Table 60 gives the comparisons between male and female employees.

For males, perhaps the most noticeable features are the figures for Italy which show a marked rise in the index for the largest size categories. Some explanation of this may be found by considering within which subsectors these largest undertakings may be found and in Italy these are restricted to 641/642- Food, Drink and Tobacco, and 656 (wide range of goods, with non food goods predominating). The variation by size of undertaking in other countries is less noticeable with no one pattern applying uniformly across all countries - though Ireland has a pattern similar to that for Italy but less pronounced.

For women, with the exception of Denmark where the index is highest for the smallest size category, there is some slight evidence of an increase in index with increasing size - though the rise is not smooth. Italy, Belgium and Ireland show the greatest values of the index in the case of the largest size of Undertaking.

TABLE 59 INDEX OF MEAN OF MEAN GROSS MONTHLY EARNINGS BY SIZE OF UNDERTAKING

MALE

					Concession of the local division of the loca		the second s		
NU. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
		404 6	87 4	100-1	89.5	103.1	101.2	89.9	98.3
10-49	98,7	101.0	- 100 A	407.4	8,00	87.2	95.0	97.5	106.0
50-99	103.5	100.6	109-1	10.5+1	27.0	0102			405 9
100-199	102.3	97.5	102.8	101.1	102.6	98.3	94.1	94+8	105.0
100-100	103.0	96-4	112.7	97.1	97.5	98.1	103.6	99•5	101.5
200-499	109.0	,		107 7	8 505		98.5	111.7	102.1
500-999	103.3	103.4	123.6	103.7	102.0	-	1 100		00.0
> 10(12)	98.6	99.2	131.2	99.1	104.8	-	100.2	111.0	90.0
≥1000 (*)	,	-		-	-	-	-	- 1	-
()							400	400	100
ALL	100	100	.100	100	100	100	100	1 100	100

.

FEMALE

NO. OF EMPS	GERHANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	97.9 98.8 96.8 98.9 98.0 103.2	99.5 99.7 98.1 95.8 102.0 101.9	88.0 96.3 95.5 107.5 103.3 120.8	97.7 93.7 95.9 98.7 97.1 106.3	77.9 91.3 98.0 92.5 93.9 112.8	96.8 98.3 100.0 114.3	90.3 93.3 93.6 97.7 96.8 103.0	89.2 99.5 91.3 97.9 107.0 111.0	104.3 97.1 94.7 98.2 91.5 98.3
ALL	100	100	100	100	100	100	100	100.0	100

MALE AND FEMALE

÷

5. 14 X 78 . 14

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	UK	IRELAND	DENMARK
		230.4	88 3	98.3	83.7	103.2	105.3	92.0	100.3
10-49	98.0	102+1	407.6	101.2	99.1	92.4	96.5	100.9	102.2
50-99	102.6	100.5	105,0	101.2		96.4	96.3	96.2	104.8
100-199	98.6	95.1	99.8	99.0	04.0	100 6	101-2-5	98.6	106.2
200-499	101.8	93.2	110.8	1 100.7	94.2	100.0	00.5	103.4	97.7
500-999	102.3	102.1	112.8	97.6	103-5	-	99+2	10,944	
>1000	100.2	100.8	123.6	101.8	107+9		99.2	110.9	97+3
(*)	-	-		-			-	-	-
ALL	100	100	100	100	100	100	100	100.0.	. 100
	Į	1					+	+	

SOURCE

COUNTRY VOLUMES

When the effect of size of undertaking is removed as is shown below in Table 60, there is some variation from the values for the ratio of overall means but the variation is less noticeable than for any of the other cases (as was also noted for Wholesale Distribution).

In Germany and France there is little variation with size with a slight increase in the ratio for the largest size category over the smallest. A similar pattern is observed for Netherlands, Belgium, Luxembourg, Ireland and the United Kingdom. For Italy and Denmark there is a decrease in the ratio with increasing size though the decrease is not a smooth one.

TABLE GO

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES - BY SIZE OF UNDER-TAKING TN SAVE CATEGORY = 100)

(NEAN	GROSS	NONTHLY	FARMINGS	C P	FULL-TILE	FALE	EFFIOTE25	TN	SAMS	CALEGOILI		
-------	-------	---------	----------	------------	-----------	------	-----------	----	------	-----------	--	--

NO OF EMP5	GERICANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10 - 49 50 - 99 100 - 199 200 - 499 500 - 999 $\gg 1000$ (*)	65.8 63.3 62.8 63.7 63.0 69.5	64.7 65.4 66.5 65.7 65.1 67.8	86.5 75.7 79.7 81.9 71.8 79.1	59.8 55.6 58.1 62.2 57.3 65.6	61.9 65.1 67.9 67.4 64.9 76.5	52.9 63.5 57.3 65.6	49.3 54.2 54.9 52.1 54.3 56.8	58.1 59.7 56.4 57.6 58.5 58.5	77.1 66.5 65.0 70.3 65.1 72.9
ALL	66.4	66.1	85.8	61.2	71.1	56.3	55.2	58.5	72.6
KAJRCE		TABLE	DS.	CODR	TRY VOLL	es	'+2/2,4239 2+2/965- 2018926269 ₫		

- 126 -

XIRCE

3.6 Gross Annual Earnings

The analysis so far has considered gross monthly earnings for full-time employees, classified by various factors. As in the case of Wholesale Distribution, a similar analysis is possible for gross annual earnings for which a frequency distribution containing 27 class intervals has been drawn up for each country.

Indices have been calculated for mean gross annual earnings in a manner similar to that for mean gross monthly earnings and are available for all full-time male employees, all full-time female employees and for these groups classified by occupational qualification group. Indices are also available to compare male and female earnings for each of these categories. The global figures are given in the summary table below

Table 61

Summary of Female/Male Mean Gross Annual Earnings (Earnings of full-time male employees = 100)

127 :

	Germany
	France
	Italy
	Netherlands
	Belgium
	Luxembourg
	United Kingdom
-	Ireland
	Denmark

65.5	(66.4)
62.0	(66.1)
80.2 60.8	(85.8) (61.2)
68.0	(71.1)
56.0 54.5	(56.3)
58.6	(58.5)
70.8	(72.6)

Figures in brackets are the ratios for mean gross monthly earnings

SOURCE TABLE C4 COUNTRY VOLUMES

The same features as were noted for Wholesale Distribution apply in general to these figures. The annual ratios are in good agreement with the monthly ratios and are in all cases (lower than in the monthly case) - apart from in Ireland where the difference is very slight. Also the greatest difference between annual and monthly ratios occurs in Italy - as was the case in Wholesale Distribution.

Table 62 gives the various indices of mean gross annual earnings by occupational qualification group and these in general in good agreement with the corresponding monthly figures (cf Table 56). France, Italy, Denmark, Ireland and to a lesser extent Luxembourg show some evidence of a widening of differentials of the higher qualification groups over the lower qualification groups as compared with the monthly figures. The remaining countries show a narrowing with the effect most pronounced in Netherlands.

The figures giving the comparisons between female and male mean gross annual earnings are given below in Table 63

20.⁷⁴²

TABLE	6	4
-------	---	---

MALE

	GERMANY	FRANCE	· ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOJRC	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	182.3 191.8 133.9 99.2 78.7 69.2	270,7 164,5 115,9 86,5 72,6 57,7	241.8 230.8 168.0 120.4 89.5 72.2	204,0 164,5 130,7 93,9 73,4	242.9 1760 129.1 95.1 79.7 64.8	197.8 140.1 119.7 91.5 82.8 52.9	230,6# 208,0# (13.6 954 80,5 74,3	178.8 132.6 109.4 84.0 76.3	173.1 160.4 129.2 97.5 80.8 72.3
ALL-	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	L

FEMALE

.

	OERMANY	FRANCE	ITALY	NEATHER- LANDS	BELGIUM	LUXEN- BOURG	1JK	TRELAND	DEIMARK
1A 1B 2 3 4 5 (*)	229,0 248,1 174,5 119,9 88,8 80,3	327.2 213.2 140.6 107.9 89.9 79.0	213.4# 210.5 181.6 135.A 99.0 81.4	230,1# 184,4 152,2 117,4 85.7	244.4 148.6 98.4 95.9 76.1	• 157:2 111:8 89.0 69.4	138.1 129.8 92.7 86.1	190.9# 131.5 95.4 85.5	214.2 149.2 110.8 91.9 81.1
ALL	100.0	100.0	100.0	100.0	100.0	(00.0	100.0	100.0	

MALE AND FEMALE

.

•

	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENMARX			
1A 1E 2 3 4 5 (*)	2.18.4 233.6 160.3 112.8 78.4 72.1	327.3 195.5 133.4 95.5 78.7 64.7	257.2 245.7 179.3 126.6 91.9 75.8	235.4 187.4 146.0 103.0 73.8	290,3 2(0.0 145.7 95,0 85.8 70,2	269,4 204,9 152,7 102,9 81,3 63,6	286.8# 260.9# 135.4 111.0 80.1 74.3	216.5 161.8 127.3 83.5 82.0	199.6 180.8 143.9 103:2 82:0 75:6			
	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0			
-121-												

:

	GERMANY	FRANCE	ITALY	nether Lands	DELCIUM	LUX 751- BOURG	UK	IRELAND	POPLANK
1A 1F 2 3 4 5 (*)	82.3 84.7 85.4 79.2 73.5 76.0	74.9 80:3 75:2 71:3 76.7 84.8	70.7# 73.1 86.6 90.2 88.6 90.4	68.6# 68.1 70.8 76.0 71.0	* 78,3 70,3 81,8 79,8	96.9 73.6 68.4 60.2 73.6	66.3 74.2 62.7 63.1	+ 84.3# 70:4 66.5 65.6	94,5 81.7 80.4 80.5 79,4
ALL	65.5	62.0	80.2	60.8	68.0	56.0	54.5	58.6	70.8

Table 63: MEAN GROSS ANNUAL EARNINGS - FULL-TIME FEMALE EMPLOYEES (EARNINGS OF FULL-TIME MALE EMPLOYEES IN EACH GROUP = 100)

As with the figures in the previous table, these are also, in general, in good agreement with the corresponding monthly ones given in Table 53. The annual ratios tend to be lower than the corresponding monthly ones - though this is not uniform across all countries - with the difference most noticeable in categories 1A and 1B in Italy.

3.7 Length of Scheduled Working Week

The figures which have been obtained for the length of scheduled working week for full-time male and female employees are shown below in Table 64.

TABLE 54 : LENGTH OF SCHEDULED WORKING WEEK - FULL-TIME EMPLOYEES

HOURS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	Lî.X	UK	IRELAND	DENMARK
≤ 36 > 36-36 > 38-40 > 40-42 > 40-42 > 42-44 > 42-44 > 44-46 > 46-48 > 48	0.0 84.6 6.1 3.1 3.6 1.5	· 0.2 0.1 37.7 12.4 11.4 13.5 12.7 11.7	0.9 0.2 89.1 1.3 5.8 0.5 1.3 0.1	1.8 28.0 65.9 1.9 2.2 0.1 0.1	0.1 0.9 63.9 34.3 0.2 0.3 0.3 0.1	60.4 2.9 28.5 1.6 5.2	7.7 13.5 62.9 3.4 3.0 1.5 0.8 0.5	7.9 23.0 50.3 5.5 7.3 1.7 1.6 2.3	3.0 28.1 63.7 0.6 1.9 1.6
	100	100	100	100	100	100	100	(100)	100
Undisclosed	(0.9)	(0.4)	(1.0)						

FEMALE

HOURS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
 \$ 36 \$ 36-38 \$ 38-40 \$ 40-42 \$ 42-44 \$ 44-46 \$ 46-48 \$ 48 	0.2 0.3 92.0 4.3 1.4 1.3 0.4	0.3 0.3 64.2 13.0 9.4 7.5 3.0 2.2	0.5 93.2 0.6 4.1 0.1 0.5	0.5 2.1 30.4 63.5 1.5 1.9	0.2 1.0 69.0 27.1 0.5 0.2 2.0	- 84.2 0.5 14.5 - 0.8 -	20.8 24.1 51.4 1.7 0.3	13.1 38.1 36.4 3.5 8.0 0.6	1.6 6.2 38.2 52.4
	100	100	100	100	100	1,00	100	(100)	100
Undisclosed	(0.1)	(0.2)	(0.9)		:		(1.7)		

SOURCE TABLE AS COUNTRY VOLUMES

Ń

134

M .

Perhaps the most interesting features are the figures for the United Kingdom and Ireland for both male and female employees which indicate that proportionately more employees work a shorter scheduled working week than in other countries - an effect also noted in Wholesale Distribution. In Germany and Italy there is a great concentration in the 38 to 40 hour band both for men and women. There is a similar, if less pronounced, effect for this grouping in Belgium and In the Netherlands and Denmark the model Luxembourg. grouping shifts upwards to the 40 to 42 hour band both for men and women. The figures for France indicate that both for men and for women proportionately more work a longer scheduled (working) week (particularly - longer than 44 hours in the case of male employees) than in other countries.

The patterns observed above are similar, in general, to those in Wholesale Distribution, due no doubt to national agreements and legislation, with some interesting variations such as the Netherlands where in Wholesale Distribution the model value is lower for females and less marked for males. In the United Kingdom also, the scheduled hours for female employees tend to be shorter in Wholesale Distribution.

3.8 Bonus Payments

The figures are given in Table 65 and for both male and female employees exhibit a pattern similar to that for Wholesale Distribution with, if anything, slightly greater concentration in the "most popular" percentage bonus categories. The United Kingdom, Ireland and Denmark are once again noticeable for the (high) figures in the "O per cent" bonus category (the UK also for the figure of 2.1 per cent in the 50 per cent and over category). Italy has a noticeable concentration in the 15 to less than 20 per cent bonus category and Netherlands in the 5 to less than 10 per cent category. The figures for Belgium are interesting for the percentages in the 20 to less than 30 per cent range and particularly the value of 27.4 in this category for female employees.
TABLE 65 BONUS PAYMENTS

MALÉ

(FULL-TIME EMPLOYEES)

2.

¢

5 BOMUS	CERMANY	FRANCE	IT/LY	NETIIS.	BELG.	IJJX.	UK	IRELAND	DERMANK
0 >05 >510 ?1015 215-720 ≥2030 ≥3050 ≥ 50	4,3 43,1 34,3 11.8 3,2 2,0 1,0 0,3	37.2 17.1 29.5 9.9 1.9 1.6 0.8 0.5	- 9.7 18.3 42.0 11.6 6.3 1.5	8.3 8.7 55.8 19.1 3.0 2.5 1.6 0.9	13,3 (1.0 19,6 31.2 8,6 14,3 1.2 0.6	28.5 22.2 36.5 8.4 1.4	68.0 16.5 4.8 3.6 1.4 2.0 1.5 2.1	78.4 11.2 4.3 1.8 1.1 0.7 1.6 1.0	88.0 3.5 2.6 1.4 1.4 0.9 1.1
(4) FENALE & BONUS	100 geruany	LOO FRANCE	(00) Italy	100 Neths.	(<i>00</i> Bei.g.	100 101	(00 / UK	(00) Instand	(OO BEISHAIAK
0) 0-5 2 5-10 2 10-15 2 15-20 2 20-30 2 30-50 2 ⁵⁰ (s)	3.8 34.5 38.4 18.3 3.2 1.3 0.4 0.1	32,1 14,9 28,1 19,7 2,0 1,2 0,5 0,2	- 15.3 15.6 10.4 36.7 10.6 9.7 1.6	11.8 12.9 53.5 18.2 1.8 0.9 0.4 0.3	11.8 12.6 15.0 25.2 5.7 27.4 1.1 0.8	25,4 17.7 38,6 10,3 5,2 2,6	66.7 19.5 7.4 4.6 0.8 0.5 0.5	77.6 14:2 3.9 2.7 0.7	93.0 4.0 1.8
	100	100	100	(00	. 100	100	100	100	100

SOURCE TABLE B1 COUNTRY VOLUMES

.

3.9 Part-time Employees

The final section of this chapter is concerned with the characteristics and earnings of parttime employees. The comments applied to the definition of part-time employees discussed in the previous chapter (and in the Technical Appendix) apply also to these figures.

It is useful to consider separately the figures for part-time employees and those given below in Table 66have also been given earlier in Table 42

Table 66 Part-time Employees

	NUMBER	S (000 *s)	AS A PE OF ALL	RCENTAGE EMPLOYEES
	<u>All part-time</u>	Part-time Female	All part-time	Part-time female
Germany	265.8	253.6	33.3	31.7
France	52.9	42.6	11.2	· 9.0
Italy	13.2	11.6	5.3	4.7
Netherlands	67.2	51.7	33.4	25.7
Belgium	21.1	20.0	16.9	16.0
Intembourg	0.4	0.3	12.9	9.7
United Kingdom	407.0	368.3	27.7	25.1
Ireland	1.0	0.8	3.7	3.0
Denmark	31.8	28,6	47.5	42.7

The above table draws attention to the interesting features which have already received some comment in Chapter 1 of this report. The contribution of parttime employees - particularly part-time female employees - to the labour force varies considerably

from one country to another. Denmark is noticeably higher than elsewhere, Germany and the Netherlands next in order with very similar percentages and then comes the United Kingdom with a percentage figure of similar order. At the other end of the scale Ireland and Italy are noticeably lower than elsewhere. In all cases parttime employees form a greater proportion of the labour force than they did in Wholesale Distribution with the contribution being anything from the order of just over twice as great (Germany, Luxembourg) over to four times as great (Netherlands and Belgium).

3.9.1 Length of Scheduled Working Week

It is of interest to see how this varies from country to country for it constitutes the main factor of classification of part-time employees. The patterns observed are similar to those for Wholesale Distribution. In all countries the greatest concentration of employees (for both men and women) fall into the shortest time category but the degree of concentration varies considerably both from one country to country and between male and female employees.

For male employees, because of the small numbers involved in some countries eg Italy and Ireland, the distribution is not completely specified for reasons of confidentiality and small sample size, (and also in Italy 33.5% of this small number disclosed no information on scheduled length of working week). Where it is (almost) completely specified eg in Germany, France and the Netherlands it is interesting to note the variation in the percentages working 30 hours or more (eg 22.9 in Germany as compared with 10.8 in the Netherlands).

For female employees, more information is available. No doubt the effect of legislation concerning part-time working in the various countries is being reflected in these figures. For example in the United Kingdom 4.7% of part-time female employees are quoted as working 30 hours or more and in Italy the figure is 3.7%. On the other hand the corresponding percentages for Germany and France are 27.3 and 27.5 respectively. The pattern for Belgium is worth a separate comment with its low percentage in the 24 hours or less category and its noticeably high percentage (25.9) in the 26-28 hours category.

M	AL	E

l

:

5

		MALE			i		.	-	<u>.</u>	%
	400	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	И.К.	FRELAN	DENMIRK
	51A	59.4	19.5	49.9	88.6	79.6	92.9	78.9	85.8	76.2
- + -	54-71	10.1	5.7	0	(13	3	8	•	8	7.4
1.	21-70	65	7.4	,	0.5		-	•		•
, ,	73-20	10.0	2.7	y .	2.0	•		ų	4	,
	20 24 27	11.7	8.2	10:4	3.3	-5,5	-	U	* -	•
3	21-24	4.2	2.8	D	7.1	•		-	-	
	31-21	113	3.6	1.7	0.4					1
	726	2.0	0	v						
u.	1. JU	- Pott	- 7.3	33.5		5.1		14,9		5.0
		ELINA	í l	.						To
:	LAAC	r requirery	TOMIT	TIMY	NETHS	RELGIUM	LUX	U.K	IRELING	LENMIKK
	MRKS.	102	10.9	89.5	77,5	48.3	78.9	70.5	89.1	60.1
•	St II	15.1	7.7	0.4	6.5	13.3	4,4	10.1	4	10.5
: _	-14-10	02	2.5	1.1	3.7	25,9	3.2	6.3	•	6.9
	140-40 Star 20	11.2	12.1	1.4-	5,6	5,4	7.3	6.8	ç	6.3
	1/28 <u>-30</u>	1115	17.7	2.7	9.1	6,5	8.2	47	-	9,3
	, 30-3		6.0	0.9	9.0	4.3		~	-	5.9
	7. 2 <u>1-4</u> ~ 174-1		29	0.6	2:6	1.4	4.1		-	4
	KTS X	E <u></u>	- 510	3	-	-		-		2.1
	7:30	1.6	4.6	5.4		•		5.8		4.7
U.	NAZUOS ED		MENT	MART				· ·		01
:		PUALE	ANDY	FOUND					·	10
•	405	C. TON MY	TOMME	TTALY	NETHS.	BELGIUM	LUX	U.K.	FRELAND	DENMACK
	< 74	511	67.6	84.8	76.2	50,0	80.9	71.3	88.3	61.7
	>>4-7	6 14.8	6.9	0.5	5.3	12.8	4,5	9.3	•	10.2
	> 7/-70	0 0.1	2.3	1.3	2.5	24.8	2.8	6:4	•	6.4
	10 10	$\frac{1}{1}$	11.9	7.3	4.7	5.3	6.3	6.4	<u>ى</u>	5.9
	7.20 7	2 14.0	15.0	3.7	2,7	6,5	7.0	4.4	-	8.7
	1.30-5		131 <u>7</u> 21	<u> </u>	Res	4.1	v	-		5.5
• •	1-26-5	$\frac{0.7}{20}$	2.7	0.8	2,1	1.3	3.8	-		4.1
										2.0
	236	A.5	5.1	4,2		0.4		6.6		2.1
i.			U -1	-		- I	•			• •

138

.

3.9.2 Age

The average ages in years of male and female part-time employees are given below:

Germany51France36Italy32Netherlands19Belgium45Luxembourg41United Kingdom54Ireland54Denmark27	4 39 3 35 3 28 3 30 1 34 2 36 7 42 NOT QUOTED 7 38	4355814

These averages show considerable variations both as between employees of the same sex and between male and female employees in the same country. For female employees the averages are consistently higher than for full-time female employees though there are variations. In particular the figure for the United Kingdom is noticeably higher than elsewhere, whereas that for Italy is low. The frequency distributions given in Table 68 illustrate these different characteristics.

> 1. 194 1. 194

The averages given for part-time male employees show much greater variation and an unexpectedly low value appears for the Netherlands (and is confirmed by the frequency distribution). As in the case of Wholesale Distribution, the average for the United Kingdom is higher than elsewhere.

TABLE 68: AGE DISTRIBUTION - PART-TIME EMPLOYEES

MALE

AGE (YRS)	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
< 21 21-29 30-44 #2-54 み 55	5.4 7.9 16.6 9.6 58.8	18.7 21.5 20.9 15.2 21.4	15.5 29.6 32.4 14.3 8.3	80.3 9.2 3.0 2.8 4.4	8.9 10.2 19.9 20.7 40.4	44.6 30.4	12.0 4.1 5.8 6.2 71.9	87.5 -	56•1 13•9 5•6 4:2 20•2
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
(%)	1.7	2.1		deserve served					

FEMALE

AGE (YRS)	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
₹ 21 21-29 30-44 45-54 ≹ 55	1.7 13.3 45.7 26.6 12.6	14.6 20.1 30.5 23.0 10.1	6.3 47.0 39.8 5.6 1.3	32.0 16.4 25.7 17.9 7.5	5•7 24.8 42•2 23•8 3•4	5.6 18.4 48.5 18.7 8.8	4.3 8.5 35.4 29.8 22.0	32.5 10.1 23.1 22.2 12.2	7.6 18.3 35.0 21.5 17.6
	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)

(%)

1.7

0.6

SOURCE TABLE A3 COUNTRY VOLUNES

0.1

3.9.3 Length of Service

The percentages obtained are shown in Table 69 and as with the case of Wholesale Distribution, a comparison between these figures and those for full-time employees (Table 52) reveals some interesting features.

For male employees the concentration is in the shortest length of service category in all countries and the concentration is noticeably greater than in the cases of full-time employees - markedly so in the case of the Netherlands (cf age distribution) and Ireland.

For female employees there are both some interesting similarities and differences between the figures for part-time and full-time employees. In both cases the concentration in all countries (apart from Belgium) lies in the shortest length of service category. Belgium is different in that the concentration comes in the 2-4 years grouping and in fact 26.5% have lengths of service of 10 years or more, considerably than in any other country apart from Italy where the

TABLE 69 LENGTH DF SERVICE

MALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BEI.GIUM	LUXEM- BOURG	UK	IRE! AND	DERMARK
(TEARS)	40,6	54.8	41.5	75.5	37.7	55.4	41.3	81.3	67.4
2 - 4	28.4	19.1	215	20,8	34.1	25.0	26.9	•	18.2
5 - 9	17.6	10.5	21.0	2.0	14.4	•	14:0	•	6.6
10 - 19	8.1	6.8	13.3	0.6	6.5	•	7.0	•	
-> ≥ 20	5.3	3.4	2.8	0.7	2,2	•	10.7	•	3,5
(*)		5,5			Sil				
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

LENGTH OF SERVICE	ĢERI4ANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 > 20 (*)	31.0 33.3 21.8 11.1 2.7	51.5 23.5 12.2 6.7 1.8 4.2	27.2 26.1 26.2 19.6 0.9	57.3 26.1 11.6 3.8 0.6 0.6	18,3 36,1 19,1 20,0 6,5	45.3 25.4 15.5 10.8 2.9	41:4 28:1 18:3 8:4 2.7 1.1	47.5 35.7 7.4 7.5	37.1 32.2 18.9 8.4 2.2 1.2
	100.0	100.0	100.0	100.0	100,0	100.0	100.0	100.0	100,0

MALE AND FEMALE

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 > 20 (*)	31.4 33.1 21.6 11.0 2.8	52.2 22.6 11.9 6.7 2.1 4,4	28.9 25.6 25.6 18.8 1.1	61.5 24.9 9.4 3.1 0.6 0.5	19.3 36.0 18.9 19.3 6.3 0.3	46.7 25.4 15.1 10.1 2.8	41.4 27.9 17.9 8.3 3.5 1.0	55.9 29.2 6.5 6.1	40.2 30.8 17.7 7.8 2.3 1.2
ALL	100	100	100	100	100	(00	100	(00	100

SOURCE

TABLE A4

COUNTRY VOLUMES

142

corresponding figure is 20.5%. In Denmark and in Germany where there are high percentages of part-time working the length of service patterns are very similar for both full-time and part-time female employees. In the remaining countries the concentration is more towards the shortest length of service category than in the case of full-time employees.

3.9.4 Occupational Qualification Group

The distribution of part-time employees over occupational qualification groups is given in Table 70. In all countries the concentration of part-time employees is into the lower occupational qualification groups with the concentration greater than in the case of fulltime employees. This feature is similar to that observed for Wholesale Distribution. Another feature which was also noted for Wholesale Distribution is the percentage of part-time male employees falling in category IA in the United Kingdom and France.

Indices

Two sets of indices are given. Table 71 gives the figures relating to mean gross monthly earnings by occupational qualification group. Table 72 gives the figures for mean gross annual earnings.

	1	13
2°.A	.1.	2.4

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEM- Jourg	UK	IRELAND	Denmark
1A 1B 2 3 4 5 (*)	- 3.2 21.2 29.7 45.9	4.4 2.7 5.3 9.4 24.9 53.2	2.9 3.2 43.2 49.7	0.7 0.7. 6.2 92.3	9.8 49.7 37.0	33.9 37.5	5.8 5.0 42.6 40.9	- - 23.4 73.2	3.5 20.4 72.6
ALL	100	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXESI- BOURG	UK	IRELAND	DENMARK
IA IB 2 3 4 5 (*)	- 0.7 12.7 66.6 20.0	0.4 0.4 3.0 10.3 31.9 53.9	0,4 1,6 71,5 26,3	0.6 1.5 11.1 86.7	0.2 0.6 15.4 72.1 11.7		0.6 1.4 73.8 23.9	- 72.3 25.8	2.2 7.0 60.2 30.4
ALL	100.0	(00.0	100.0	100.0	100.0	100.0	100.0	100.C	100.0

MALE AND FEMALE

i

	GERMANY	FRANCE	ITALY	NETHER- LAND3	BELGIUM	LUXEN- BOURG	UK	IRELAND	DEIMÄRK
IA IB 2 3 4 5 (*)	- 0.8 13.1 64.9 21.2	1.2 0.9 3.5 10.1 30,6 53.8	0.3 0.7 1.8 68.1 29.1	0.1 0.6 1.3 10.0 88.0	0.2 0.7 15.1 70.9 13.0	3.0 23.1 43.2 28.9	0.7 0.8 1.8 70.8 25.6	60,2 37,6	2:2 0:4 6:6 56:2 34:6
	100.0	(00.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
SOURCE	والتكريبة بلغاية ليترب ويتهوهم	ጥለን የ		-144-	COBUTRY	VOLUME	an a	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	

%

•

Most of the comments made about the corresponding figures for Wholesale Distribution apply to those for Retail Distribution. The main interest lies in the figures for part time female employees who form the major part of the part time workforce and tend to be concentrated in job categories There are however entries for 3, 4 and 5. female employees in category 1 for France, Belgium and Denmark, (the French and Belgian figures being subject to large standard errors) and for category 2 in Germany, France and Belgium. Both in France and Belgium there is not a smooth decrease in index as job category ranges from 1 to 5. The effect disappears in France when the figures for male and female employees are considered together (but appears for the Luxembourg figures). Over the job categories 3, 4 and 5 the least variation is observed in the United Kingdom, the most in Denmark with a variety of patterns being observed for the other countries.

	INDEX	OP	IJCAN	GROSS	MONTHLY	EARNINGS	- B	Y	OCCUPATIONAL	QUALIFICATION	GROUP
TABLE 7										-	
356 7 12											

	GEIGIANY	FRANCE	ITALY	Nether- Lands	BELGION	LUXEM- BOURG	UK	IRELAND	DENMARK
1A 1B 2 3 4 5 (*)	195,2# 126.9 100.6 80.2	328.9# 280.2# 174.5# 136.8# 76.6 63.6	187.0# 125.2# 104,4 87,1		90,6# 96.8	42.2 79.5	78.0 <i>(</i> / 80.2		157.0 71.6
ALL	(00.0	100.0	100.0		100.0	100.0	100.0#		(00.0

FEMLE

.

	GERMARY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	Luxen- Boung	UK	IRFJAND	DENHAPE
1A 1B 2 3 4 5 (*)		338. # 282.3⊭ 112.7 119.0 111.5 84.8	i29.4 102.0 91.0		146.2# 151.9# 97.3 102.7 80.1	1245 94.7 81.3	103,3# 101,2 93,9		(83.3 142.4 101.9 78.9
ALL	100.0	100.0	100.0		100.0	100.0	(00.0		(30.0

ļ

.

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DEIMARK
1A 1B 2 3 4 5 (*)	- 187.6 122.4 100.3 81.3	364,9# 302,0# 136,0 122,1 104,5 80,2	150.0 182.0# 129.9 101.4 91.1		136.9# 150,8# 98:4 102.7 79.7	122.0 125.2 94.7 80.8	105.11 99.3 92.2		188,5 148,8 105,6 74,4
and a second	100.0	100.0.	100.0	- 146-	100.0	100.0	100.0		100.0

INDEX OF MEAN GROSS ANNUAL EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 72.

•• •• ••

LUXEN-NETHER-GERMANY FRANCE BELGIUM UK IRELAND DENMARK ITALY BOURG LANDS 300.5# 18 ٠ ø . . ۲ 222.8# ۲ 0 1B . 6 ¢ 199.6# 189,9# ø 2 • c 99,8# 1235# . 127.7 3 73,5# 135.6 84.2# 67.4 102.2 96.5 4 76.4 71.7# 66.6 112.7# 85.3 78.7 50.2 5 (*) 100.0# 100.0 100.0# 100.0 100.0 100.0 100.0 ALL

FEMALE

Ì

KALE

	GERMANY	FRANCÉ	ITALY	NETHER- LANDS	BELGTUM	LUXEN- BOURG	ιĸ	TRELAND	DEMINEK
1A 1B 2 3 4 5 (*)		271.4# 121.0 118.2 108.1 82.3	0 133.7# 102.2 88.7		161.4# 170:6# 101.7 100.9 82.8	- 123.4 94.8 71.5	100.7 92.9		190:2 141.0 100:6 78.1
ALL	100.0	100.0	100,0		100.0	100.0	100,0		100.0

MALE AND FEMALE

	GERHANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEM- BOURG	UK	IRELAND	DEIMARK
1A 1B 2 3 4 5 (*)	- 188,1 121.8 100,0 81.2	386.8# 288.1# 150.2# 14.0 98.5 74.7	20 4 ,3 ^{ff} 132,2 100,8 89,1		150.8# 168,7# 102,8 1005 83.6	111.4 125.5 94.2 72.3	!64,9 ! !22,9 ! 98,1 89,9	· · · · · · · · · · · · · · · · · · ·	191,8 144,1 102,5 75,2
	(00.0	100.0.	100.0		100.0	100.0	100.0		100.0

· ...

110

No figures are available for Ireland and the Netherlands. The annual figures reveal patterns very similar to those for the monthly figures (with slightly more entires available)

3.9.5 Size of Undertaking

The distribution of part-time employees over different sizes of undertakings is shown in Table and it is interesting to note the different patterns observed both as between male and female employees and as between countries.

For male employees, part-timers in the majority of countries are split between the smallest size of undertaking and the largest size. In Italy, where there are few male part-time workers, the concentration is very mcuh into the smallest size of undertaking. In the Netherlands, and to a slightly lesser extent in the United Kingdom, the concentration comes in the largest size of undertaking.

For female employees, there are two main patterns in evidence (three if we count that for Luxembourg where the concentration into the smallest size of undertaking is most pronounced). Germany, Denmark and to a

slightly lesser extent France each has a pattern where the majority of part-time female employees are almost equally split between the largest and the smallest size of undertaking. On the other hand in Belgium, Italy and the United Kingdom (and to a lesser extent the Netherlands, and perhaps even Ireland over a restricted range) the concentration comes in the largest size of undertaking category. No doubt these figures 'reflect the policy of various sizes of retail distribution undertakings towards the employment of part-time employees - and also the capacity to do so.

TABLE 73 SIZE OF UNDERTAKING

MALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IREI.AND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	35.1 7.6 6.5 7.8 6.0 37.1	40.8 7.9 7.3 7.0 5.7 31.3	56,7 10,7 2.5 2.8 1.6 25,6	12.6 5.8 3.7 3.1 3.1 71.8	30.0 12.1 2.0 12.9 43.0	58.9	21.1 6.6 4.1 6:2 59.5	21.9 31,0	37.1 4.4 4.7 50.4
ALI.	100.0	100.0	100.0	(00.0	100.0	100.0	100.0	100.0	100.0

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELĠIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	30.9 7.4 8.4 8.5 6.7 38.2	28:4 7:7 9:4 9:7 6:5 38:3	14.2 3.0 2.4 1.2 1.0 78.2	29.0 7:1 4:7 4:5 4:1 50.8	7.8 2.8 0.8 5.2 1.4 82.1	67.5 14:3 7:9 10:2	12.3 3.7 3.7 7.7 6.3 66.2	22.7 13.3 11.1 11.8 38.4	39,3 9,1 4,6 1,4 6,7 38,8
ALL	100.0	(00.0	(00.0	100.0	100.0	100.0	100,0	100.0	100.0

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK (IRELAND	DFNMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	31.1 7.4 8.3 8.5 6.6 38.1	30,8 7,7 9,0 9,2 6,3 36,9	19,3 3,9 2,4 1,4 1,1 71,9	25.2 6.8 4.5 4.2 3.9 55.5	9.0 3.3 0.8 5.6 1.3 80.0	66.3 13.8 9.5 10.3 -	13.2 4.0 3.6 7.4 6.3 65.5	22,5 12.8 10.3 11.5 36.6 6.3	39.1 8.7 4.6 1.5 6.2 40.0
ALL	100.0	100.0	100.0	100.0	100.0	(00.0	100.0	100.0	100.0
SOL	JRCE		TAL	LE A2		co	UNTRY VOL	.UMES	,,

SOURCE

LUUNIKI

150

Chapter 4: Banking NACE 812/813

This chapter considers the structure of earnings and the characteristics of the labour force in the Banking sector as defined under the NACE headings 812 and 813. It is useful to remember that central banking authorities are excluded and that, on the other hand, some other financial institutions (for example the Building Societies in the United Kingdom) have been included. The terms are definitions to be used are similar to those of the first three chapters and a separate list of occupational qualification groups is given at the end of this report

4.1 Some Overall Features

Some important characteristics of the labour force are given in Table 74 below - the number of employees, the number of part-time employees, the number of full-time female employees and the number of part-time female employees. These are also given as percentages of all employees.

Of particular interest are the percentages quoted for full-time female employees. The countries may put into three broad grownings according to the values which these percentages take. Germany, France, Luxembourg, Ireland and the United Kingdom all have percentages which fall in the range 42 to 50 (approx).

Belgium, the Netherlands and Denmark have values in the range 30-35% (approx) and Italy is noticeably lower with a value of 14.6%. This low percentage is a feature to which we shall return in subsecuent sections.

• •

.

.

TABLE 74 EMPLOYEES 000's

. .

i

:

Carde Jon Sold & David Street - Three submitte	-	Şirinin Artistan yanışınıştır.		an the statement of the statement					
Landing and a state of the stat	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENMARX
FJLI-TIME AND PART-TIME	399.9	327.9	353.8	88.8	65.3	5.4	395.8	15.3	39.9
FOLL-TIME	345.8	320.9	351.6	78.3	63.6	5.3	373.5	15.3	30.7
PART-TIME	54.1	7.0	2.2	10.6	1.7	0.1	22.4	-	9.2
FULL-TENE FEMALE	168.1	153.0	51.5	28.8	19.3	2.3	196.7	7.7	13.9
PART-TILE FEMALE	51.6	5.4	1.7	9.3	1.6	0.1	19.0		8.2

....

FULL-TIME AND PART-TIME = 100

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEM- EOURG	UK	INELAND	DENMARK
FULL-TIME AND PART-TIME	100	100	100	100	100	100	100	100	100
FULL-PIME	86.5	97.7	99.4	88.1	97.4	98.1	94.3	100	76.9
PART-TIME	13.5	2.3	0.6	11.9	2.6	1.9	5.7		23.1
FULL-TIME FEMALE	42.0	46.7	14.6	32.4	29.6	42.6	49.7	50.3	34.8
PART-TIME FEMALE	12.9	1.6	0.5	10.5	2.5	1.9	4.8		20.6

SOURCE

.

TABLE A1

COUNTRY VOLUMES

152

There are also considerable variations in the percentages of part-time employees in the labour force. In Ireland, the numbers observed are too small to be quoted and in Italy the percentage is extremely small. On the other hand in Denmark the figure is considerably higher than elsewhere with just under a quarter of the labour force falling into the part-time category. This high figure for part-time working is consistent with the similar high figures noted in earlier chapters. As has also happened in previous sectors, the majority of part-time employees arc female.

An interesting point to remember as background to these figures is that both in Italy and Iuxembourg the Banking sector, as defined and presented by the figures collected in this survey, is proportionately much larger than in any of the other EEC countries.

153

į, į

4.2 Composition of the Banking Sector

()

The figures are given in Table 75 and the following notation has been used:

τ	Credit Institutions with	n staff	of
	Civil Service status		

- II Credit Institutions with staff of collective agreement status
- III Credit Institutions with staff of other status

The figures reveal the varying ways in which the Banking industry is organised in different countries. In Germany, employees are split almost evenly between categories I and II. In the UK the main split is between categories II and III, with category III proportionately larger than elsewhere. In Denmark, employees are concentrated in category II (with the position for Ireland somewhat similar). The remaining countries follow patterns between these with category II uniformly the largest, and, apart from Luxembourg, category I the smallest.

TABLE 75 COMPOSITION OF SECTOR (PART TIME + FULL TIME EMPLOYEES)

	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
I	195.5	7,5	18.9	9.1	7.5	0.9	32.4	-	-
II	204.5	284.1	296.9	53.9	46.7	4.4	203.3	14.0	39.9
ÍH	-	36.3	37.9	25.8	11.1	0.1	160.2	1.2	-
TOTAL	399.9	, 327.9	353.8	88.8	65.3	5.4	395.8	15.3	39.9

FULL TIME + PART TIME = 100

	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
I	48.8	2.3	5.3	10.2	11.5	16.7	8.2	1	-
II	51.1	86.6	83.9	60.7	71.5	81.5	51.4	91.5	100.0
III	-	11.1	10.7	29.1	17.0	1.8	40.5	7.8	-
	100	100	100	100	100	100	100	100	100,0

SOURCE TABLE AL COUNTRY VOLUMES

 \bigcirc

()

4.3 The Sector by size of Undertaking

The figures relating to this aspect of the sector are given below in Table 76 and illustrate the concentration of employees into the largest size of undertaking, particularly in the United Kingdom.

0/

TABLE 🌠 EMPLOYEES BY SIZE OF UNDERTAKING

·~ . ·

\\$]	ZE	OF	
1111	-		1.

 \cap

UNDERTANIN (EMPLOYEES)	Ŵ GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	10 Denmark
10- 49	9,9	2.7	9.8	14.1	3.4	11.6	2.6	3.8	5.2
5 0- 99	8.3	2.7	4.8	5.1	3.8	8.0	2.1	4.7	4.5
100-199	10.1	3.3	5.3	2.6	5.4	12.9	4.2	7.6	5.3
200-499	16.6	10.9	12.2	6.1	7.9	4.1	5.8	8.0	9.8
500-999	9.8	17.6	6.8	3.0	10.0	63.4	6.2	4.7	14.4
≫1000	45.2	62.8	61.0	69.1	69.5	-	79.2	71.2	60.8
TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
NUMBERS 000's	399.9	327.9	353.8	88.8	65.3	5.4	395.8	15.3	39.9

SOURCE TABLE AS COUNTRY VOLUMES

Apart from Luxembourg, Germany has proportionately the least in the largest size category. In Germany it is also noticeable that a larger proportion than elsewhere are in enterprises with between 200 and 499 employees (the disaggregated figures show that these smaller enterprises tend to come in "credit institutions with staff of Civil Service status"). The proportion in the smallest size enterprises in the Netherlands is also of interest.

4.4 The Distribution of the labour force according to gross monthly earnings full-time employees

 \bigcirc

The methodology of this section is the same as has been used in the corresponding section in earlier chapters, (cf pp25,26 and 27 in the chapter in Wholesale Distribution).

The graphs which have been produced are given in Figure 3. <u>Graphs</u> These have been constructed from the frequency distributions of gross monthly earnings for full-time male and female employees in each country.

Relative Deviation, $u = (x-\bar{x})/\bar{x}$

where x denotes earnings in the national currency \bar{x} -z mean gross monthly earnings (male and female employees combined) Δx = class interval in x Δu = $\Delta x/g$

<u>Scaled Frequency</u>, $Y = f/_{\Delta u}$ where f is the percentage frequency. The values of Y have been scaled such that the area between the graph and the horizontal axis in each case is proportional to the number of full-time male/female employees.

•••••• Female ••••• Male







11 . . .

1



.





•



As has happened in the two previous sectors, the

graphs illustrate the differing percentages of

full time female employees in the full-time labour

force. These percentages are given below. TABLE 77

Full-time female employees as a % of all full-time employees :-

Germany	48.6
France	47.7
Italy	14.6
Netherlands	36.8
Belgium	30.3
Iuxembourg	43.4
United Kingdom	52.7
Ireland	50.7
Denmark	45+3

SOURCE TABLE A1

COUNTRY VOLUMES

Several features are very noticeable - for example the low percentage of full-time female employees in Italy, the very sharply peaked earnings distribution for female employees in Denmark, the separation of the distributions in the Netherlands and the very flat distribution for male employees in the United Kingdom. As has been observed in other sectors the graphs are in general unimodal (male employees in Denmark, almost an exception), positively skewed and with the earnings of female employees towards the lower end of the distribution. /The pattern for female employees in Italy does seem to be different from elsewhere, in that it follows the distribution for male employees more closely than elsewhere - though it has to be remembered that in these graphs as they stand 16.1% of male employees have been excluded as they fall in the open-ended interval at the top end of the earnings distribution7.

The variation in the distributions is summarised in Table 78 below which gives the relevant (%) coefficients of variation for gross monthly earnings.

Table 78: Coefficients of Variation (%)

Gross Monthly Earnings of full-time male and female employees

	Male	Female
Jermany	33.5	24.4
France	61.3	34.7
Italy	48.1	30.3
Netherlands	56.1	40.8
Belgium	40.4	29.7
Luxembourg	45.5	35.8
Jnited Kingdom	61.3	37.6
Ireland	51.9	35.3
Denmark	38.4	23.3

This table serves to confirm the features displayed in the graphs (eg the small variation in earnings about the the mean of earnings for remain omployees in Demark and the more extended distributions for male employees in the United Kingdom (and France)).

4.5 Gross Monthly Earnings as a function of the structural characteristics of the Labour force - full time employees

This section of the report considers the effects which the various characteristics of the labour force have had on the distribution of gross monthly earnings around the mean value in each case. The characteristics to be considered are as follows:-

- a. sex
- b. age
- c. length of service
- d. occupational qualification group
- e. size of undertaking

()

The definitions of the various indices are the same as for those used in the chapters on Wholesale and Retail Distribution.

4.5.1

.1 Employees classified by Sex

The graphs and tables of the previous section illustrated the position of the earnings of full-time male and female employees. Table 79 below gives a summary of these characteristics.

Table 79:

Summary of Female/Male Mean Gross Monthly Earnings (Full-time employees, earnings of full-time male employees = 100)

Germany	70.9
France	71.8
Italy	79.5
Netherlands	56.6
Belgium	71.2
Iuxembourg	62.4
United Kingdom	53.8
Ireland	61.7
Denmark	70.0

SOURCE TABLE C2 COUNTRY VOLUMES These figures follow a pattern similar to those observed in the Wholesale and Retail Distributions sectors. In Italy the difference is least whereas in Netherlands and the United Kingdom the difference is largest. The following sections consider the effects of various structural characteristics on these overall figures.

4.5.2 Employees classified by age

The average ages in years are given below for full-time male and female employees:-

	Male Employees	Female Employees
Germany	35.5	28.0
France	32.4	29.3
Italy	24.4	29.5
Netherlands	33.3	22.0
Belgium	33.6	26.9
Inxembourg	30.1	21.7
United Kingdom	33.2	24.2
Ireland	31.7	23.0
Denmark	33.2	28.2

SOURCE TABLE A6 COUNTRY VOLUMES

These averages are of similar order to those in the previous sectors with the variation in the averages for male employees less than that for female employees. The female averages might almost be put into two groups, the Netherlands, Luxembourg, the United Kingdom and Ireland with average roughly in the range 22-24 years and Germany, France, Italy, Belgium and Denmark with averages roughly in the range 27-30 years. A comparison with the average ages for female employees observed in the previous sectors reveals that for all countries apart from Italy the average age in Banking is lower than in the other two sectors. (The figures for age in Italy in Banking also have the lowest coefficient of variation (33%)).

()

 \bigcirc

The age distributions are given in Table **%**) below. For men there are interesting variations in the proportions in the youngest age group (under 21 years of age) from 1.1% in Italy to 13.9% in the United Kingdom. There are also interesting variations in the percentages of males in the 55 years and older age group from Italy and Luxembourg with 4.7% to Denmark with 11.5% possibly reflecting different retirement ages.
For women the most noticeable variations come in the youngest age group (under 21 years of age) where Netherlands, Luxembourg and the United Kingdom are noticeably high and Italy and Denmark noticeably low.

TABLE 80 AGE DISTRIBUTION FULL-TIME EMPLOYEES

MALE		· .	-		, ·				*
CEARS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
× 21	3.3	4.6	1.1	9.8	2.9	10.7	13.9	12.8	2.6
21-29	27.0	43.1	29.9	33.6	36.9	41.0	28 .9	34.7	43.5
30-44	39.9	29.2	45.9	29.4	36.6	32.2	30.5	31.5	29.9
45-54	19.5	14.4	18.5	16.4	16.8	11.4	16.6	11.0 -	12.4
> 55	10.2	8.0	4.7	10.8	6.8	4.7	10.1	10.0	11.5
UNDISCLO	(100) DSED	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
CEMAI E		• •	•			ť_		•	
YEARS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
Q 21	19.3	11.2	6.9	44.7	11.0	37.9	38.4	32.4	4.1
21-29	42.3	47.4	46.7	41.8	56.2	49.4	38.1	51.7	59.1
30-44	:23.1	- 23.9	32.5	7.3	24.2	9.3	12.7	9.7	25.2
45-54	12.0	13.5	12.8	4.3	6.8	2.5	8.7	4.2	7.8
> 55	3.3	3.6	1.2	1.9	1.9	0.8	2.1	2.0	3.8
	(100) OSED	(100)	(100.0)	(100.0)	(100)	(100)	(100)	(100)	(100:0)

SOURCE TABLE A3 COUNTRY VOLUMES

169

Indices of Mean Gross Monthly Earnings - Employees classified by Age and Sex

Two main sets of indices are considered one giving comparisons between groups of employees of the same sex and the other giving comparisons between male and female employees of the same age. A further table considers the particular case of the earnings of those aged under 21 with those aged 21-29 years. Table **%** gives the comparisons between employees of the same sex. For men, the indices rise from values in the range 42-61 (Netherlands 42.1, Italy 60.1) for the under 21 years of age category to peak values in the range 111-147. (Germany 111.5, Italy 146.3). For all countries apart from Germany and Italy the peak comes in the 45-54 years age group. In Germany the peak is in the 30-44 years age group.

For women the indices rise from values in the range 69-86 for the under 21 years age group (Belgium 69.9, Denmark 85.1) to peak values in the range 113-170. These peak values come in the 45-54 years age group for all countries apart from Denmark and Italy where the peak values come in the 55 years and older age group. Of particular interest are the indices for the youngest

age group. To illustrate the change in earnings between this group and the next youngest group Table 32 has been constructed which gives the mean gross monthly earnings of the under 21 years of age group as a percentage of the corresponding mean for the 21-29 years of age group.

TABLE 31 INDEX OF SEAN GROSS MONTHLY EARNINGS - BY AGE

MATE

AGE IN YEARS	GERMANY	FRANCE	ITALY	METHER- LANDS	BELCIUN	LUXEM- BOURG	UK	IRELAND	DENNARK
< 21 21 - 29 30 - 44 45 - 54 > 55 (*)	58.3 78.6 111.5 110.4 103.7	50.3 73.4 121.7 143.8 111.5	60.1 69.7 100.7 137.5 146.3	42.1 75.1 119.5 133.4 126.1	49.5 74.6 110.0 133.8 125.0	48.7 81.8 125.9 141.1 115.1	44.3 81.4 119.3 134.6 113.2	55.3 74.9 127.2 131.0 122.2	58.2 76.3 112.6 133.0 1 29 .9
Al-L	100.0	100	100.0	100	100.0	100.0	100.0	100	100.0

.

FEMALE

AGE IN LUXEM-NETHER-DENEMARX IRELAND UK. GERMANY ITALY BELGUIM FRANCE YEARS BOURG LANDS 85.1 81.5 73.7 73.8 73.7 69.9 72.8 < 21 76.4 71.1 98.7 93.4 107.9 107.7 91.8 86.7 109.2 21 - 29 95.0 89.5: 112.8 150.9 144.9 130.4 121.9 116.0 115.5 109.6 153.2 30 - 44158.4 109.7 169.9 132.3 132.8 169.0 119.4 126.4 133-3 45 - 54 120.1# 116.9 113.1 143.6 141.8 162.2 119.8 2 35 118.6 . . (*) 85.6# . --. 100 100.0 100.0 100 100 100 100.0 100.0 100 ALL

MALE AND FERALE

AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCUIN	LUXEM- BOURG	UK	IRELAND	DERMARK
< 21 21 - 29 30 - 44 45 - 54 > 55 (*)	63.8 83.8 117.9 117.5 115.4	58.6 79.5 121.4 137.8 120.0 90 /#	61.1 71.7 102.3 138.2 149.3	49.7 82.7 137.1 152.7 146.4	54.4 77.7 114.9 139.9 131.8	55.4 89.7 142.9 162,4 134.1	53.9 89.1 137.2 147.1 137.0	63.8 82.1 147.5 153.1 144.2	68.4 81.6 114.6 131.6 137.1
LILL.	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

SOURCE

COUNTRY VOLUMES

TABLE S2

Mean Gross Monthly earnings of full-time employees under 21 years of age. (Earnings of full-time employees aged 21-29-100, male and female employees considered separately)

.4
•4
•7
•2
.1

The patterns revealed in this table are similar to the ones seen in the corresponding tables in the previous The change with age is greater for male employees chapters. than for female employees in all countries with once again the ratios high for Italy (though not the highest for female employees in this case) and low for the United Kingdom, the Netherlands and Luxembourg. The change is smaller both for male and female employees in Banking in Germany, Italy and Denmark and greater for both sexes in the United Kingdom than it was in the previously considered The remaining countries fall into the two sectors. groupings (a) France, Luxembourg and Italy where the change for males is smaller, and that for females greater and (b) the reverse position in Netherlands and Belgium.

Table \$3 gives comparisons between male and female

employees when the effect of age is removed.

TIBLE 83

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EXPLOYEES (EY AGE) (MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EXPLOYEES IN SAME CATEGORY - 100)

AGE IN YEARS	GERMANY	FRANCE	- ITALY	NETHER- LANDS	BELCIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54 ≥55 (*)	93.0 85.7 73.8 76.6 81.1	101.6 87.7 68.1 63.2 77.2 71.3#	97.7 98.9 86.6 77.1 77.1	99.1 82.3 72.6 71.7 72.8	100,4 87,5 78,9 70,6 68,4#	93.2 82.2 71.8 75.1 77.8	89.4 71.1 58.7 52.8 55.5#	90.9 81.3 73.2 74.6#	(02,4 85,8 70,1 57,8 61:0
ALL	70.9	71.8	79.5	56.6	71.2	62.4	53.8	61.7	70.0

SOURCE

TABLE D8

COUNTRY VOLUMES

The figures for France, Belgium and Denmark for the under 21 age group are interesting in that they exceed 100. The figures for all countries in this age group are high though those for the United Kingdom slightly lower by comparison. The figures for the United Kingdom show a marked fall-off with increasing age. Similar, if less pronounced, patterns are observed up to the age range 30 - 44 years in Germany and Luxembourg and up to the age range 45 - 54 years in the remaining countries.

Employees classified by length of service 4.5.3

> As in the previous chapters it is to be expected that there will be a relationship between the length of service distribution and the age distribution.

The percentages which have been obtained are shown in Table 84 below:

LENGTH OF SERVICE - FULL TIME EMPLOYEES TABLE 94

	-
MAL	
1.0.1	. 1

	IRELAND 1	UK	LUX 1	BELGIUM	NETHS	ITALY	FRANCE	GERMANY	MALE VEARS
6.7	15.6	20.9	22.9	12.5	22.2	17.6	15.9	10.5	
20.4	22.0	18.2	28.9	20.1	24.2	17.3	26.1	22 4	> 2
25,9	19.8	20.7	20.8	24.3	20.1	15.0	21.5	27.7	2-4
24.2	22.2	22.0	- 14.9	23.8	17.3	32.4	21.1	21.1	30.10
22.8	20.4	18.2	12.4	19.4	16.1	17.6	14.5	15.4	10-19
(100.0)	(100)	100.0)	(100.0)((100.0)	(100)	(100)	(100) (0.7)	(100) OSED	UNDISCL
DENMARK	IRELAND	UK	LUX	BELGIUM	NETHS	ITALY	FRANCE	GERMANY	CEMALE VEARS
14.3	31.0	36.2	35.6	19.5	39.1	27 0	A OF	20.0	
36.6	37.0	28.3	37.2	26.9	37.0	21.5	25.5	20.0	< 2
30.8	20.9	23.1	19 . 9	25.9	16.1	16.9	23.0	30.1	2-4
14.2	7.2	8.9	5.9	20.5	5.2	23.4	18.4	12 5	5-9
1		• • • ·	1.5	6.0	0.6		1017	12.5	10-13
3.8	3.4	3.4	1.0	· 0.2	3 Z.O .	11.2	127.	62	
	1RELAND 31.0 37.0 20.9 7.2	UK 36.2 28.3 23.1 8.9	LUX 35.6 37.2 19.9 5.9	BELGIUM 19.5 26.9 25.9 20.5	NETHS 39.1 37.0 16.1 5.2	ITALY 27.0 21.5 16.9 23.4	FRANCE 19.4 25.5 23.0 18.4	GERMANY 20.0 36.1 26.2 12.5	< 2 2-4 5-9 10-19

SOURCE TABLE A4 COUNTRY VOLUMES

For male employees perhaps the most noticeable variations come in the percentages for the group with under 2 years of service - from 6.7 for Denmark to 22.9 for Luxembourg. Another interesting feature is the relatively high percentage of 32.4 in the 10-19 years of service category in Italy. It is also noticeable (of Tables 20 and 52) that the percentages in the longest length of service category in Banking are (much) higher in all countries than in other sectors no doubt reflecting different career patterns in the sectors.

For female employees the Netherlands, Luxembourg and the United Kingdom have high percentages in the shortest length of service categories reflecting the age distribution figures of Table 80. France, Italy and, to a lesser extent, Belgium exhibit a different pattern with percentages in the longest length of service categories which are noticeably higher than elsewhere. A comparison with the figures for the other sectors indicates a slight shift in the distribution towards longer lengths of service in Banking. The percentage in the longest length of service category is however very similar for all countries to that in the corresponding category in Wholesale and Retail Distribution apart from in France and Italy as mentioned above. Belgium is an exception to this rule with the percentage in the longest length of service category in Retail Distribution higher than elsewhere.

Ċ

Indices

Two sets of indices are considered - Table 85 gives comparisons between employees of the same sex. Table 86 gives comparisons between male and female employees with the same length of service.

For male employees alone there is a smooth rise in the index in all countries (apart from Denmark) as length of service increases from values in the range 67-85 (France 67.1, Denmark 84.7) for the under 2 years length of service category to values in the range 129-160 (Germany 129, France 159.8) for the 20 years and over length of service.

For female employees, the values tend to be higher than the entries in the corresponding cells for men (apart from Italy and France in the longer length of service categories) reflecting the different length of service distributions for men and women. As in the case of men the indices rise from values in the range 74-85 (France 74.4, Germany 85.0) for the under 2 years length of service to values in the range 135 to 207 (Germany 135.3, Luxembourg 207.0) for the 20 years or more length of service.

TABLE 85

MALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
< ? 2 - 4 5 - 9 10 - 19 ≥ 20	83.8 87.1 91.1 110.8 129.0	67.1 78.1 91.9 118.1 159.8 87.7	66.2 78.4 87.5 107.7 150.9	69.2 85.6 101.6 121.6 138.0	68.5 77.7 91.3 110.4 142.3	76.9 84.6 102.1 122.5 149.5	67.5 84.8 89.1 109.7 152.3	68.8 75.2 83.9 109.0 154.5	84.7 76.6 83.5 102.7 139.8
• <u>(*)</u> ALL	100	100.0	100	100	100	100	100	100.0	100

FEMALE

ł

	LENGT 1 OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
	<pre>< 2 2 - 4 5 - 9 10 - 19 > 20 (*)</pre>	85.0 93.4 103.4 120.8 135.3	74.4 87.4 97.7 117.5 141.1 85.1	78.3 90.5 96.5 115.0 143.4	80.2 98.8 117.7 158.7 181.1	77.1 88.6 101.5 119.9 148.8	84.9 95.5 113.9 143.0 207.0	77•3 95•1 115•0 146•3 163•7	80.4 92.8 107.6 152.9 193.5	84.9 90.8 102.1 120.8 151.7
1	ALL	100	100.0	100	100	100 ·	100	100	100	160

MALE AND FEMALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMAR
< 2 < 2	79.9	69.2	67.3	68.0	68.9	76.5	66.5	68.9	· 79 •3
2 - 4	86.7	81.8	79.7	85.2	79.2	86-4	83.9	79•1	7 9•5
5 - 9	96.4	93.8	88.5	107.6	93.3	106.5	97.8	92:7	89•9
10 - 19	119.4	119.2	:09.2	138.9	113.6	137.2	132.3	130.4	112.2
≽ 20	140.8	154.5	151.8	160.3	150.7	177.0	187.5	184.1	157-1
(*)	•	<u>8</u> 4.0	-	•					
ALL	100	100.0	100	100	100	100	100	100	100

SOURCE

TABLE DIO

177

COUNTRY VOLUMES

.

Once again the values of the index corresponding to the longest length of service category show some extremely interesting variations. In particular the values for Luxembourg, the Netherlands and Ireland are markedly higher than elsewhere (a feature also observed for the Netherlands and Luxembourg) in other sectors and no doubt reflects the different underlying age distribution patterns of female employees in these countries. The figures below in Table 36 illustrate the comparisons between the mean gross monthly earnings of male and female employees having the same length of service.

TABLE 86

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES -BY LENGTH OF SERVICE (MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE ENTLOYEES IN

	SAME	CATEGORY	# 100)		District.				
LENGTH OF SERVICE	GERMANY	FHANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 ≥ 20 (*)	71.9 76.0 80.5 77.3 74.4	79.7 80.5 76.4 71.5 63.4 69.7	94.2 91.8 87.7 84.9 75.6	65.7 65.3 65.6 73.9 74.3	80.0 81.2 79.1 77.3 74.4	68.8 70.4 69.6 72.8 86.4	61.6 60.3 69.4 71.7 57.8	72.1 76.1 79.1 86.5 77.3	70.2 83.0 85.6 82.4 76.0
ALL	70.9	71.8	79.5	56.6	71.2	62.4	53.8	61.7	70.0

SOURCE

TABLE D10

COUNTRY VOLUMES

As in previous sectors, the figures for Italy tend to be higher than elsewhere but they are only noticeably higher than elsewhere for the shortest length of service categories. In fact Luxembourg is highest for the longest length of service category. The figures for the United Kingdom and the Netherlands are lower than elsewhere with the figure for the United Kingdom particularly low for the longest length of service category. In France, Italy, Belgium and the United Kingdom there is some evidence of a decrease in the ratio with Increasing length of service with the reverse pattern in the remaining countries - though in neither case is the change a smooth one.

In the majority of cases the values in the individual cells are greater than the value of the ratio of overall means, reflecting the different distribution patterns for male and female employees.

3

4.5.4 <u>Employees classified by Occupational Qualification</u> Group

The general remarks given in Sections 2.5.4 and 3.5.4 relating to the difficulties of classifying employees into broad groupings apply also to this section. In this case the broad groupings are:-

- 1. Directors, top management
- 2. Senior executives
- 3. Executives (junior management)
- 4. Highly-qualified clerical staff
- 5. Qualified clerical staff
- 6. Other employees

Ċź

A fuller explanation of these is given in the Techical Annex, With the definitions given above, the figures given in Table 87 were obtained and they exhibit some features which, at first sight would suggest some very different structures. The figures for Netherlands are particularly noticeable with proportionately far more employees than in other countries being classified as category 6 (other workers). Even when these figures are combined with those for category 5 (qualified clerical staff) the figures for the Netherlands are high. On the other hand the proportions falling in these categories in Italy are noticeably lower and show up particularly for women when the figures are considered separately. (The extremes for women are 23.4% category 5 or 6 in Italy, 94.3% category 5 or 6 in Netherlands.) The figures for France are also of interest, particularly the proportion (26.2%) in category 3 (executives/junior management) for women which apart from Italy is much higher than elsewhere.

TABLE 87 OCCUPATIONAL QUALIFICATIONS

FULL TIME EMPLOYEES

MALE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
1	0.9	4.4	1.0	2.1	1.3	4.3	1.2	2,3	1.9
2	4.2	15.2	12.6	3.9	5.8		5.7	12.8	8.5
3 .	30,3	30.9	29.0	5.7	21.5	12.8	22.4	27.4	22.4
4	42.0	12.7	40.7	23.6	33.1	28.1	26.3	16.9	25.4
5	15,1	22.4	6.4	30.4	30.3	31.4	28.9	29.1	24.9
6	7.5	14.3	10.3	34.3	. 7.9	15.9	15.6	11.4	16.9
	100.0	100.0	100.0	100.0	100	100	iuù	100	100.0
NUMBERS	177.7	167.9	300.1	49.4	44.3	3.0	176.8	7,5	16.8
FEMALE	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK.	IRELAND	DENMARK
1		0.1	0.1				-		-
2	0.1	2.7~	0.8	0.2		•	-	£ 	•
3	5.6	26.2	17:0-	0.7	4.0	2.3	1,1	1.6	2,6
4	41.2	16.2	58.6	4.8	27.8	15.9	13.7	25.7	11.7
5	34.3	33.9	20.9	31.3	56.0	53.8	55.5	68.2	40.4
6	18.7	20.9	2.5	63.0	11.9	27.7	29.6	4.5	45.1
	100.0	100.0	100.0	100.0	100.0	j 100.0	100.0	100.0	100.0
NUMBERS	168.1	153.0	51.5	28.8	19.3	, 2.3	196,7	7.7	13.9
NALE AN	D FEMALE								•
	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
1	0.5	2.4	0.9	1.4	0.9	2.4	0.6	1.1	1,1
2	2.2	9.3	10.8	2.5	4.2	4.3	2,7	6.4	4.7
3	18.3	28.7	27.3	3.9	16.2	8.3	11.2	14.4	13.5
4	41.6	14.4	43.3	16.6	31.5	22.8	19.7	21,3	19.2
5	24.5	27.9	8.5	30.7	38.1	41.1	42.9	48.9	31.9
6	13.0	17.4	9.2	44.9	9.1	21.0	23.0	7.9	29.6
	100	100	100	100	100	1.00	100	100	100
NUMBERS 000's	345.8	320.9	351.6	78.3	63.6	5.3	373.5	15.3	30.7

181

%

The very noticeable differences say between the Netherlands and Italy are rather puzzling. Italy has proportionately a large banking sector but with proportionately few women which could mean that the women who do enter the banking sector do have particular qualifications etc which might enable them to reach higher levels than in other countries. Netherlands has a full time female workforce which is noticeably biased towards the youngest age groups who might be expected to occupy the more junior positions. The differences are so noticeable, however, that it leads me to speculate that the question asked in the Netherlands might have made reference to the possession of some banking qualification, which in other countries was not included?7

)

Indices

As in previous sections and chapters two sets of indices are considered, one giving comparisons of mean gross monthly earnings of employees of the same sex, the other giving comparisons between male and female employees in the same occupational qualification group.

Table 88 gives the figures for comparisons between employees of the same sex. The considerations mentioned on page (re cut off values etc) complicate the interpretation.

These considerations may help to explain why the figures for France, Italy and Ireland for category 1 sales are much higher than elsewhere, (in that fewer of the top earners have been excluded from the calculations). (On the other hand Germany is lower than might have been expected (?) even allowing for the problems of cut-off values 7. At the other end of the occupational qualification scale, the figure for males in France is noticeably lower than elsewhere. In terms of occupational qualification structure France has proportionately more males in the top two categories whose earnings might be expected to pull up the overall mean and so depress the value of the index figure for category 6. This would, however, tend to work against the high figure for category 1, unless the inclusion of high earners in France in category 1 has had a marked effect. The high figures for the Netherlands in 3, 4 and 5 may, in part, be explained, I feel,

TABLE 88

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEN- BOURG	UK	IRELAND	Denmark
1 2 3 4 5 6 (*)	187.4 176.1 121.2 91.7 74.0 69.4	286.0 158.9 100.7 76.8 64.3 53.9	263.6 194.3 106.0 77.9 75.7 66.6	205,1 179,7 166,9 124,8 89,5 67,1	203,6 173,0 130,7 94,3 77,1 69,4	229.0 176.6 139.6 98.1 78.0 67.5	209.8 138.3 97.5 67.5 60.2	259,2 165.3 126.1 84.4 59.0 57.5	222.2 164.2 128.4 97.1 69.5 71.0
ALL	100	100	.100	100	100	100	100	100	100

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	Un.	IRELAND	DENGJARK.
1 2 3 4 5 6 (*)	- 2.23.9# 141.5 109.4 92.7 78.7 -	294.2 194.6 126.7 100.9 86.8 71.2	303.7# 218.6 128.0 97.2 81.7 72.9	192.2# 143.8# 155.6 112.2 89.1	- 155,4 115.5 91.3 83.2	- 181.0 131.4 93.2 86.2	201.4# 142.4 97.3 81.0	- 203,4# 128,6 87,5 54,6#	167.8 129.4 99.3 88.6
ALL	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

.

÷

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BCURG	UK	IRELAND	DEFINARK
1 2 3 4 5 6 (*)	218.4 204.5 137.6 98.9 79.7 69.8	327.6 180.1 111.3 85.9 72.8 60.4	271.1 189.8 108.8 80.2 74.1 68.2	243.4 211.1 191.6 143.9 94.9 69.6	222.4 188.6 141.5 99.8 78.6 71.6	274,4 210,9 163,3 111,7 80,0 7!.6	• 277,3 180,6 1(8,6 75,5 64,6	320,3 204,2 155,8 100,4 68,5 67,4	257.4 189.3 147.6 110.5 80.5 75.1
	100	100 .	100	100	10.0	100	:00	100	100
SOURCE	s Eanair seannaisteach S	na n		184	CONTEN	A NULLING			n k

by its occupational qualification structure which is different from elsewhere.

For females the entries tend to be higher than in the corresponding cells for men, a feature noticed in other sectors and probably attributable to the more noticeable concentration of women in the lower occupational qualification groups. The figures for categories 1 and 2 have been particularly affected by the problems of small numbers with France, and Italy (to a lesser extent), the exceptions. The figures for Netherlands reflect the occupational qualification structure, though the figure for category 3 is unusual.

Table 89 gives comparisons between male and female employees.

TABLE 89

INDEX OF MEAN GROSE MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES - BY OCCUPATIONAL QUALIFICATION GROUP

		-							مراک میرون برای زران و مادرا برای ا
•	GERMANY	FRANCE	ITALY	METHER- LANDS	BELGIUM	LUXES- BOURG	UK	TRELAND	DEMIARK
1 2 3 4 5 6 (*) ALL	90.1# 82.8 84.6 88.9 80.4 70.9	73.9 88.0 90.4 94.4 97.0 94.9 71.8	91.6# 94.4 96.1 99.2 85.9 87.1 79.5	60,6# 48,8# 70,6 70,9 75,2 56,6	- 84.7 87.2 84.2 85.4 - 71.2	- 80,9 93,6 74,5 79,7 	• 78,3# 78,5 77,5 72,4 53,8	99,5 [*] 94,0 91,4 58,6 [#] 61,7	91.5 93.3 100.0 87.4 70.0

(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY = 100)

SOURCE

TABLE C2

COUNTRY VOLUCES

For categories 1-3 the ratios for Italy are higher than elsewhere (category 3 in Ireland, being the exception), a feature noted in other sectors. The figures for the Netherlands are lower than elsewhere (apart from category 6, where Ireland records the lowest figure) with the value for category 3 particularly low. There is some evidence of a decrease in ratio over grades 1 to 6 in Germany, Ireland and the United Kingdom - though only in Ireland is the decrease an unbroken one. The reverse effect is observed in France, Netherlands and Belgium (to a lesser extent) - though once again the change is not a smooth one. In Italy and Denmark the values show an eventual decline though passing through higher values in between. The values for all categories are almost consistently higher

than in other sectors, perhaps reflecting some difference in career patterns across the sectors. Once again, the values in the individual cells are higher (in all cases apart from the Netherlands, category 3 and Ireland, category 6) than the ratio of overall means, reflecting the different distributions of male and female employees over the occupational qualification groups.

4.5.5 Employees classified by size of undertaking

The final set of indices which we consider is concerned with the size of undertaking as the factor of classification. In Table 76, the figures for all employees were given. Table 90 gives the figures for full-time employees only and also for male and female employees separately. Since the percentage of part-time employees in the Banking sector is small in most countries these figures agree closely with those in Table 76 and comments were given then.

Indices

Table 91 gives the figures comparing employees of the same sex.

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BFLGIUM	LUXEM- Bourg	UK	IRELAND	DETIMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	11.0 8.2 10.1 16.0 9.6 45.1	2.5 2.8 3.5 11.6 14.3 59.9	9.8 4.7 5.2 12.2 6.7 61.3	11.1 4,3 2,8 6,6 3,0 72,2	2.9 3.1 4.5 7.0 10.8 71.6	10.3 7.0 10.6 4.0 68.1 -	2.8 2.4 5.1 6.3 6.8 76.7	4.4 4.1 6.9 8.3 4.6 71.7	5.2 4.5 4.7 10.3 14.8 60.5
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

MALE

NO. UT EMPS	GERHANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49 5099 100-199 200-499 500-999 ≥1000 (*)	8,1 8,5 10,5 17,7 10,1 45,1	2.2 2.1 3.0 9.8 15.1 67.8	9.3 5.6 5.9 11.8 7.6 59.7	18.9 6.6 2.5 5.9 3.4 62.7	4.5 5.2 7.7 9.8 7.4 65.4	12.8 9.0 14.9 4.3 58.9	2.0 1.7 3.4 5.3 5.6 82.0	3.2 5.4 8.2 7.7 4.8 70.7	5,4 3.4 6.6 10.7 14.9 59.0
ALL	100.0	100.0	100.0	(00.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

2011					· · · · · · · · · · · · · · · · · · ·		,		
ALL	130.0	100.0	100.0	100.0	100.C	100.0	100.0	100.0	16.0
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	9,6 8,3 10,3 16:8 9,9 45:1	2.3 2.5 3.3 10.8 17.5 63.7	9.7 4.8 5.3 12.2 6.9 61.1	14.0 5.1 2.7 6.3 3.1 68.7	3.3 3.8 5.5 7.9 9.8 69.7	11.4 7.9 12.5 4.1 64.1	2,4 2.0 4.2 5.7 6.2 79.5	3,8 4,7 7,6 8,0 4,7 71,2	5.3 4.0 5.6 10.5 14.9 59.8
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK .	IRELAND	DENMARK

SOURCE

z

ł,

TABLE A2

188

TABLE 91

INDEX OF MEAN OF MEAN GROSS MONTHLY EARNINGS BY SIZE OF UNDERTAKING

ł

MALE

NO. OF EMPS	GERMANY	FRANCE	1TALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49	87.5	123.3	95.6	98.2	98.8	128.9	•	117.5	101.5
50-99	93-2	113.8	95.4	102.5	99.7	107.0		116.6	103.0
100-199	97•3	108.2	94.6	126.8	91.5	103.3	110.0	96.2	96.8
200-499	96.6	94.6	99.0	107.8	93.9	95.0	110.0	103.9	97.0
500-999	102.0	92.8	102.4	101.3	99.2	95.2	95.9	98.17	100.7
≽1000	105.8	101.4	101.5	98.3	101.3	-	96.6		100.2
(*)	-	-	-	-				-	-
ALL	100	100	100	100 .	100	100	100	100	100

FEMALE

NO, OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	JK	I RELAND	DENMARK
10-49	91.5	108.4	95.9	88.1	86.4	115.1	103.2	83.0	100.2
50-99	96.9	113.7.	96.7	88.1	90.7	104.8	115.6	98.1	94.7
100-199	96.9	100.4	96.2	119.6	93.0	9 9 •7	92.6	81.7	104.4
200-499	100.1	92-1	104.8	101.3	90.5	110.2	113.1	9 4•7	103.4
500-999	102.2	94.4	106.6	101.8	93.3	95•4	10 1. 9 i	103.7]	97.7
≥1000	102.3	101.7	99.5	103.7	104.9	-	98.9	<u> </u>	99.7
(*)	-	-	-	-	-	-	-		
ALL	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

SOU	IRCE		TAI	BLE D5	T	CO	אדאני TRY VƏl	LUMES	
ALL	100	100	.100	100	100	- 00-	100	100	100
(*)	-	• ·	-	-	-	-		-	
≥1000	104.3	100.5	101.3	101.1	102.8		96•5	L .	100.2
500-999	101.5	95•3	102.6	99.6	100.3	97.0	101.1	100.07	99-8
200-499	97.3	95-1	99•7	107.6	90.6	98.7	113.6	101.0	99.0
100-199	96.9	106.4	94.5	127.1	99.3	97.6	110.6	87.9	97.3
50-99	94.6	116.6	95.1	93-0	92.5	103.1	•	105.4	102.6
10-49	91.2	118.9	95•7	88.1	91.4	119.0	•	111.1	101.0
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK

TABLE **D5**

There are some interesting differences. For males in France, Ireland and Luxembourg there appears a decrease in index with increasing size (apart from the largest size category in France), Belgium and Italy show slight increases with size, an effect slightly more noticeable in Germany - though the increase is not smooth. Netherlands has a pronounced peak in the 100-199 employees group. Denmark shows some signs of a "U" - shaped distribution and the United Kingdom a drop from the middle range to the largest range.

For women in Germany, Belgium, Ireland and Italy (apart from the largest size category) there is some evidence of an increase in index with increasing size - though the increase is not uniform. Netherlands has a pronounced "peak" in the 100-199 employee size range, as does Denmark but to a much lesser extent. France shows signs of a "U"-shaped distribution, as does Luxembourg over the range up to 499 employees. The United Kingdom has two noticeably higher values, one in the 50-99 employee size range, the other in the 200-499 size range.

Table 92 gives the comparisons between the earnings of male and female employees.

table 92

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EXPLOYEES - BY SIZE OF UNDER-

(MEAN GROSS MONTHLY EARNINGS OF FULL THE MALE EMPLOYEES IN SAME CATEGORY = 100)

no of EIPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	Denmark
$10 - 49$ $50 - 99$ $100 - 199$ $200 - 499$ $500 - 999$ $\gg 1000$ (*)	74.1 73.7 70.6 73.5 71.1 68.5	63,1 71,7 66,7 70,0 73,1 72,1	79.8 80.6 80.9 84.2 82.8 78.0	50,8 48,6 53,4# 53,2 56,9 59,7	62.2 64.8 72.4 68.6 66.9 73.7	55.7 61.1 60.3 72.4 62.5	40,6# 46,8# 45,3 55,3 57,2 55,1	43.6 51.9 52.4 56.2 65.1]	69.1 64.4 75.6 74.7 68.0 69.7
ALL	70.9	71.8	79.5	56.6	71.2	62.4	53.8	61.7	70.0

SOURCE

TABLE D6

COUNTRY VOLUMES

The values of the ratios in Italy are consistently higher than elsewhere and those in the United Kingdom low but only consistently lower than elsewhere for the three smallest size categories, and the largest size category. For the remaining two categories the Netherlands has the lowest values. In Germany there is some evidence of a

decrease in ratio with increasing size of undertaking. In all other countries apart from Italy, the reverse pattern may be observed though only in Ireland is the movement smooth and in Denmark there is little variation between the largest an smallest size categories, the main variation coming in between. Italy shows only small variations with change of size of enterprise.

As has been observed with this factor of classification in other sectors there is the least variation between the values in the individual cells and the ratio of the overall averages, as compared with other factors of classification (age, length of service and occupational qualification group).

4.6 Gross Annual Earnings

(______

The analysis so far has considered gross monthly earnings for full-time employees, classified by various factors. As in the case of the Wholesale and Retail Distribution sectors, a similar analysis is possible for gross annual earnings for which a frequency distribution containing 27 class intervals has been drawn up for each country.

Indices have been calculated for mean gross monthly earnings and are available for all full-time male employees, all full-time female employees and for these groups broken down into occupational qualification groups. Indices are also available to compare male and female earnings for each of these categories. The global figures are given in the summary table below:

Table 93 :

Summary of Female/Male Mean Gross Annual (Earnings of full-time male employees = 100)

Germany France Italy Netherlands Belgium Luxembourg United Kingdom Ireland	70.4 70.1 77.9 56.2 70.2 62.6 53.4 60.3 68 0	(70.9) (71.8) (79.5) (56.6) (71.2) (62.4) (53.8) (66.7) (70.0)	Figures in brackets a for mean/g monthly ea
Denmark	68.0	(70.0)	

are those gross earnings

SOURCE TABLE C4 COUNTRY VOLUMES

In general the agreement between the annual and monthly ratios is very good, with the annual ratios tending to be slightly lower than the monthly ones (Iuxembourg - being the exception - but the difference is very small) - a feature also noticed in the Wholesale and Retail Distribution sectors.

Table 94 gives the various indices of mean gross annual earnings by occupational qualification group. In general the agreement between annual and monthly figures is very good. For male employees the only noticeable differences come in category 1. In most cases these are small though in Ireland, Denmark (and Luxembourg) the annual ratios are of a higher order and are noticeable. Similarly for female employees the agreement is good.

المحادثين فيشاد تروي والان

나다

The comparisons of mean gross annual earnings between male and female employees in the same occupational qualification groups is given below in Table 95. Here the figures are in good agreement with the monthly ones with the United Kingdom slightly lower in category 3 as compared with the monthly figures.

INDEX OF MEAN GROSS ANNUAL EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 94

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- LOURG	UK	IRELAND	DENMARK
1 2 3 4 5 6 (*)	185,3 174,3 119,7 91,5 74,0 68,2	298,0 155,3 98,2 74,4 61,7 54,5	261.2 181.3 102.1 77.3 74.2 67.4	202.8 171.4 162.4 120.5 88.0 66.6	196.5 167.8 129.4 92.4 76.2 69.8	224,5 172,3 136,2 96,5 77,7 68,0	0 203.1 133,1 93.2 66.0 62.6	241,8 162.6 126.2 82.6 57.6 57.5	2.10.0 164.0 124.6 95.1 68.5 70.4
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	FELGIUM	LUXE: BOURG	UK	IRELAND	DENMARK
1		302.9	308,9F	•		-			
2	220,0#	194.0	2215	186.6TF	Ð	•	-	0	• •
3	140.8	124,1	123.4	136.7#	156.9	176,4	179.3#	194.4	164,5
4	108,3	98,7	97,1	156.6	1161	128.8	135,0	124,4	128.5
5	91.7	84.0	80.6	109,9	90.7	91.8	95.0	87.1	48.2
6	78.9	72,4	73.7	88.4	82.5	88.0	83,4	52.2₩	88.
(*)									
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

	GERMANY	· FRANCE	ITALY	NETHER- LANDS	BELGIU14	LUXEM- BOURG	UK	IRELAND	DEMMARK
1 2 3 4 5 6 (*)	215,9 202,3 136,0 98,0 78,6 69,5	332.5 177.4 108.5 83.2 69.8 60.9	268.7 186.7 104.7 79.4 72.7 69.0	237.9 199.3 184.3 138.1 92.3 68.4	214.7 182.6 140.0 98.1 77.1 71.3	266.8 204.0 158.2 109.1 78.9 72.3	264.2 170.6 111.1 72.6 65.9	295.0 198.3 153.4 95.4 66.1 66.5	245.0 190:4 144.1 108:5 78,7 73.8
ALL	100.0	100.0	100.0	100.0	100.0	(00.0	;00.0	100.0	100.0
SUIDAL		Lin Tri	1. 1.1.12	- 195-	- (Pg.) +	and a fait of the			

Table 95

1.32

Mean Gross Annual Earnings - Full-time Female Employees

(Earnings of full-time male employees in each group = 100)

	CEHMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXIM. BOURG	UK	IRFLAND	DENMARK
 2 3 4 5 6 (*)	- 88,8# 82,7 83,3 87,3 81,4	73.7 87.6 88.6 93.0 95.4 93.1	92,1# 95,2 94,2 97,8 84,7 85,2	61.2 # 47.3# 73.0 70.2 74.6	- 85.1 88.2 83.5 82.4	- 81.3 83.6 74.0 81.1	- 71,9# 77,4 76.9 71,2	92.9 90.9 91.2 54.7#	89,8 91,8 97,5 85,1
ALL	70.4	70.1	77.9	56.Z	70.2	62.6	53.4	60,3	68.0

SOURCE TABLE C4 COUNTRY VOLLMES

4.7 Length of Scheduled Working Week

The figures which have been obtained for the length or scheduled working week for male and female employees are shown below in Table 96. These have been obtained using the terms and definitions which were set out on p of this report (and in section 8.11 of Volume 1). One of the main points of interest in the figures which are given in Table 96 is the position of the UK and Ireland where the proportions working shorter scheduled hours are noticeably higher for both men and women than elsewhere. There is little

variation between the figures for men and women for each country. For Germany, France and Luxembourg

there is great concentration into the 38-40 hours grouping, and for Italy similarly in the 36-38 hour group. Netherlands, Belgium and Denmark also have 38-40 hours as the modal group but with less concentration than elsewhere.

TABLE 96:	LENGTH OF	SCHEDULED	WORKING	WEEK
FULL TIME	EMPLOYEES		•	

MA	LE									•
HO	URS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
\$	36	-		0.8	•	11.9		83.5	83.9	6.9
>	36-38	0,3	0.7	91.5	3.0	<u>22.</u> 6	4	10.6	3.8	29.3
>	38-40	98.0	97.4	6.2	86.4	63.5	99.2	1.8	11.6	60,9
>	40-42	1.2	1.2	0.2	10.6	1.9	~	•		2.4
>	42-44	0.1	0.5	0.0	-	-	٤	-	•	•
> .	44-46	0.2		0.8	•	-	•	-	•	•
> >	46-48 48	0.2	0.1	0.3		-	-	-	•	-
TO UN FE	ITAL IDISCLOS	100 ED	100	100 (0.1)	100	100	100	100 (3.8)	100	100
но	OURS	GERMANY	I-RANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
*	36			0.2	•	9.9		90.3	92.7	9,9
>	36-38	0.5	0.9	9 5.5	4.7	19.8	. •	8.4	2.0	28.2
>	38-40	98.7	97.8	3.7	81.4	68.9	9 9. ē		5,2	58.6
>	40-42	0.7	1.1	J	13.7	1.3	. بعن	-	•	3.2
>	42-44	•	0.1	0.3	-	-	•	•	-	· · ·
>	44-46		0.1	0.1	•	**	۰.	and .	• •• 1	-
>	46-48	• -	•	v	-	-	t -	-	-	-
T	TAL DTAL DISCLOS	100 SED	100	100 (0.1)	100	100	100	100 (0.7)	100	100

SOURCE TABLE AS COUNTRY VOLUMES

4.8 BONUS PAYMENTS

The figures are given in Table 97 and the patterns for male and female employees within each country are similar, though in Ireland proportionately far more women than men are in the 0 per cent bonus category, It is interesting to note the figures for Denmark, Ireland and the United Kingdom with the concentration in the "O per cent" bonus category and also the very small (or lack of) entries in the categories of bonuses of 15 per cent and above in each of these countries. At the other extreme the figures for Italy are noticeably high for both male and female employees in the 30 but less than 50 per cent category. Italy has also a "high" percentage in the 20 to less than 30 per cent bonus category a feature also noticeable in France. In Germany, and for males in the Netherlands (and to a lesser extent in Belgium) the main concentration comes in the 15 to less than 20 per cent bonus category. The 10 to less than 15 per cent bonus category is the largest single category for female employees in Netherlands, Belgium and Inxembourg (where the main concentration for me is in the 5 - less than 10 per cent category).

TABLE ST BONUS PAYMENTS

LD .	~ - · ·		·	•		. (FULL-T	ine Enploy	555) % .
% BONUS	GERMANY	FRANCE	ITALY	NETHS.	BELG	LUX.	UK	IRELAND	LIEIMAE
-0	0.6	3.6	-	0.5	2,2	1.3	62.2	44,4	79.
>0-5	2.1	3:2	1.2	2.3	3.5	0.1	22.1	444	141
25-10	14.0	6.9	1.6	10.8	710	27.5	141	4.8	41
7/10-215	27.2	7,6	4.5	211T	25.4	25.4		113	
215-20	40.7	13,3	516	2012	23.1	13.0		-	
720-30	12.1	42.4	35.6	(617 7 A	10.6	1.9			•
≥>3050	2.8	20,6	43.1	3.0	1.8	0.5	•		•
> 50	0.5	215	2.6	0+1			۰.	•	
(*)									
	100.0	100-0	100.0	100.0	100.0	100.0	100.0	109.0	100
PIALE									
% BONUS	GERMANY	FRANCE	ITALY .	NETHS.	БELG.	LUX.	UK	IRELAND	DENMA
	0.4	4.5		1.0	2.9	2.2	68.5	63.8	83
v د ۵ <u>–</u> 5	1.7	2.8	1.4	4.6	2.2	0,1	20.2	30.4	15
ע=י ז₀5–10	17.3	6.4	11.6	1519	11.7	24.6	81	4.4	0,
·7/10-15	29.8	8.1	4.8	31.1	32.7	21.0	4.0	0.6	
2/15-20	42.7	12:5	6.1	28.0	11012	2411			
/ 20-30	7,3	50.7	27.8	10.1	1912			÷	
7.30-50	0.7	13.9	45.9		214	0.5			
7 50	0.1	0.9	2.4	015	0.7				
(19)	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100
		+		·	\$	4		·	1

. .. .

SOURCE TRBLE B4 COUNTRY VOLUMES

.

.

•

4.9 Part-time Employees

The final section of this chapter is concerned with the characteristics and earnings of parttime employees. The comments applied to the definition of part-time employees discussed in the previous chapter (and in the Technical Annex) apply also to these figures.

Figures were given earlier on part-time employees in Table 74 but it is useful to consider them separately in Table 98 below:

.	Kunber	a (000°s)	As a percentage	of all employees
	All part-time	Part-time female	All part-time	Part-time female
Germany	54-1	51.6	13.5	12.9
France	7.0	5.4	2.3	. 1.6
Ttaly	2.2	1.7	0.6	0.5
Netherlands	10.6	9.3	11.9	10.5
Relginm	1.7	1.6	2.6	2,5
Invenions	0.1	0.1	1.9	1.9
United Kingdon	22,4	19.0	5.7	4.8
Ireland		-		
Denmark	9.2	8.2	23.1	20,6

Table 98 Part-time Employees

These figures have received comment in an earlier section (4.1), but it is worth noting the variation in the percentages of part-time employees from Italy and Ireland where the percentages are very low to Denmark which is considerably higher than elsewhere - a feature also noted for Denmark in Wholesale and Retail Distribution.

4.9.1 Length of Scheduled Working Week

The sample numbers on which many of the tables in this section on part-time employees are based are very small and the information collected is therefore subject to the problems of arge standard errors and disclosure which would breach confidentiality. As a result the tables for male employees in particular, contain many empty cells and in several instances are incomplete.

The general pattern, where figures are available however, is for the concentration of employees to come in the 24 hours or less category. The degree of concentration varies from one country to another and in all countries apart from Germany is greater for female employees than for male employees.

Some interesting features relate to the figures for both male and female employees in the Netherlands and the proportionately large percentages which occur in the 30-34 hours band. Also of interest are the figures quoted for Italy and Denmark in the 36 hours and over category.

TABLE 99 LENGTH OF SCHEDULED WORKING WEEK

									-
	MALE		· · · · ·	• • • • • •	· · · · · · · · · · · · · · · · · · ·	···· ·	. .	· -	
	C-COLLAUN	FOALLOT			0-000				10
HRS	GERMINI	PRANCE	I LALY	NEINS	BELGIUM	Lux	U.K.	FRELIN	DENMUICH
1	76.5	82.1	62.0	66.0			3815		50.5
7. 24-40	- 4.4	. 0	902	6.5		•••••	4		-
7 26-28						1000-	å 		
728-20		6.3	11.1			0			
7-30-32	6.6	8.7	10.8	24.3					
>32-34	*	4.8		9,9					- <u> </u>
734-36	3	•							1: 4
7.36		+	16:2						1819-
(*)	FEMAL						(385)		(235)
Hers.	GERMANY	FRANCE	TUAN	NETHS	BELGIUM	LUX	U.K	IRELING	DEWMIKK.
\$24	. 61.9	83.8	89.0	68,1	72.2	68.8	69.9		79.0
>.24-26	20.0	2.9	5.3	2.7	•	12.5	6.8		3.0
7:26-28	3,9	1.6	•	2.6	7.7	0	•	_	2.7
7:28-30	6.7	3,3	1.8	6,1	8,6	÷	1	-	1.7
7 30-52	8,4	4.6	2.1	20.0	9.0	J	y		3.3
> 32-24	4.2	2.1		14.0	•	•		-	2.2
2:24-26	1.9	2.7	•	4.6		,		-	
7.36	0.5	0	4	-				-	3.4
(*)		(2.7)					(18.5)		(7.1)
	MAIE	AND P	EMPLE				C*0.77	:	01
			<u>vijor</u>						10
Hes	C.F.P.MANY	ERGINCE	TTALY	NETHS	BELGIUM	LUX	U.K.	FRELAND	DENMACK.
5 74	62.6	83.6	83.2	67.9	72.6	69.4	65.1		75.9
> 24-76	19.5	2.4	6.1	3,1		11.1	64-	-	2.9
2.74-24	3.1	1.6		2.4	7.8	•	3	-	2:6
2:28-30	6.6	4.11	3.8	5,7	8.6	1	,	-	1.6
7 20-27	<u> </u>	5.5	4.0	20.5	9.0	4,3	•		3.4
21.27-34	4.A	2.7		135		\$	-		2.2
24-26	2.1	2,5		4.7		2			ð
> 21 26	1.5	1	5.3						5.1
<u> </u>		12.0					(a, -1	er∰t - 10∓	
C */		· (611)					(215)		(8.4)
	t,		•						

SOURCE TABLE AS COUNTRY VOLUMES

.

,

4.9.2 Age

The average ages in years of male and female part-time employees are given below:

	Male	Female
Germany	52.6	37.8
France	50.5	43.3
Italy	40.5	44.9
Netherlands	33.7	32.8
Belgium	50,2	36.6
Luxembourg	60.8	34.3
United Kingdom	57.1	40.7
Ireland	NOT QU	OTED
Denmark	46.7	35.9

As might be expected, and as has been noted in most countries in the previous sectors the average ages of part-time employees are higher than those for full-time employees. There are however considerable variations both between the averages for male and female employees in the same country and amongst the averages for employees of the same sex from one country to another. For male employees, the averages for Luxembourg and the United Kingdom are high, while those for Italy and the Netherlands in particular are low. A similar variation is observed in the averages for female employees. In particular, those for France and Italy are higher, while the Netherlands takes on a lower value.

The frequency distributions are given below in Table 100 and though there are problems resulting from small sample sizes they show up the variations about these mean values.

TABLE 100 AGE DISTRIBUTION - PART TIME EMPLOYEES

MALE									
YEARS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRELAND	DENMARK
< 21	-	•	s	25.0	.	-	-	~	14.2
21-29	•	9.6	16.4	19.8	••	-			12.1
30-44	13.4	16.7	35.3	21.7	•	•		~	10.6
45-54	19.3	17.8	29.9	11.6	•	-		-	13.7
≱ 55	59.1	51.7	17.4	22.0	•	91.7	73.1	-	49.5
TOTAL UNDISCL	(100) OSED	(100)	(100)	(100)	(100)	(100)	(100)		(100)
FEMALE Years	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK	IRFLAND	DENMARK
< 21	0.9	2.4	-	14.7				_ ·	1.3
21-29	17.9	12.4	2.6	27.0	22.3	34.4	9.6	<u> </u>	29.4
30-44	46,5	31.2	32:0	31.3	43.9	32.3	43.2		41.7
45-54	24.8	27.3	51.4	19.7	17.9	21.9	34.2		16.9
> 55	9,8	25.7	14.0	6.9	13.5	•	11.0	-	10.7
TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)		(100)

UNDISCLOSED

SOURCE TABLE AS COUNTRY VOLUMES

When considering them it should also be remembered that there are considerable variations in the percentages of part time employees in different countries. (Denmark has the highest percentage (23.1) of part time employees, with Germany and the Netherlands also high with 13.5 and +1.9% respectively. Apart from the UK with 5.7% the others fall below 3%) The majority of part time employees are female and in the younger and middle range age groups (21-29 and 30-44). Netherlands has a noticeably higher figure in the under 21 years age group.
4.9.3 LENGTH OF SERVICE

The percentages are shown in Table 101 and a comparison with the figures for full-time employees given in Table 84 is interesting. Where sample sizes have permitted disclosure of the figures, there is evidence in all countries apart from Italy of a greater concentration of male employees into the shortest length of service category for part-time employees than was seen full-time The degree of concentration varies employees. from one country to another and is most pronounced in the Netherlands and Denmark. Italy, (where the proportion of part-time employees is very small in any case) exhibits a different pattern with a surprisingly large percentage (40,2) in the 10-19 years length of service category.

For female employees the patterns of some countries are in agreement with those for full-time and for other countries there are differences. The similar patterns occur in Germany, the Netherlands, Belgium and Luxembourg. In France, the concentration of part-time employees into the shortest length of service category is greater than for full-time employees and in Italy, the United Kingdom and Denmark it is less.

TABLE OI

LENGTH OF SERVICE

. ••

MALE

								and shares of the state of	THE REPORT OF THE OWNER WATER OF THE OWNER OWNER OF THE OWNER
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 ≥ 20	27.6 32.8 17.8 9.4 12.5	25.1 22.8 19.1 20.4 10.9	14.2 11.4 18.2 42.2 13.9	52,2 27.6 12.5	• • •	• • • •		1111	44.0 30.1 16.7 —
ALL	100.0	100.0	100.0	100.0	100-0	100.0	100.0	100.0	100.0

..

FEMALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
< 2. 2 - 4 5 - 9 10 - 19 > 20 (*)	20.0 34.9 28.2 13.9 2.9	29.9 28.1 23.6 13.3 2.4	18.3 12.9 27.4 30.5 10.9	38.9 31.8 21.2 6.8 1.0	23,1 26,6 25,4 23,3	38.5 37.5 18.8	29.5 28.8 33.6		20.8 31.5 28.7 15.7 2.4 0.9
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

· · · · · · · · · · · · · · · · · · ·									
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 19 > 20 (*)	20,4 34,8 27,7 13.7 3,3	28.8 26.9 22.6 14.9 4.3	17.4 12.6 25.4 33.0 11.5	40,5 31,3 20,1 6,4 1,3	22.9 26.1 25.6 23.5	38.9 37.0 17.6	27.9 29.7 30.2 7.0		23.4 31.4 27.4 14.7 2.3 0.9
ALL	(00.0	100-0	100.0	100.0	100.0	100.0	100.0	(00.0	100.0
SOURCE	IAT	BLE A4	<u></u>	COUNTRY	VOLUMES	•			

206

%

4.9.4 <u>Occupational Qualification Group</u>

The percentages of employees falling in the various occupational qualification groups is given in Table 107 Because the sample sizes have been very small, in particular for male employees, the tables have many empty cells and because of the need for confidentiality and distributions are also incompletely specified.

For male employees, where figures are given, the concentration is in the lowest occupational qualification group (category 6) with as might be expected the degree of concentration varying from one country to another. It is of interest to see, though, that in France 5.9% are classified in category 1 and 10.9% are classified in category 2 in Italy.

For female employees, the concentration also comes in the lower occupational qualification groups, though the concentration is not so pronounced in some countries (eg Germany, Netherlands and Denmark) as for male employees perhaps reflecting the different nature of the part-time jobs for male and female employees. As was the case for male employees, the percentage occurring in category 2 for Italy is of interest.

TABLE 102

DCCUPATIONAL QUALIFICATION GROUP

HALE								·	6
	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	ŪK	IRELAND	Djanmank
1	-	5.9	0	ø		1	\$	-	
2			10.9	•	•	-			•
3	•	7.8	. 8.0	•	-	-	•	-	•
4	17.6	11.3	21.4		•			-	ų.
5	19.3	21,4		416	0	•			9,4
6	58.0	51.4	56.4	85,2		.	53.8	-	83.1
(*)									
ALL	100.0	100.0	100.0	163.0	100.0	100.0	100.0	_	100.0

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGTUM	LUXEN- BOURG	UK	IRELAND	DENMARX
1 2 3 4 5 6 (*)	- 1.6 19.9 29.5 48.9	2.0 4.4 14.1 78.8	9.1 2.0 87.0	1.7 17.1 80.9	- 15,4 19.4 62.7	- 10:4 18:8 67.7	- 37.0 62.3		- 3.6 27.1 69.3
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	-	100.0

MALE AND FINIALE

	GERMANY	FRANCE	ITALY	NETHER LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DERMARK
1 2 3 4 5 6 (*)	- 1.8 19.8 29.0 49.4	1.4 1.0 3.3 6.0 15.7 72.7	9.5 2.4 6.2 1.0 80.6	• 1,8 16,2 81,4	15.0 19.0 62.9	- 9,3 22,2 65.7	• • • • • • • • • • • •		- 3,7 25,2 70,8
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0		100.0
SOURCE	· · · ·	TABI	e Al	208	COUNTRY	. Aomineer			·

Indices

The only indices of average earnings which have been calculated for part-time employees are given in Table 103 and 104. Table 103 gives the indices by occupational qualification group of mean gross monthly earnings and Table 104 gives the figures for mean gross annual earnings.

For male employees there are entrires only for Germany, France and Italy and all apart from categories 4 and 6 in Italy are subject to the standard error of the estimate of the mean lying between 5 and 10 per cent. /This is not surprising given the small members of part time male employees recorded (of Table 74 of this chapter). It is also interesting to note the percentages of part time employees - both male and female in category 2 in Italy, 10.9 per cent for men, 9.1 per cent for women7. Given the availability of the figures, and the fact that the majority of part time employees are female we can see that the greatest variability over categories 4, 5 and 6 (for female employees) occurs in Belgium (though the figures are subject to large standard errors) and the figure for France for category 4 is not available and that for category 5 this is extremely high). The least variation appears to be for Italy but Inxembourg could well hold that position given the

-INDEX OF MEAN GROSS NONTHLY, EARNINGS - BY OCCUPATIONAL QUALIFICATION CHOUP TABLE 103

.

KALE

	GERMARY	FRANCE	ITALY	NETHER- LANDS	PELGIUM	LUXEN- BOURG	UK	IRELAND	PIPNMARK
1 2 3 4 5 6 (*)	- 120,7# 115,9# 83.2#	57,2# 66.C#	158.8 107.6 76.6				•		83,5
ALL	100.0	100:C #	100.0		•	100.0	•		100

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- Lands	Belgin	LUXEN- BOURG	UK	IRELAND	DENMARK
1 2 3 4 5 6 (*)	- 176.1 136.2 110.7 76.3	i84.4# 71.0	204.3 123.7 87.5		- 194,3# 135,0# 670	158,7 99,4 86.2	 112.7 91.9		- 30,3 22,3 89,7
ALL	100.0	100.0	100.0	9	100.0	100.0	100.0		100.0

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	Nether- Lands	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARX
1 2 3 4 5 6 (*)	- 74,5 35.8 11.0 76.6	150.8⋕ 134.7⋕ 66.3	206.4 176.1# 118.9 85.4		18 7.9# 130.1# 66.4	153,5 100.0 88.3	, 100.0 80:9		144.6 124.6 88.8
ALL	100.0	100.0	100.0		100.0	100.0	100.0#		100.0
SOURC		TABI	.5 09	210.	COURTRY	vounes			

INDER OF MEAN GROSS FRAMME EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE (04 ł

...

·, ·

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND [*]	DENMARK
1 2 3 4 5 6 (*)	121.2# 120.0# 80.6#	。 。 。 。 3#	107.9# 77.4		•		• • •		• • 84.3
ALL	100.0	٥	100.0		Ö,	3			100.0
FEMALE		•					t		
,	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
1 2 3 4 5 6 (*)	 175.3 136.6 111.2 74.6	209.1# 195.1# 66.8	210,5		- 49,3# 61.4#	- 151,3 91,1 84,6	- 116.1 91.1#		- 122.9 125.8 89,1
ALL	100.0	100.0	100.0		100.0#	(00.0	100.0		100.0

MALE AND FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM BOURC	UK	IRELAND	DENMARK
 1 2 3 4 5 6 (*)	- 173.0 136.1 111.6 74.8	i50.5# 136.3# 63.3	215,3 106,4# 83.7		202,5# 140.4# 60:2#	- 146.7 93.4 87.6	° — 100.2 77.8#		135.2 126.9 88.6
 	100.0	100.0	100.0	-211-	100:0#	100.0	100.0#]	(00.0

ł

lack of an entry for category 4 there.

Where annual figures are given, these tend to be in good agreement with the monthly ones.

4.9.5 Size of Underthing

The distribution of employees over differently sized undertakings is given in Table 105. It is interesting to note the different patterns which occur both for male and female employees and from one country to another.

13

Where the figures are given, for male employees the concentration of part-timers is mainly in the largest size category of undertaking in the Netherlands, Italy, Denmark and the United Kingdom. In France, and to a slightly lesser extent in Germany, the concentration comes in the smallest size category.

For part-time female employees in all countries apart from France, and Luxembourg - which is a special case, the concentration comes in the largest size of undertaking category with the degree of concentration varying across countries - the United Kingdom the most pronounced and Germany the least. In Italy

more than 80% of part-time employees (who form a small percentage of all employees) fall either into the largest size of undertaking category or the smallest. France, on the other hand, exhibits a pattern where employees are fairly evenly distributed across the different size category (100-199 employees being the exception).

TABLE 105 SIZE OF UNDERTHIKING

MALE

·····				• • • •			%		
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49	34.3	30.7	8.2	8.0	•	. 0			4.1
50-99	13.1	21.9	•	7.7	-	•	. •		
100-199	10.1	7.6	8.0	•	•	•	. •	-	
200-499	9,6	9,4	10.2		•	-		-	
500-999		11.6	•	4.0	•	•		·	18.4
≥1000 (*)	25,4	18,9	63.7	76.4		-	46.2	-	60.0
ALL	100.0	100.0	100.0	100.0	100.0	(00.0	100.0		(00.0

...

FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	10.7 8.2 4.2 15.9 9.5 46.5	17.2 13.7 5.8 16.2 23.8 23.3	31.1 3.8 3.4 9.1 51.4	15,7 4,6 1,5 4,5 2,2 71,5	3.2 7.2 6.1 19.3 61.7	20.8 15.6 31.3 31.3	6.8 79.5	11111	4,8 6,4 4,7 7,5 12,2 64.5
ALL	100.0	(00.0	(00.0	100.0	100.0	i00.0	100.C		100.0

MALE AND FEMALE

NO, OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	11.8 8.5 9.2 15.6 9.4 45.5	20,2 15,6 6,2 14,7 21,0 22,3	26.2 4.4 4.4 9.3 1.7 54.0	14.8 5.0 116 4.1 2.5 72.1	3.9 6.9 7.1 18.3 59.8	21.3 14,8 33.3 • 29,6	6.4 6.4 74.4	11111	4,7 6,3 4,7 7,5 12,8 64.0
ALL	(00.0	105.0	100.0	100.0	100.0	100.0	100.0		100.0

SOURCE

TABLE AZ

COUNTRY VOLUMES

• •

<u>ہ</u>۔

Chapter 5:

Insurance NACE 82

This chapter considers the earnings of employees classified under the NACE heading 82. - Insurance, except for compulsory Social Insurance. An important characteristic to note when interpreting these figures is that in Germany, Belgium, the United Kingdom and Ireland, employees in the Insurance sector account for the order of 7% of employees in the "services" sector (as defined in the survey). The equivalent figure for Italy and the Netherlands is 5%, for France and Denmark 4% and for Iuxembourg 3%.

5.1 Some Overall Features

The figures for full-time plus part-time employees, full-time, and part-time employees, full-time female employees, and part-time female employees are given in Table 106 . These are also given as percentages of all employees (full-time plus part-time). The percentages of part-time employees (the majority of whom are female) show interesting variations, with Denmark (23.5%) much higher than elsewhere and Italy (0.4%) much lower (if we exclude Lixembourg and Ireland where the percentages are too small to be recorded). The United Kingdom, Germany and Netherlands have

similar percentages of part-time employees (10% to just over 12%) and France and Belgium similar to each other but at a lower level (1.4 and 2.3 per cent respectively). The percentages of full-time female employees also show interesting variations with France (58.7%) noticeably high and Netherlands low (23.2%), Italy, too, is low with 34.9% and apart from Ireland and Luxembourg the other countries have percentages in the range 40-45%.

TABLE 106 EMPLOYEES 000'S

and a subscription of the	and the second state of th	متعلومون وموندات وتراوين	Part Artistic Agencies				and the first of the state of the	and a state of the second second	
	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXELL- BOURG	Un	TRELAND	DEEMARK
FULL-TIME AND PART-TIME	136.7	82.8	54.5	28.9	22.0	0.4	154.1	5.2	9.8
FULLTIME	121.4	81.6	54.3	25.5	21.6	0.4	165.0	5.2	7.5
PART-TIKE	15.3	1.2	0.2	3.5	0.5	•	19.1		2.3
FULL—TIME FEMALE	60.0	48.6	19.0	6.7	9.1	0.2	74.0	2,6	4.2
PART-TILE FENALE	14.6	0.9	0.1	3.0	0.4	•	17.0	-	2.2

FULL-TIME AND PART-TIME	10	100
-------------------------	----	-----

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	UK	IRELAND	Denmark
FULL-TIME AND PART-TIME	100	100	100	100	100	100	100	100	100
FULL-TIME	88.8	98.6	99.6	87.9	97.7	100	89.6	100	76.5
PART-TIME	11.2	1.4	0.4	12.1	2.3	•	10.4	-	23.5
Pull—Time Female	43.9	58.7	34.9	23.2	41.4	50.0	40.2	50	42.9
PART-TIME FEMALE	10.7	1.1	0.2	10.4	1.8	•	9.2	-	22.4

SCUHCE

COUNTRY VOLUMES

5.2 The Sector by Size of Undertaking

The figures are given in Table 107 . In Germany and the United Kingdom there is a very noticeable concentration of employees into the largest size (1000 employees or more) category. France and Denmark show a similar if less pronounced distribution. Italy has an interesting distribution with the largest single percentage appearing in the 1000 employees or more category but also having noticeably large percentages in the 10-49 employees and 200-499 employees cadegories. Belgium has just over 72% of its employees in Insurance distributed almost evenly over Ireland. the top three size categories. is interesting in that the concentration of employees tends to come in 200-499 and 500-999 employee size categories.

TABLE 107:	ENPLOYEES	by size	OF ENTE	RPRISE (FULL-TINE	+ PART-	tine en	PLOYEES)		
SIZE OF UNDERTAKLIC (NUMBER OF EMPLOYEES)	W GERNARY	FRANCE	ITALY	NETHS	BELGIUM	w.	UK	IRELAND	DENMARK	50
1049	1.3	2.9	21.3	5.9	6.3	29.9	2.9	2.7	7.0	
50-99	2.1	5.1	5.8	5.7	7.7	-	1.0	5.6	9,6	
100-199	3.8	5:8	7.7	12.5	13.8	70.1	2.5	13.7	4.6	
200-499	10.7	12.4	24.7	18.6	22.9	-	6.4	38.0	5.2	
500-999	11.7	19.3	5.6	672]	26,1	-	10.4	29.8	19.1	
≽ 1000	70.4	54.5	54.8	5 ".3}	23.2	-	76.8	10.2	54.6	
TOTAL	(100)	(100)	((:)0)	(100)	(100)	(100)	(100)	(100)	(100)	
NUMBERS .000 ' 6	136.7	82,8	54•5	28.9	22.0	0.4	184.1	5.2	9.8	•••

SOURCE TABLE AS COUNTRY VOLUMES

5.3 <u>Distribution of the full-time employees</u> labour force according to gross monthly earnings

> The methodology of this section is the same as has been used in the corresponding sections in previous chapters (2.4, 3.4, 4.4) and the graphs produced are given in Fig 4 As has happened in previous sectors, the graphs illustrate the differing percentages of full-time female employees in the full-time labour force from one country to another.

Graphs

These have been constructed from the frequency distributions of gross monthly earnings for full-time male and female employees in each country.

Relative Deviation, $u = (x-\bar{x})/\bar{x}$

∆u= ∆x/_x

<u>Scaled Frequency</u>, $Y = f/_{\Delta u}$ where f is the percentage frequency. The values of Y have been scaled such that the area between the graph and the horizontal axis in each case is proportional to the number of full-time male/female employees.

•••• FEMALE











Table 108

Pull-time	female	employees	a8	a %	of	all	full-	-time
employees								
Contraction of the second s				1	•			•

Germany	49.4
France	59.6
Italy	35.0
Netherlands	26.3
Belgium	42.1
Luxembourg	50.0
United Kingdom	44.8
Ireland	50.0
Denmark	56.0

Source: Table A1 Country Volumes As has been noted earlier in section 5.1, the figure for the Netherlands is low while those for France and Denmark are high. In general, the graphs are positively showed. In the case of female employees the graphs are unimodal, and with less relative variation than for their male counterparts in each country. The variation is particularly small in Denmark and Germany. A similar situation exists for male employees though in the United Kingdom the distribution is very flat and almost has two equal peaks. In Ireland, too, the feature of two almost equal peaks is observed. As has been noted for other sectors very few female employees reach the highest salary levels. The variation of earnings about the respective mean values which has received comment earlier is summarised in the table of coefficients of variation given below.

Coefficients of Variation (%)

Table 109

Gross Monthly Earnings of full-time male and female employees

Male Employees	Female Employee
32.3	23.2
62.5	41.7
53.8	28.7
48.4	45•7
41.3	32.0
43.3	41.6
62.1	36.6
49.1	33.5
38.7	21.7
	Male Employees 32.3 62.5 53.8 48.4 41.3 43.3 62.1 49.1 38.7

5.4 Gross Monthly Earnings as a function of the structural characteristics of the labour force full-time employees

> This section of the report considers the effects which the various characteristics of the labour force have had on the distribution of gross monthly earnings around the mean value in each case. The characteristics to be considered are as follows:-

ο.	RAT	
-		
b.	age	
c .	length of service	
đ.	occupational qualification	group
8.	size of undertaking	
U .	BING OF ARMALAGETINE	

The notation, and definitions of the various indices to be considered are the same as for the previous chapters.

The notation, and definitions of the various indices to be considered are the same as for the previous chapters.

5.4.1 Employees classified by sex

The previous section gave in pictorial form an illustration of some of the differences between the earnings of male and female employees. Table 110 below gives a summary of these characteristics.

Table 110

Summary of Female/Male Mean Gross Monthly Earnings. (Full-time employees, mean gross monthly earnings of full-time male employees = 100)

Germany	71.7
France	67.6
Italy	71.1
Netherlands	60.4
Belgium	68.4
Luxembourg	54.4
United Kingdom	50.1
Ireland	58.9
Denmark	63.1

These figures follow a pattern which is similar to those observed in other sectors, but with the notable exception that the ratio for Italy is no longer much higher than in other countries. A comparison with the Hanking sector, perhaps more closely related to Insurance than the other sectors, reveals that only in Germany and the Netherlands is the ratio in Insurance

higher than in Banking.

These are however global figures and we now go on to consider the effects which the various characteristics of the labour force have upon them.

5.4.2 Employees classified by Age

The average age in years is given below for full-time male and female employees:-

	Male Employees	Female Employees
Germany	37.6	31.2
France	35.3	31.0
Italy	34.7	28,4
Netherlands	36.5	25.5
Belgium	36.1	27.9
Luxembourg	33.6	22.8
United Kingdom	36.2	27.8
Ireland	31.8	25.3
Denmark	39.8	33.5

Source Table A6 - Country volumes

These averages are interesting both in respect of comparisons between male and female employees and also in respect of comparisons with the average ages quoted in previous sectors. For all countries the averages for male employees are higher than for female. For Germany, the Netherlands, Belgium, Luxembourg and Denmark, the averages for male employees are higher in the Insurance sector than in any of the other sectors, and this is true also for female employees in the Netherlands and Denmark.

For male employees the averages may be put into three broad groupings: Ireland by itself with average age 31.8 years, Denmark also by itself with average age 39.8, and the remainder with (rounded) averages in the range 34-38. Similarly, for female employees we may use three groupings:- Iuxembourg, by itself with an average of 22.8 years; Italy, the Netherlands, Belgium, the United Kingdom and Ireland with averages (rounded) in the range 25-28 and the remaining three countries Germany, France and Denmark with averages (rounded) in the range 31-34 years.

Table 111 , which gives the frequency distributions illustrates the different patterns behind these averages.

TABLE 111 AGE DISTRIBUTION - FULL TIME EMPLOYEES

|--|

							•		. %
YEARS	GERMANY	FRANCE	ITALY	NETHS	BELOIUN	TIX	UK	TRELAND	DENHARK
≪21 21-29 30-44 45-54 ⋧55	3.1 19.6 43.1 20.4 13.8	6.9 32.8 27.9 16.9 15.3	1.3 26.3 49.4 17.1 5.9	6.8 27.3 28.8 22.5 14.5	4.8 24.3 39.5 21.9 9.5	5.8 31.9 40.7 10.6 11.1	7.8 24.5 33.8 18.9 15.0	15,6 28,6 35,4 11,8 8,6	20.3 36.0 23.2 19.0
	(100)	. (100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
UNDISCI	LOSED	0.3		5-43 					
YEAR3 < 21 21-29 30-44 45-54 ≥ 55	FEMALS GERMANY 13.3 35.5 28.5 17.1 5.7 (100)	FRANCE 14.0 39.9 23.3 15.4 7.2 (100)	ITALY 8.3 48.7 32.8 9.3 0.9 (100)	NETHS 37.3 35.4 12.9 9.7 4.7 (100)	BELGIUM 16.8 44.2 26.7 9.7 2.6 (100)	100 38.2 44.0 10.5 6.8	UK 31.0 33.2 14.1 15.7 5.3 (100)	IRELAND 31.7 41.8 14.2 9.4 2.9 (100)	DEHMARK 6.7 39.7 25.2 16.4 12.1 (100)
UNDISCI	LOSED	0.1							

Source: Table A5 - Country Volumes

For male employees, the lack of a figure in the under 21 years age group in Denmark is of interest as is the percentage in that age group in Italy. On the other hand, the percentage in that group in Ireland is particular high. At the other end of the age range Italy and Luxembourg have noticeably lower percentages of male employees aged 45 years and older and Denmark a noticeably higher figure.

For women there are also considerable variations in the percentages in the youngest age group with Italy and Denmark low, and Netherlands, Luxembourg and the United Kingdom and Ireland (both to a lesser extent) high. Apart from in Netherlands, the age group 21-29 years has uniformly the largest proportion of women in it. The percentages falling in the 30-44 years age group also show interesting variations (similar to those in the Banking sector) with the United Kingdom, Netherlands and Luxembourg lower than elsewhere. In the age group 45 years of age or older the percentage for Denmark (28.5) is noticeably higher than elsewhere, and those for Italy, Belgium and the Netherlands noticeably lower.

Indices of Mean Gross Monthly Earnings - Employees classified by Age and Sex

Two main sets of indices are considered, one giving comparisons between groups of employees of the same sex, the other giving comparisons between male and female employees of the same age. A further table considers the particular case of the earnings of those aged under 21 years with those aged 21-29 years of age.

Table 112 gives the comparisons between employees of the same sex.

For men the index rises from values in the range 38-60 for the under 21 years age group (it is interesting to note that Denmark has recorded no value here) to values in the range 107-140. In Germany this peak comes in the 30-44 years age group, in Italy and Luxembourg in the 55 years or older age group, and in all other countries in the 45-54 years group.

For women the index rises from values in the range 65-80 for the under 21 years age group to values in the range 112-179. These peak values occur in the 55 years and older age group in France, Belgium and Denmark and in the 45-54 years group in all other countries, (the figures for Luxembourg, and to a lesser extent, Netherlands, being noticeably higher than elsewhere).

AGE IN YFARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUN	LUXEN- BOURG	UK	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54	59•7 80•6 107•5 106•1 102•9	53.7 72.1 114.4 132.9 115.5	52.2 72.9 100.1 131.1 138.8	42.9 72.1 111.0 126.4 117.0	45.2 ' 69.0 107.1 124.9 119.5	38.1 72.6 112.2 132.0 138.9	47.8 74.8 117.1 120.8 103.2	46.4 75.0 122.3 144.8 130.9#	72.7 102.5 125.1 99.4
(*) (*) A).J.	100	100		100	100				100

FEMALE

AGE IN YEARS	GEEMANY	FRANCE	ITALY	NETHER- LANDS	BELGUIM	LUXEM- BOURG	UK	IRELAND	DEIMARK
< 21 21 2) 30 44 45 54 ≥ 55 (*)	77•5 93•1 109•7 112•0 110•3	76.6 89.9 111.2 119.2 120.4 67.9#	79•3 92•0 106•5 132•9 123•8	65.2 98.7 139.7 160.3 159.4	67.2 89.3 119.9 136.2 148.6	68.8 101.8 144.2 178.9	75.9 105.3 115.5 122.8 102.6	75.6 92.6 139.4 152.6 135.5 #	73.6 91.5 104.7 109.6 117.6
ALL	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

...

;

•		and the subscription of the				A DESCRIPTION OF THE OWNER OF THE		CONTRACTOR PORTAGE VALUE	Contraction of the second states
AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGUIN	LUXEN- BOURG	UK	TRELAND	DENMARK
i nacemanication of the second s	65.5	64.7	61.6	45.3	52.7	47.6	51.9	56.5	58.0
21 - 29	83.5	80.3	76.9	76.1	74.4	80.5	81.9	78.8	78.3
30 - 44	111.8	115.C	104.2	119.8	114.0	134.5	131.7	139•3	107.7
45 - 54	109.9	127.4	136.3	136.7	134.8	151.6	124.0	152.5	123.5
≥ 55	111.8	125.6	150.0	128.1	134.1	174.1	118 . 0 [‡]	149.4#	110.6
(*)		67.6#	-	•	-	_	-		NAME OF TAXABLE PARTY
ALL	100	100	100	100	100	100	100	100.0	100 ′

SCURCE

TABLE DS 232

232. COUNTRY VOLUMES

Of particular interest are the indices for the youngest age group. To illustrate the change in average earnings between this group and the next youngest group. Table 113 has been constructed. This table gives the mean gross monthly earnings of the under 21 years of age group as a percentage of the corresponding mean for the 21-29 years of age group.

Table: 113

Mean Gross Monthly earnings of full-time employees under 21 years of age

(Mean Gross Monthly earnings of full-time employees aged 21-29 = 100, male and female employees considered separately)

	Male	Female
Germany	74.1	83.2
France	74.5	85.2
Italy	71.6	86.2
Netherlands	59.5	66.1
Belgium	65.5	75.2
Luxembourg	52.5	67.6
United Kingdom	63.9	72.1
Ireland	60.5	81.6
Denmark	ä	80,4

Source: Table D9 -- Country Volumes

The figures reveal some interesting variations with, as in previous sectors the change with age greater for male employees than for female employees. For male employees the change is smallest in France, Germany and Italy and greatest in Luxembourg and the Netherlands. It is interesting to note that in this sector the value of the ratio for Italy is not the highest, as it was and noticeably so - in previous sectors. A value for Denmark has not been quoted for reasons of small cell size.

For female employees the change with age is smaller than for male employees with Italy, France and Germany showing particularly high values of the ratio and the Netherlands and Laxembourg values which are considerably lower than elsewhere.

Table 114 gives comparisons between male and female employees when the effect of age is removed.

For the under 21 years of age group the figures for Italy, Denmark and Belgium are noticeable in that they exceed 100. For all other countries apart from the United Kingdom the ratio takes values over The decline in the ratio values with increase 90%. age is most noticeable in the United Kingdom though similar patterns if less pronounced are observed for other countries. For Italy and Ireland there is a smooth decrease over the whole age range. In Germany, Netherlands and Luxembourg the decline is to the 30-44 age group, then an upturn follows. In France, Belgium and Denmark the decline continues into the 45-54 years of age category.

TABLE 114

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES (BY AGE) (MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY - 100)

AGE IN YEARS	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENMARK
< 21 21 - 29 30 - 44 45 - 54 ≥ 55 (*)	43.1 82.9 73.2 75.7 76.9	96.4 84.2 65.6 60.6 70.4	108.0 89,8 75.6 72.1 63.4	92.0 82.7 76.0 76.6 82.3	101.6 88.5 76.6 74.6 85.1	98,3 76.3 69.9 73.7	79.5 70.5' 49.4 50.9 49.8#	96.0 72.7 67.1 62.1 61.0 #	106.7 79.5 64.5 55,3 74.7
(") ALL	71,7	14,811 67,6	71,1	60.4	68.4	54.4	50.1	58.9	63.1

SOURCE

TABLE DS

COUNTRY VOLIMES

VOUNTIG TOTAL

5.4.3 Employees classified by length of service

The percentages occurring in the various categories are shown below in Table 115.

TABLE 115 : LENGTH OF SERVICE - FULL-TIME EMPLOYEES

	MALE			· ·					
YEARS	GERMANY 16.3 22.1 22.7 22.5 16.3	FRANCE 21.0 20.6 21.8 20.8 15.5	17ALY -15.6 25.7 26.0 22.5 10.2	NETHS 19.1 20.4 20.5 21.6 13.5	BELGIUN 14.1 17.8 18.7 27.0 22.5	LUX 16.8 24.3 20.8 22.6 15.5	UK 18.6 20.1 18.3 20.1 21.9	IRELAND 17.3 23.1 13.3 22.5 23.8	DENMARK 9.6 18.7 17.8 24.0 29.4
ALL	(100)	(100)	(100)	(100)	(100)	(100)	· (100)	(100)	(100)
UNDISCI	OSED	0.4							
YEARS <2 2-4 5-9 10-19 ≥20 ALL	FEMALE GERMANY 22.7 31.7 23.5 15.7 6.4 (100)	FRANCE 25.2 23.3 21.5 18.5 11.2 (100	ITALY 20.0 25.5 26.5 21.5 6.5) (100)	NETHS 39.0 30.4 14.6 8.9 7.2 (100)	BELGIUN 21.5 28.2 22.9 17.7 9.7 (100)	LUX 25.1 35.6 22.5 12.0	UK 35.3 29.3 15.7 10.4 8.4 (100)	IRELAND 28,9 31.7 17.4 8.7 13.2 (100)	DERIMARK 17.7 31.7 19.4 16.7 13.6 (100)
UNDISC	LOSED	0.4							

SOURCE:

TAPLE A4 COUNTRY VOLUMES

For male employees amongst the most interesting features are the high percentages in the longest length of service categories. Particularly noticeable in this respect is Denmark where 29.4% of men have served 20 years or more. When combined with the 10-19 years length of service Denmark still has the highest figure (53.4%) but is closely followed by Belgium with 49.5%, and Ireland with 46.3%.

For female employees, Netherlands and the United Kingdom came up highest in the shortest length of service category. Denmark and France and Ireland the largest percentages in the longest length of service category. It is interesting to note that when the two longest-serving categories are obtained, in France, Italy, Belgium and Denmark more than 25% of women have served more than 10 years.

Indices

Two sets of indices are considered - Table 116 gives comparisons of mean gross monthly earnings between employees of the same sex; Table 117 gives comparisons between male and female employees who have the same lengths of service.

For male employees, there is an increase (apart from Italy) in the values of the index from values in the range 57-90 for the less than 2 years length of service to values in the range 120-153 for the 20 years and over category.

For female employees, there is a similar increase in the index with increasing length of service from values in the range 69-88 for the shortest length of service category to values in the range 127-181 for the longest length of service category (with Netherlands at 180.1 noticeably higher than elsewhere).

The figures in Table 117 illustrate the position when the effect of length of service is removed.

SB TABLE 116 .

INDEX OF MEAN GROSS MONTHLY EARNINGS - BY LENGTH OF SERVICE

MALE

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM-	UK	IRELAND	DENMARK
< 2	84.9	65.7	89.9	65.9	65.9	60.8	66.7	57.5	74.8
2 - 4	89.0	79.1	83.7	82.2	77.4	93.2	87-1	70.1	78.1
5 - 9	97.9	94.2	95.5	95•9	92.2	92.7	98.3	92.2	86.8
10 - 19	108.9	119-4	111.1	118.1	112.4	113.5	108.6	114.4	111.0
> 20	120.0	152.5	143.1	138.4	130.1	145.9	133.9	150.2	123.3
(*)		•	-	•		- j			
ALL	100	100	100	100	100	100	100	100	100

ł.

FEMALE

ALL	100	100	100	100	100	100	100	100	100
(*)	-	94.2	-	-	-				
> 20	127-3	152.1	142.9	180.1	148.4	•	137.1	155.3	134.0
10 - 19	115.8	114.9	112.5	142.7	119.9	139-4	124.1	135.5	108-9
5 - 9	102.9	96.3	96.7	117.4	101.7	101.3	110.6	102.7	97.7
2 - 4	92.9	87.3	90.6	91.7	87.4	92.9	98.1	87.0	91.0
< 2	87.6	79.0	88.3	75.8	74.9	69.9	79.7	76.5	86.3
LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMAR

!

LENGTH OF SERVICE (YEARS)	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMAR
< 2	83.9	71.6	87.6	64.0	67.1	60.7	64 .6	62.3	76.5
2 - 4	88.3	82.4	85.6	81.4	78.3	88.9	85.9	74.4	80.7
5 - 9	99.7	95.5	95•7	101.5	94.1	94.3	104.2	93.2	91.4
10 - 19	114.4	118.7	111.9	127.2	118.2	128.8	122.5	131.2	114.6
≥ 20	130.2	157.0	147.1	150.4	141.9	173.5	152.2	162.9	137.0
(*)	•	85.7#		•	-	-	•		•••
AL Û	100	100	100	100	100	100	100	100	100

239

THELEDIO

SOURCE

For the United Kingdom, and to a lesser extent Luxembourg, the values of the ratios are lower than elsewhere. On the other hand the ratios for Belgium and Germany are both high, of the same order and show little variation with length of service. Apart from the value for the 2 to 4 years of service the figures for Italy are fairly stable also. Luxembourg shows some increase in ratio with increasing length of service (a non-uniform increase). France, Ireland, Denmark and the United Kingdom show the reverse pattern but the movement is not a smooth one.

TABLE 117

INDEX OF MEAN GROSS MOUTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES -BY LENGTH OF SERVICE

(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EMPLOYEES IN SAME CATEGORY = 100)

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
(10403) 2 - 4 5 - 9 10 - 19 $\gg 20$ (*)	74.1 74.9 75.4 76. 76.1	81.2 74.6 69.1 65.1 67.4 82.9	69.9 76.9 71.9 72.0 71.0	69,5 67.3 74.0 73.0 78.6	77.7 77.2 75.5 73.0 78.0 –	62.6 54.2 59.5 66.8 -	59.9 56.4 56.4 57.3 51.3	78.3 73.0 65.5 69.7 60,9	72.9 13.6 71.1 61.9 68.6
ALL	71.7	67.6	71.1	60.4	68.4	54.4	50.1	58.9	63.1

SOURCE

TABLE D10

COUNTRY VOLUMES
It is also interesting to note that in the majority of cases the values in the individual cells are greater than the value of the ratio of the overall means reflecting the different length of service distribution patterns for male and female employees.

5.4.4 Employees classified by the Occupational Qualification Group

The general remarks about the problems of classifying employees into broad groupings apply also to this chapter. In this case the broad groupings are:-

I. Managers - Senior management executives
II. Middle management executives
III. Junior executives and personnel with equivalent qualifications
IV. Highly skilled employees

V. Skilled employees

VI. Other wage-earning employees

A fuller explanation of these is given in the Technical Annex. With these definitions the figures given in Table 118 were obtained and indicate (at first sight) some interesting differences. When the figures for male and female employees are considered together, the percentages in category V1 (other wage-earning employees) alone show marked variations with Belgium, Netherlands and the United Kingdom high and Ireland, Denmark, Italy and Germany low. Even when the figures for category V (skilled employees) and category VI are combined the percentages for Netherlands, Belgium and the United Kingdom (of the order 60-68%) are noticeably higher than elsewhere and the figure for Germany (29.2%) noticeably lower. Iuxembourg has proportionately the most falling in category 1 (Managers - senior management executives) and even when this is combined with category 11 (middle management executives) the resulting percentage (15.8) is noticeably higher than elsewhere, the corresponding next highest figure being 11.4% in Denmark (Germany, on the other hand has 5% in the combined category 1 and 11). The high percentage in category IV (Highly skilled employee) in Germany is also of interest.

When the figures for male and female employees are considered separately these differences are accentuated. Proportionately more female than male employees fall into the lower categories (V and Vl) with noticeable veriations in the percentages involved (the extremes are 41.4% category V and Vl in Germany to 94% in the United Kingdom). The percentages of women reaching the highest categories (1 or 11) are very small and even when this definition is extended to cover category III (junior executives and personal with equivalent qualifications) only in France does the figure exceed 10%. In Germany there is a noticeable concentration in category IV.

For male employees there are similar variations from under 20% in categories V and VI combined in Germany, Luxembourg and Denmark to over 40% in the same grouping in Netherlands, Belgium and the United Kingdom. The percentages falling into the tcp twc categories (combined) also show interesting variations from under 10% in Germany, Italy and the Netherlands to between 21 and 29% in France, Denmark and Luxembourg (which is particularly noticeable for the percentage in the top most category).

24-3

.

TABLE (118): COCUPATIONAL QUALIFICATION - FULL-TIME EMPLOYEES

.

MALE	· ·	• • •	•	· .		••			ħ
OCCUP QUAL	GERMANY	FRANCE	ITALY	NETHS	BELGIUN	LUX	UK	IRELAND	DENMARK
I II II V V V	3.1 6.5 29.0 44.0 13.9 3.4	3.0 18.2 22.4 22.7 18.8 14.9	2.4 6.7 27.0 39.6 18.1 6.2	1.8 5.4 9.9 31.0 33.1 18.7	3.8 10.4 18.3 20.0 25.5 21.9	13.7 15.0 17.3 35.8 13.3 4.9	2.2 8.2 12.4 30.9 29.9 16.4	2.4 8.1 20.7 30.3 34.5 3.9	5.8 19.3 36.4 19.4 13.7 5.4
TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
NUMBERS 000 B	61.4	33.0	35.3	18.8	12.5	:0,2	91.0:	2.6	3.3 &
OCCUP QUAL	GERMANY	FRANCE	ITALÝ	NETHS	BELGIUM	INX	UK	IRELAND	DENMARK
I II IV V VI	0.3 6.0 52.2 31.6 9.8	2.5 11.2 31.1 40.7 14.3	0,3 3,3 34,1 60,9 1,3	0.6 5.4 13.2 32.0 48.7	0.9 5.2 10.9 32.7 50.2	- 17.3 55.5 24.6	- 5.7 55.8 38.2	- 29.4 68.0	6.6 30.2 55.0 7.4
TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
NUMBERS .000 s	60.0	48.6	19.0	6.7	9.1	0.2	74.0	2.6	4.2
MALE AND OCCUP QUAL	FEMALE GERMANY	France	CTALY	NETHS	Belgium	mx	UK	IRELAND	% Denmark
I II III IV V VI	1.6 3.4 17.7 48.1 22.6 6.6	1.3 8.8 15.8 27.7 31.9 14.6	1.6 4.5 18.7 37.7 33.1 4.5	1.4 4.2 8.7 26.4 32.8 26.6	2.3 6.4 12.8 16.2 28.6 33.8	7.4 8.4 10.3 27.3 32.6 13.9	1.2 4.5 7.0 19.6 41.5 26.2	1.2 4.1 11.2 29,9 51.1 2.5	2.5 8.9 19.7 25.5 36.9 6.5
TOTAL NUMBERS	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
000'8	121.4	81,6	54.3	25.5	21.6	0.4	165.0	5.2	7.5

SOURCE TABLE 61 COUNTRY VOLUMES

•

2.44

Indices:

As in previous sections and chapters two set of indices are considering:- one (Table 119) gives comparisons between groups of employees of the same sex; the other (Table 120) gives comparisons between male and female employees in the same occupational qualification group.

The interpretation of these indices is complicated as in previous chapters by the treatment of the cut-off points for the highest earning employees. (cf Technical Annex). These considerations may help to explain why the index value for men in category 1 in France, Italy (and to a lesser extent) Ireland is much higher than elsewhere in that fewer of the highest earners have been excluded from the calculations. At the lower end of the scale it is interesting to note that apart from France and Denmark which are noticeably lower than elsewhere the values taken by the index for category VI are all in the range 59-64% of the overall mean.

For female employees, the values of the indices exhibit similar features to those seen in other sectors and tend to reflect the occupational structure with female employees occupying proportionately fewer of the jobs in the higher categories.

24-5

INDEX OF MEAN GROSS MONTHLY MARNINGS - BY COCUPATIONAL QUALIFICATION GROUP TABLE 119 HALE

erunden an de de stand alle alle de la come	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	UK	IRELAND	DENMARK
I II III V V (*)	181.6 148.8 116.0 88.6 72.4 64.5	299.1 162.7 110.6 73.7 61.5 53.6	285.0 206.2 114.5 82.0 67.8 62.3	217.7 182.7 143.2 108.6 80.9 64.2	200.0 151.8 123,6 98.4 81.8 63.8	175.0 124.9 106.0 79.9 60.2 59.4	191,1# 131.3 99,7 74,8 61,3	232.2# 165.2 125.5 105.9 59.4 63.8#	193.7 131.9 106.1 73.1 65.5 51.6
ALL	100	100	100	0	100	100	100	100	100

TENALE

					بالمتقادين وعمره	1944 - موجد بالماريخ ميزد المرجعي		18 - 19-10 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20 - 20-20-20-20-20-20-20-20-20		-
		OFEMANY	PRANCE	ITALY	NETHER LANDS	BELGIUM	LUXEN- BOURG	UX	IKELAID	DENMARX
and the second se	I II II V V (*)	87.0 143.5 104.9 88.6 78.3	200.0 151.4 102.5 86.5 71.2	254.0 148.8 112.0 89.7 67.5	: 122,5 150,0 102,5 80,0	196.9 143.2 131.5 105.5 82.8	- 141.7 94.1 71.3 -		 126.6 86.8 	- 143.5 107.5 92.7 76.2
	ALL	100	100	100	100	100	100	100	100	100

MALE AND FEMALE

	, .	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DERIMARK		
-	I II III IV V VI (*)	210.1 172.4 132.3 94.7 77.1 67.8	362.7 195.8 132.6 87.6 73.3 62.4	314,9 228,3 126.6 90,3 72,4 67,5	240.9 202.8 147.2 118.5 84.9 63.2	229.0 173.6 137.5 110.6 88.9 68.4	222.0 157.9 135.5 100.3 67.3 54.1	246.8# 168.9 125.7 78.2 63.4	290.6# 206.8 155.1 113.6 67.6 79.8#	244.3 164.5 130.1 87.8 75.3 62.3		
	ALL	100	100	100	100	100	100	100	100	100		
10000	SOURCE	eleva surv e E	174 BI	LE C2	246	COUNTRY	volumes					

Table 120 gives the figures when the effect of occupational qualification group has been removed.

.

TABLE 120

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES - BY OCCUPATIONAL QUALIFICATION GROUP

(MEAN GROSS MONTHLY EARNINGS OF FULL-TIME MALE EXPLOYEES IN SAME CATEGORY = 100)

	CERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM BOURG	UK	IRELAND	DEIBIARK
1 Π Π Π Γ Γ Γ Γ Γ Γ Γ Γ Γ Γ Γ Γ Γ	90.2 88.8 84.9 87.8 87.1 -	83.1 92.5 94.0 95.0 89.7	87.6 92.4 97.0 94.0 77.0 -	• 51.7 83.4 76.5 75.2 -	88.7 79.2 91.4 88.2 88.8	- 96.5 85.1 65.3	- - 81.4# 69.0 69.9 -	- 70,4 86.0 - 58.9	- 85,4 92,8 89,3 93,3

SOURCE

TABLE C2

COUNTRY VOLUMES

There are no entries for category I for any country and for category 11 entries are missing for Netherlands, Luxembourg, the United Kingdom, Ireland and Denmark. For category 111 there are no figures for Luxembourg and the UK and Ireland. In general the ratios are higher than in other sectors. Movements in the ratios with change of category are irregular but with some signs of a decrease in ratio with decreasing job category in Germany, Luxembourg and the United Kingdom and the reverse pattern to category V in France and to category IV in Italy. The Netherlands has a particularly low value in category 111. The figures for Belgium are fairly stable, apart from category 111, and those for Denmark irregular.

5.4.5 Employees classified by size of undertaking

The final set of indices which we shall consider in this section is that where the factor of classification is the size of the undertaking in which the employee is employed. The figures for all employees (full-time plus part-time) were given earlier in Table 107. Table 121 gives the figure for full-time employees alone, and for male and female employees considered separately.

MALE

•

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	1.1 1.8 3.7 9.9 11.6 72.0	2.1 4.2 6.8 13.0 20.3 53.6	16.3 55 7.7 25.3 6.1 39.1	5.2 5.6 11.9 18.7	4.3 7.1 12.5 21.8 27.1 27.3	28.8 71.2 -	2.9 2.0 6.5 11.6 76.0	2.2 5.4 12.4 36.4 33.9 9.7	7.3 6.5 2.9 6.3 22.7 54.4
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

FEMALE

					and the second sec			and the second s	the second s
NU. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	אַוי	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	1.3 2.1 3.8 11.8 11.6 69.4	3,2 5,5 5,1 12,0 18,6 55,7	30.7 6,1 7,9 23,7 4,8 26,7	6,5 6,4 14,1 19,8	8,9 8,4 16,1 24,6 24,0 18,0	28.3 71.7 -	2.6 2.9 6.2 9.4 77.6	3,3 5,8 15,1 39,6 25,6 10,7	6,9 12.8 5.1 4.1 16,5 54,5
ALL	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

					•		WATEN VAL	INCO	
ALL	(00.0	(OI) 0	100.0	100.0	100.0	(30.0	100.0	100.0	100.0
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	1.2 2.0 3.7 10.8 11.6 70.7	2.7 5.0 5.8 (2.4 19.3 54.8	21,4 5,7 7,7 24,8 5,7 34,7	5,6 5,8 12,4 19.0	6.3 7.6 14.0 23.0 25.8 23.4	28.5	2,8 1,2 2,4 6,3 10,6 76,7	2.7 5.6 13.7 38.0 24.8 10.2	7,1 10,0 4,2 5,1 19,2 54,4
NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK

SOURCE

TABLE A2

COUNTRY VOLUMES

249

Since the percentages of part-time employees are small there is very good agreement between the figures in Table 107 and the figures quoted for "men and women" in Table 121. It is of interest to note the slight variations which occur when the figures for male and female employees are considered separately. In most countries the distributions for male and female employees are very similar. Italy is an exception. The concentration of female employees shifts more towards the smallest size of undertaking category whereas male employees tend to be in the largest size category. Ireland and Denmark show similar, but much less pronounced, effects.

Indices

Once again two sets of indices are considered - one giving comparisons between employees of the same sex, the other giving comparisons between male and female employees in the same size of undertaking category. Table 122 gives the comparisons between employees of the same sex.

There is a variety of patterns in evidence in the figures. For male employees there is little variation with size in Germany. In Italy, France and Denmark there is some evidence of higher values in the smaller size categories and in Belgium of the reverse pattern. In the United Kingdom, (Netherlands) and, Luxembourg and Ireland because of some problems with the availability of data, the patterns are not pronounced.

For female employees, Italy, Belgium, France and Ireland, show some signs of higher values of the index for larger size categories. The peak for the United Kingdom and Denmark comes in each case in the 200-499 size category. Germany shows little variation over the range with a slightly higher value in the smallest size category.

Table 123 gives the comparisons between male and female employees in the same size of undertakings.

INDEX OF MEAN OF MEAN GROSS MONTHLY EARNINGS BY SIZE OF UNDERTAKING

1 1

MALE

ì

tarat-r-				. 1				:	
NO. OF EMPS	GERMANY	FRANÇE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49	. 101.0	106.5	110.1	100.6	97-7	.101.0	•	•	101.7
50-99	99.7	102.4	102.2	101.6	99.9	- ·	•	95.5	113.2
100-199	102.8	105.0	98.3	88.9	94.4	99.6	•	95.4	•
200~499	98.9	105.2	95.2	97.8	96.2	-	92.8	112.1	90,4
500-999	101.3	99•7	106.1	-	95.9	-	120.7	91.7	102.0
≽1000	99.8	97.8	98.0	-	110.0	-	97.3	•	98.6
(*)	-	-	-	-	-	-	-	-	-
ALL	100 .	100	100	100	100	100	100	100	100

FENALE

۲ ۵ ۹

きいますとうちのときの書をたいい

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM~ Bourg	UK.	IRELAND	DERMARK
10-49	103.4	96.9	99.0	95.4	90.9	112,8	· 98.5	· 96.7	100.3
50-99	· 97•1	94.5	101.9	95.0	90.1	-	•	93•9	98.3
100~199	101.3	90.6	103.5	92,9	95.6	94.9	102.0	88.1	99.6
200-199	99.8	101.1	94.1	94.7	101.3	-	104.7	102.7	109.1
500-999	99.0	95.2	110.8	-	100.2	-	101.0	103.5	102.6
≥1000	100.1	103.0	103.0	-	110.8	-	99•7	•	98.9
(*)	-	-	-		-	·	-	-	
ALL	100	100	100		100.0	100	100.0	100•0	100

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
10-49	101.0	96.5	100.5	97.7	87.9	105,0	•	95•3	101.7
50-99	97.1	96.0	101.4	99.1	95•3	-	•	94.6	95.8
100-199	102.0	102.0	99.6	88.4	92.8	.98.0	•	90.Z	93•7
200-499	98.0	104.0	95.3	96.8	96.9	-	96.3	106.9	102.1
500-999	100.2	98.3	108.9	-	98.2	-	120.3	97•4	105.8
≥1000	100.2	100.0	101.5	-	113.9	-	97•7	•	98.7
(*)	-		-		-	-	-	-	-
i vi j Times	100	100.0	100	100	100	100	100	100	-73

SOURCE

TABLE D6

COUNTRY VOLUMES

There are no figures presented for the "500-999" and "larger than 1000 employees" categories in Netherlands and in Luxembourg (where in addition there are no figures for the "10-49" and "100-199" employee categories). The figures for the United Kingdom tend to be noticeably lower in most categories than elsewhere particularly in the The ratios for Italy tend to be larger size categories. higher than elsewhere but not so noticeably as in other In France there is an increase in ratios with sectors. increasing size (but the increase is not smooth). In Germany the highest value of the ratio comes in the smallest size category in the remaining countries the peak value tends to come in the "100-199" or "200-499" employees category.

TABLE 123

INDEX OF MEAN GROSS MONTHLY EARNINGS OF FULL-TIME FEMALE EMPLOYEES - BY SIZE OF UNDER-TAXING

(MEAN GROSS MONTHLY EARNINGS OF FULL-THEE MALE EMPLOYEES IN SAME CATEGORY = 100)

NO OF EMPS	GERMANY	FRANCE	ICALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
$10 - 49 \\ 50 - 99 \\ 100 - 199 \\ 200 - 499 \\ 500 - 999 \\ \geqslant 1000 \\ (*)$	73.5 69.9 70.7 72.4 70.1 72.0	61.5 62.3 58.3 64.9 645 71.2	63.9 70.9 74.9 76.3 74.2 74.2	57.3 56.4 63.1 58.5	63.7 61.7 69.3 72.1 71.5 68.9	60.7 51.8 	46.0 * 49.4# 56.51 [*] 41.9 51.4	55.2# 57.9 54.3 53.9 66.4	62.3 54.9 64.3 76.2 63.5 63.3
ALL	71.7	67.6	71.1	60,4	68.4	54.4	50.1	58.9	63,1
SOURCE	·	TABLE	116	COUR	TRY VOLUM	ES	**************************************	a noter retering a sure of	and a second

An interesting feature with these indices is that the values in the individual cells do not, in the majority of cases, differ a great deal from the ratio of the overall means. Where there is variation it is neither consistently higher nor consistently lower than the overall ratio. France is almost an exception to this rule in that the overall ratio is higher than any of the individual categories, other than that for the largest size category. A consideration of the indices for male and female employees separately suggests a possible explanation. Male employees in smaller size of undertakings tend to do relatively better than their not counterparts in larger sizes of undertaking, whereas the reverse is true for female employees.

5.5 Gross Annual Earnings

This section considers the gross annual earnings of full-time employees and follows the same pattern as for previous sectors.

Indices have been calculated for mean gross annual earnings in a manner similar to that for mean gross monthly earnings and are available for all full-time male employees, all full-time female employees and for these groups broken down into occupational qualification groups. Indices are also available to compare the mean gross annual earnings for male and female employees in these categories. The global figures are given in the summary table below.

TABLE 124

SUMMARY OF FEMALE/MALE M									
(Mean Gross Annual Earnin	Mean Gross Annual Earnings of male employees = 100)								
Germany	70.5	(71,7)							
France	63.7	(67.6)							
Italy	70.4	(71.1)	Figures in brackets						
Netherlands	61.0	(60.4)	are the ratios for						
Belgium	67.8	(68.4)	mean gross						
Luxembourg	51.8	(54.4)	monthly earnings.						
United Kingdom	48. 5	(50.1)							
Ire]and	58.2	(58.9)							
Denmark	60.8	(63.1)							

The ratios are very similar to the monthly ones and are in all cases, apart from the Netherlands - where the difference is very small - lower than the monthly ones. The more noticeable differences occur in France, luxembourg and Denmark.

Table 125 gives the various indices of mean gross annual earnings by occupational qualification group and in general these are in good agreement with the monthly figures. For male employees there is some evidence of a narrowing of the differentials of the higher occupational qualification groups over the lower ones in Germany, Belgium, the United Kingdom, Ireland and Denmark. In the remaining countries there is a slight widening in this differential apart from in Luxembourg where there is little change.

INDEX OF MEAN GROOS ANNUAL EARNINGS - BY OCCUPATIONAL GUALIFICATION GROUP TABLE 125

.. ..

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DERMARK
I II III IV V V (*)	175.8 146.8 114.2 88.4 71.9 65.2	296.1 159.4 107.7 69.6 57.1 51.8	297.3 200.9 (15.2 80.5 68.6 64.9	223.0 179.0 136.7 105.5 80.2 64.3	197.4 146.3 121.0 95.6 80.9 65.4	174.0 123.6 101.7 83.7 58,6	152.8# 124.9 96.2 70.3 58.4	224.3# 157.8# 122.6 104.1 58.7 63.6#	(87.7 129.2 104.0 71.9 63.8 55.3

. .

.

4

FEMALE

	GEPMANY	FRANCE	ITALY	NETHER- I-ANDS	Belgium	LUARM- BOURG	UK	IRELAND	DERMANN
Ţ	0	•	•	• •	o			Names -	(mananananananananananananananananananan
11	200,3#	203.1	225.3件	•	194,4				
III	143.5	150.1	152.9	122.7	138.5	•	4	•	143.8
IV	104.2	99.6	un -	140.9	127.9	140.3	150.2	122.9	106.8
. v	87.8	84.0	89.3	101.2	103,9	95.1	101,8	86.4	91,1
VI (*)	77.6	71.9	67,4	78.7	82,4	70.0	85,4		80.8
ALL	100.0	(00.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

MALE AND FEMALE

Ì

	CERMANY	FRANCE	ITALY	NETHER- LANDS	eelgium	LUXEM- BOURG	UK	IRELAND	DENMANK .
I II III IV V V VI (*)	204.7 171.5 131.0 93.8 76.1 67.1	369.7 1965 1305 83.3 69,1 615	327,3 221,9 127.3 88,6 72,2 70,5	243,1 196.6 140:2 113.4 83.4 62.6	225.1 166.9 133.9 106.8 87.2 68.3	221.6 156.7 130.3 102.8 65.3 51.2	231,8# 157,4 118,2 73,1 61,1	277.6# 195.3# 149.6 109,9 66.0 78.8#	239,4 162,9 128,6 85,9 72,4 65.8
	1.00.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	1000
SOURCE	3.	TABL	e c4	256	COUNTRY	VOINES	13 an am 1999 an 1997 a Tha an 1997 an 1	a ta milito i fanta milito sense de la Santa y en es	

For female employees there is a slight narrowing in this differential in Italy, Belgium, the United Kingdom and Ireland and a slight widening elsewhere - apart from Luxembourg where as for male employees, the figures are very similar to the monthly ones.

The figures giving comparisons between the mean gross annual earnings of male and female employees in the same occupational qualification groups are given in Table 126.

The agreement between monthly and annual figures is close, with the annual ratios in most cases tending to be lower than the monthly ones. The most noticeable differences are for category II in Germany, and Italy, and category IV in Luxembourg and the United Kingdom,

TABLE 126 INDEX OF MEAN GROSS ANNUAL EARNINGS OF

FULL-TIME FEMALE EMPLOYEES BY OCCUPATIONAL QUALIFICATION GROUP

· · · · · · · · · · · · · · · · · · ·	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM BOURG	UK	IRELAND	Denmark
I II III IV V VI (*)	96.2# 88.6 83.1 86.1 83.9	81.1 88.7 91.1 93.6 88.3	78,9# 93,4 97,1 91,6 73,1	54.8 81.5 77.0 74.6	90.0 -77.6 90.7 87.0 85.3	86.9 84.0 63.5	- 75.7 70.1 70.8	- 68,6 85,6	84.1 90.4 86.9 88.9
ALL	70.5	63.7	70.4	61.0	67.8	51.8	4815	58.2	60.8

SOURCE TABLE C4 COUNTRY VOLUMES

257

SB

5.6 LENGTH OF SCHEDULED WORKING WEEK

The figures are given in Table 127. As with other sectors is is interesting to note that for both men and women Ireland and the United Kingdom have the highest percentages working the shortest scheduled working week. (It is also interesting to note that the UK also has the only percentage in the "undisclosed" column). There is little difference between the figures for men and women for each country. Germany, France and Luxembourg have the most noticeable concentrations in the 38-40 hours category. Netherlands and Belgium have their largest percentages in the 38-40 hours category but with less concentration and the remainder distributed towards the shorter hours categories. Both Italy and Denmark have concentration in the 36-38 hours category with Denmark having noticeably more in the 36 hours and under category.

TABLE	7271 LE	NGTH OF S	SCHEDULED	WORKING	WEEK: FU	ll-Time	EMPLOYEE	S	. %
HOURS	GERMANY	FRANCE	ITALY	NETHS	BELGIUM	LUX	UK 🚬	IRELAND	DENHARK
<pre>\$ 36 > 36 - 38 > 38 - 40 > 40 - 42 > 42 - 44 > 44 - 46 7 46 - 48 > 48</pre>	99.5 0.2 -	96.9 1.0 1.6 0.4	0.5 82.2 16.7 0.4 0.1	0.5 10.3 87.9 1.3 - -	2.3 33.6 63.5 0.5	98.2 - - -	78.1 4.0	80.1 19.6 	11.0 71.7 15.2 - - -
TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)
UNDISCLOS	SED	· ·					16.5		7
FEMALE HOURS < 36 > 36-38 > 38-40 > 40-42 > 42-44 > 42-44 > 44-46 > 46-48	GERMANY 99.4 0.2	FRANCE 95.2 0.6 4.1	ITALY 0.6 78.5 20.3 	NETHS 1.0 13.4 84.3 1.3 	BELGIUM 4.4 31.2 63.8 0.6	WX - 96.3	UK 89.8 5.0	IRELAND 91.4 8.6	DENHATEK 12.8 73.0 12.1 2.1
>48	-	-	-	-	-				
> 48 TOTAL	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)	(100)

SOURCE TABLE AS COUNTRY VOLUMES -

٠.

5.7 BONUS PAYMENTS

The figures are given in Table 128 and follow patterns for male and female employees similar to those in the Banking sector. Denmark, Ireland and the United Kingdon have high percentages in the "O per cent" bonus category and Italy its highest percentages in the 30 to less than 50 percent bonus range. In Germany for both male and female employees the main concentrations come in the 10-15 and 15-20 per cent bonus ranges. Belgium and Luxembourg have peak figures in the 10-15 per cent range. Netherlands is noticeable for the figure of 38.3 per cent for male employees in the 20 - less than 30 per cent range. In France the two "most popular" ranges are 10-15 per cent and 20-30 per cent though the concentration into these categories is less pronounced than for the "most popular" categories in other countries.

TABLE 128 BONUS PAYMENTS

(FULL-TIME ESPLOYEES)

Ś

% BOHUS	GERMANY	. FRANCE	ITALY	meths.	BELC.	HUX.	UK	TRELAND	DENTIALIC
0 > 05 >> 510 71015 71520 2 2030 ≥ 3050 ≥ 50	1.9 0.9 5.3 36.6 44.0 8.7 2.2 0.4	3.3 6.0 9,6 25.9 21.2 23.2 8.2 2.5	- 21.3 3.6 1.8 1.9 12.3 50.5 8.7	0.8 0.6 11.0 16.9 26.2 38.3 4.5 1.5	3.1 4.0 9.4 44.9 18.0 17:5 2.6 0.4	6.8 5.0 28.1 42.5 13.6 •	67.0 15,0 8,6 3.8 2,2 •	77.6 14.5 3.7 •	65,8 27,2 2,1 • •
(*) FETTALE % BONUS	100.0 Germany	UO O FRANCE	100.0 ITALY	(00.0 Neths,	100.0 Belc.	inx.	100.0 UK	100.0 Inelaîd	(00.0
0 > 05 > 5-10 >/10-15 ?/15-20 ?/20-30 ?/30-50 ?/50 (*)	2.0 0.7 4.2 38.9 47.1 5.9 0.9	4,2 5,6 7,9 26,4 20,1 28,? 5,1 1.8	19.7 3.4 2.6 4.8 8.5 54.7 6.3	1.4 1.3 15.5 22.6 24.9 29.6 4.1 0.5	4.9 5.8 15.5 44.3 19.3 9.5 0.6	6.3 41.4 42.4	79.7 11.8 4.7 1.9	86.5	70,1 26,5 • •
	1 100.0	100.0	100.0	10.0	1000	100.0	1000	100.0	1 (00

-259-

5.8 PART-TIME EMPLOYEES

This chapter concludes with a section on the characteristics and earnings of part-time employees. A summary of the numbers and percentages of part-time employees (already given in Table 106) is given below:

TABLE 129 PART-TIME EMPLOYEES

	NUMBERS (000'S)	AS A PERCENTAGE	OF ALL EMPLOYEES
	ALL PART-TIME	FEMALE	ALL PART-TIME	FEMALE
Germany	15.3	14.6	11.2	10.7
France	1.2	0.9	1.4	1.1
Italy	0.2	0.1	0.4	0.2
Netherlands	3.5	3.0	12.1	10.4
Belgium	·~ 0.5	0.4	2.3	1.8
Luxembourg	•	•	•	•
United Kingdom	19.1	17.0	10.4	9.2
Ireland	-		-	-
Denmark	2.3	2.2	23.5	22.4

These figures have received comment in an earlier section (5.1) but it is worth noting again the variation in the percentages. Denmark is once again conspicuously high. The United Kingdom, Netherlands and Germany form a group with percentages in the range 10-12, while the remaining countries record low percentages, with no entries recorded for Ireland and Italy. These low figures, of interest in themselves, mean that many of the cells in the subsequent tables are empty. This is particularly true in the case of male employees and makes comparisons, across all countries, difficult.

TABLE 130 LENGTH OF SCHEDULED WORKING WEEK

.

.

	MALE				· · · · · · · · · · · · ·	· 			0/
1440	GEDMANN	EPHILE	TTUN	12740	RTALIUM		111	· TA-VAID	10 Tomusou
KKS L	TI 7	T MINUE	CA A	NE (NO	DELUIUM	Lux	u.K.	FRELING	PENMUIRI
·24-74	T11-		0-774	<u>- TT J</u>	· · · ·				
176-70	-								
2020			·····	18.1			•		
121-37	12.4			26.6					
1. 37-24	· · · ·		•	10.8	s				
24-71		5	_	11.4	,			•••••••••	
236			0						
	FRAM	<u> </u>	100.0	100.0			(70.6)		
	1 Ground	5					2		°la :
Hers.	GERMANY	FRANCE	ICALY	NETHS	BELGIUM	LUX	U.K	IRELINS	DENMICK!
524	. 541	71.1	59.7	52.7	82.2	~	59.2	~	74.5
24-22	20.4	7,7	-	3.6	13.0	~	1	~	7.6
26-28	4.9	7.0	•	3:8			¥	~	4.6
(28-30	8.1	,	-	11.0	Q	~	7.0	~	5.9
.30-32	14.4	714	-	27.7	;		ų	1	10.4
132-34	8,0	Ů	-	15.3	-	/	-	<u> </u>	\$ 1
2.34-36	2,2	v	17.9	8,9	<u></u>	~			
7.36	1.7	-	•	-					
	100.0	[00.0	100.0	100.0	100-0		(23,7)	100.0	100.0
	MALE	AND P	EMALE				1		0/
									10
tees.	GERMANY	FRANCE	ITALY	NETHS.	BELGIUM	LUX	И.К.	TRELAND	DENMICK.
<u>\$24</u>	54.9	74.1	64.1	52.0	82.9	-	55,3		74.6
24-26	19.9	5.6		3.4	11.8				7.4
26-28	4.7	5,0	ŧ	3.7			•		4.4
128-30	8.2	4.6	-	11.9	•		6.9		6.1
1-10-32	15,1	7.0		23.4	6				10:8
37-34	7.4	¢	>	14.7	•				•
13426	2.2	5.6	14.7	9.3					¢
7.26	1.6								angenta 1. s. s. s. s. s. s. g.
	100.0	100.0	100.0	100.0	100.0		(30.0		į80.(
		-	•		- • •		(28.5)	1	· · · ·

. .

.

Source TABLE AS COUNTRY VOLUMES 26λ

. . .

.

- - - - - - - -

5.8.1 LENGTH OF SCHEDULED WORKING WEEK

The figures obtained are given in Table 130. Because of problems of small cell size, the distributions in most countries are incomplete particularly in the case of male employees. Where figures are available the concentration of employees is in the less than or equal to 24 hours category - though there are high percentages in the 30 to 32 hours category for both male and female employees in the Netherlands (and the percentages in both cases total to more than 100).

The average ages in years of male and female part-time employees

5.8.2 AGE

are given below:	MALE	FEMALE
Germany	51.5	38.9
France	45.8	44.9
Italy	30.7	37.2
Netherlands	35.2	33.5
Belgium	•	37.6
Luxembourg	-	-
United Kingdom	58.4	42.1
Ireland	-	-
Denmark		37.6

As in the other sectors these figures show considerable variations. Where comparisons are possible these averages are of similar order to those observed in the Banking Sector except in Italy where the averages in this sector tend to be noticeably lower.

The frequency distribution given in Table 131 illustrate the different characteristics. The figures for Netherlands are interesting in that they reveal percentages of part-time workers under 21 years of age both for men and women. France has proportionately the most part-time female employees aged 55 years or older. Germany and Belgium have the most noticeable concentrations in the 30-44 age groups.

TABLE :131	,	age	DISTRIBUTION	-	PART-TIME	EMPLOYEES
------------	---	-----	--------------	---	-----------	-----------

MALE YEARS < 21 21–29 30–34 45–54	GERMANY 17.4 13.7	FRANCE 19.0 41.2	ITALY • •	NETHS 20.4 23.6 19.1 10.8	BELGIUM	- - TTX	UK	TRELAND 	DEINHARK • •
> 55 TOTAL	59 . 9 (100)	25.6 (100)	- (100)	24.6	•		04x ((100)	(100)	(100)
FEMALE YFARS < 21 21-29 30-44 45-54 2.55 TOTAL	GERMANY 0.2 14.5 49.1 25.6 10.7 (100)	FRANCE 11.2 29.1 26.7 31.0 (100)	ITALY 27.9 21.4 38.6 (100)	NETHS 10.8 29.0 30.9 22.9 6.4 (100)	BELGIUM 16.6 50.2 20.1 13.0 (100)	1JJX	UK 11.3 35.9 33.1 18.3 (100)	IRELAND ((00)	DENMARK 24.2 38.0 17.9 17.3 (100)
	. Personal surger				-				

SOURCE TABLE AS COUNTRY VOLUMES

ļ

5.8.3 LENGTH OF SERVICE

The figures obtained are given in Table 132 and though incomplete, particularly in the case of male employees, have some interesting features. What are perhaps most interesting are some of the higher percentages observed in some of the longer length of service categories - in particular 10-19 years for female employees in Italy, Belgium, Denmark and Germany. In most countries the concentration of employees is still in the shortest length of service category though Denmark, Belgium and Germany are exceptions to this rule.

TABLE 132 LENGTH OF SERVICE

MALE

			a local sectors and						10
LENGTH SERVIC	OF E GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- Bourg (UK	IRELAND	DENMARK
< 2 2 - 4 5 - 9 10 - 1 > 20 (*)	27.3 37.2 19 19	27.5 17.4 266 17.9	• • • • • • • • • • • • • • • • • • • •	56.1 19.4 11.3 10.4	••••		•		•
ALL	100-0	100.0	100.0	100.0	• .				100

FEMALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER-	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK
 (YE4PS) < 2. 2 - 4 5 - 9 10 - 19 > 20 	23.6 30.8 22.0 19.4 4.2	39.7 30.6 15.7 8.1	34.9 17.3 31.0	38:2 29.9 22.0 7.5 2:3	27.5 26.1 13.3 33.2		43.0 24.6 18.3 12.0		15.0 28.8 23.2 21.4 8.4 3.3
<u>(*)</u> All	100.0	100.0	100.0	100.0	100.0		1.00.0		100:0

MALE AND FEMALE

LENGTH OF SERVICE	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
<pre> (YEARS) < 2 2 - 4 5 - 9 10 - 19 > 20 (*) </pre>	23.8 31.1 21.9 19.1 4.2	36.3 26.9 18.7 10.8 6.7	36.9 16.2 • 33.3	40.7 28.4 20.6 7.9 2.4	29,1 24,6 14,1 31.0		39.6 22.6 18:2 12.6		16.5 29.7 22.2 20.4 8.0
ALL	100.0	100.0	100.0	(00.0	100.0		100.0		0.001
SOURCE	TA	BLE A4	<u></u>	COUNTRY	VOLUMES		265		

5.8.4 OCCUPATIONAL QUALIFICATION GROUP

The distribution of part-time employees over occupational qualification groups is given in Table 133 and because of the problems of small cell size most of these distributions are incomplete, particularly in the case of male employees. Where percentages are available for female employees, the concentration tends to come in categories V and VI, though in Germany and Denmark substantial percentages are to be found in category IV.

INDICES

The figures are given in Tables 134 and 135.

Apart from part-time female employees in Germany and Denmark there is not a great deal of information available, with figures for the remaining countries (excluding Netherlands and Luxembourg) having entries only in categories V and VI. In Denmark there is not a smooth decrease in index from categories III to VI and less variation is shown there than over the same range of job categories in Germany. Over the categories V and VI Italy shows the least variation and France the most.

5.8.5 SIZE OF UNDERTAKING

The figures are given in Table 136 and where available the concentration tends to come in the two largest size of undertaking categories. The exceptions to this occur for male employees in France and the Netherlands which do record sizeable percentages in the smallest size of undertaking category.

DCCUPATIONAL QUALIFICATION GROUP

TABLE 33

orts		GERMANY	FRANCE	ITALY	NETHER- LANDS	BELCIUM	LUXEN- BOURG	UK	IRELAND	DENMARK
	I II IV VI (*)	 6,1 25.7 59.7	29,6 24.0 • 28,7	63,2	5.9 18.6 24:5 45.4			•		•
	ALL	100.0	100.0	100.0	100.0	•		100.0		(00.0

FEMALE

	orrmany	FRANCE	T TALY	NETHER- LANDS	BELGTUM	LUXEII- BOURG	UK	IRELAND	DENMARK
I II III IV V VI (*)	- 1,4 38,2 27,6 32,8	• • 15,1 77,3	- 36.8 59.4	• 2.7 11.8 32.5 52.6	20.9 66.1		23,9 75,4		- 3.5 25.7 48.4 22.5
ALL	100.0	100.0	100.0	100.0	100.0		100.0		100.0

MALE AND FEMALE

		GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK	IRELAND	DENMARK .
and annual mart	I II III IV V VI (*)	- 1.7 36.8 27.5 34.0	8,2 7,2 5,2 12,4 63,9	12.3 32.3 48.8	1.0 3.1 12.7 31.4 51.6	• 21.0 64.0		- 24,5 69.2		- 3,3 24,4 47,9 24,3
		100.0	100.0	100.0	100.0	100.0		100.0		100.0
	SOURCE)	TABL	ЕĄ	- 267-	COUNTRY	VOLUMES			

%

INDEX OF MEAN GROSS MONTHLY EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 134

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BFLGIUM	LUXES- BOURG	UK	IRELAND	DENMARK
I II III V V (*)	- 188,2#	•			-		•		
ALL	6	1000#							•

FEMALE

- 3: 1990 Child	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEN- BOURG	Uñ	IRELAND	DEEMARK
I II III IV V VI (*)	- 174.8 122.8 103.8 66.3	127.5# 75.6	- • • • • • • • • • • • • • • • • • • •		119.0 82.4		- 113.7 94.1		- 119.1 123.2 103.6 62.2
ALL	100.0	100.0#	100.0		100.0		100.0		100.0

MALE AND FEMALE

Jung and a spin and a s	GERMANY	FRANCE	ITALY	NETHER- LANDS	Belgium	LUXEM- BOURG	UK	IRELAND	DENIJARK
I II III IV V VI (*)	- 181.7 124.4 103.4 65.8	96.0# 36:6	138.4# 88.0 83.8		112.6 79.9		- 109,0# 85.3		122.0 126.2 104.9 60.7
	0.00)	100.0#	100.0		100.0		0.001		100-0
SCURCE	6	TABL	е с 9	248	COUNTRY	r vounæs	•		

INDEX OF MEAN GROSS ANNUAL EARNINGS - BY OCCUPATIONAL QUALIFICATION GROUP TABLE 135 .. · .

.

MALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEN- BOURG	. UK	IRELAND	DEMMARK
	1	•	·. •		· · ·		•	and a state of the	n an
II.		•	•		-		•	•	
III	•	U	-			•	••••		_
IV	•	•	No. 🖌	• •	. •		•		a 🖉 🖬 👘
V	•	•	<u>`</u>				•	•	•
VI (*)	•	•	— .		•				•
ALL	•	100.0#	ø		•				•

FEMALE

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXENI- BOURG	DK	IRELAND	DENMARK
I II III IV V VI (*)	- 176.0 122.3 103.8 64.6	- • • • • • • • • • • • • • • • • • • •	- • • • • • •		- 110.6 82.3#				
ALL	100.0	100.0#	(09.0		100.0		100.0		(00.0

n n New States

MALE AND FEMALE.

	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK -	IRELAND	DENMARK
I II IV V VI (*)	- 183,9 123,4 103,9 64,2	80.0 52.6#	81,2 85.8		- 108,7 82,1#		81,3	•	
ALL	100.0	109.0#	100.0#		100.0		100.0#		100.0
SOURCE	; ;	TABL	8011	2 69	COUNTRY	volumes	• •		•

TABLE 136

SIZE OF UNDERTAKING

MALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- Lands	BELGIUM	LUXEM- Bourg	UK	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 >1000 (*)	4.2 7.8 65.1	36,8 12,7 • 25,1	L • • •	23.2	-		• • • 76,5		-
ALL	100.0	100.0	-	100.0			100.0		100.0

FENALE

NETHER-LUXEM-NO. OF BELGIUM 0... IRELAND DENMARK FRANCE ITALY **GERMANY** BOURG LANDS EKPS 6.9 7.1 0 5.7 10-49 7.7 . 2.2 -8.0 5.9 . 0 16.0 50-99 3.1 6.4 7.6 Φ 12.8 12.7 10,5 4.4 100-199 5.8 7.0 16.4 17.8 21.1 95 13.3 200-499 18.6 9.2 40.3 18.1 12.7 500-999 543 77.5 37.1 17.8 34,5 68.2 ≥1000 (*) 100.0 100.0 (00.0 100.0 100.0 ALL . 100.0 100.0

MALE AND FEMALE

NO. OF EMPS	GERMANY	FRANCE	ITALY	NETHER- LANDS	BELGIUM	LUXEM- BOURG	UK .	IRELAND	DENMARK
10-49 50-99 100-199 200-499 500-999 ≥1000 (*)	2.3 3.1 4.5 9.5 12.6 68.0	15.7 15.1 7.9 (2.1 17.4 31.9	* 12.0 18.4 38.9	8:1 5:3 12:4 16:0	6.4 10.7 7.? 18.2 39.6 17.1		6.9 8.8 77.4		6.7 7.6 6.1 5.5 18.7 55.3
ALL	100.0	(00.0	100.0	100.0	100.0		100.0		100.0
SOURCE		.	TABLE A2		-270 - COUNTRY VOLUMES				

CHAPTER 6. SUMMARY AND CONCLUSIONS

This short concluding chapter brings together the main findings of the report. These findings are restricted to the results on earnings and deal only briefly with the characteristics of the labour force which are discussed in other reports.

The findings are also presented as a series of facts and any interpretation of them is limited. This has been done for two main reasons. One is that it is outwith the scope of the present report. The other is that any such interpretation would be better done after detailed discussions with representatives of the national statistical institutes. The results quoted are as follows:-

1. Indices of Mean Gross Monthly Earnings by:-

8. 50X

b. age

. ł

c. length of service

d. occupational qualification group

e. size of undertaking

These are given primarily for full-time employees. For part-time employees the results presented in the report have been limited to the lower occupational qualification groups. Separate figures have not been quoted for mean gross annual earnings as these results follow the monthly ones very closely.

2. Summary of the graphs of relative earnings, and coefficients of variation.

3. Bonus Payments.

4. Scheduled Working Week

1. Indices of Mean Gross Monthly Earnings - Full-time Employees

a. By Sex

The values taken by ratio of the mean gross monthly earnings for all female employees to that for all male employees may be used to put the countries into broad groupings. In all sectors the values for Italy are high (higher than elsewhere if we exclude Insurance) taking values in the range 71-86% (79-86% if we exclude Insurance)

Another group is Germany, France, Belgium and Denmark where the ratios for all sectors fall in the range 63-73%.

Luxembourg, the Netherlands and Iveland nave ratios for all sectors in the range 56-62%.

The United Kingdom is consistently low for all sectors with ratios in the range 50-55%.

b. By Age

There are many interesting features to be noted in these indices. Amongst the most interesting are the following. 1. Male employees under 21 years of age have mean gross monthly earnings which range from 38% of the overall mean for all male employees in Insurance in Luxembourg to 68% of the overall mean in Retail Distribution in Italy. In most countries and sectors, however, the percentages lie in the range 45-60.

2. Female employees under 21 years of age have mean gross monthly earnings which range from 61% of the overall mean for all female employees in Retail Distribution in Belgium to 85% of the overall mean in Banking in Denmark. In most cases the percentages lie in the range 70-80.

3. In most countries and sectors the peak for the index for male employees comes either in the 30-44 or 45-54 years age group. Banking and Insurance in Italy and Insurance in Luxembourg where the peak comes in the 55 years and over category are the exceptions. The percentages by which these peak values exceed the overall average vary considerably from country to country and sector to sector.

In Wholesale Distribution, peak values fall in the range 8-17% above the overall mean; in Retail Distribution in the range 11-30%; in Banking 31-46% (excluding Germany which is low at + 12%); and in Insurance 21-39% (excluding Germany once z_{5}^{-1} , where the percentage is + 8%).

4. Perhaps the most interesting result from a similar analysis of the figures for female employees is the position of the

Netherlands and Luxembourg where the peak values are noticeably higher than in other countries. The values in Ireland, too, are noticeably high in Banking and Insurance. As was the case for male employees the percentages tend to be higher in Banking and Insurance than in Retail and Wholesale Distribution. The peak values in Wholesale and Retail Distribution in the United Kingdom and Denmark are noticeably lower than elsewhere.

5. When male and female employees in the same age groups are compared the values of the index are higher than the overall ratio, reflecting the different ege distribution patterns for male and female employees. What is particularly noticeable is that the female/male figure for the under 21 years of age group exceeds 100% in several cases (eg Wholesale and Retail in Italy, Banking in France, Belgium and Denmark, and Insurance in Italy, Belgium and Denmark) - and in most other cases is of the order of 80-90%.

c. By Length of Service

For both male and female employees in all sectors the index increases with ego. Perhaps the most striking features are the values taken by the index for the longest serving female employees in the Netherlands and Luxembourg. These values are noticeably higher than elsewhere. In general female employees with longer lengths of scruice do better relative to other female employees than is the case for male employees. When the effect of length of service is removed from comparisons between male and female employees the values taken by the index tend to be higher than the overall ratio.

d. Occupational Qualification Group

The interpretation of these indices has been complicated by problems of classification and the exclusion from the calculations of varying percentages of employees in different countries because their earnings have exceeded certain cut-off values.

The conclusions are therefore restricted to those in lower occupational qualifications groups and in particular the lowest two groupings. Perhaps because of different career patterns the average earnings of male employees in these categories tend to be 'twor relative to all male employees than is the case for female employees in these groups relative to all female employees. The percentages for male employees tend in most cases to be of the order of 70% (60% in Insurance) for the lowest category whereas for female employees figures of the order 75-85% occur most often.

When the effect of occupational qualification is removed from the comparison between overall female and male average earnings, the values of the indices obtained are uniformly higher, being of the order of 80-90% in most cases. The main exception to this is in Wholesale Distribution in the United Kingdom and Luxembourg.

.e. By Size of Undertaking

For both male and female employees the variation in the values taken by the index is less from one category to another than for any of the other factors of classification. In most sectors and countries there is some evidence of an increase in the index with increasing size of undertaking for both male and female employees. The variation is most noticeable in Wholcsale Distribution in Ireland and

Italy for both male and female employees, and for male employees in Retail Distribution in Italy. Banking in Netherlands and Luxembourg also have peaks values which for male employees are of the order of 25-30% above the overall mean-but here these peak correspond to the smaller size categories of undertaking.

When the effect of size of undertaking is removed from the comparison of overall mean earnings for male and female earnings there is little change in the values taken by the individual entries as compared with the overall ratio.

2. Summary of the graphs of relative earnings, and coefficients cf variations

In general, the graphs obtained have been unimodal, and positively skewed for both male and female employees with the earnings of female employees tending to be towards the lower end of the distribution. The coefficients of variation observed have been lower for female employees than for their male counterparts. In particular, those for female employees in Denmark (of the order of 21-28%), and in Banking and Insurance in Germany (23-24%) are noticeably lower than elsewhere. For male employees, the coefficients of variation in Germany (of the order 32-38%) are lower than elsewhere while those in the United Kingdom of the order 61-74% are high, with those for France of a similar order.

3. Bonus Payments

Here the most noticeable feature is the high percentage which appears in the "Oper cent" bonus category in Denmark, the United Kingdom and Ireland for both male and female employees. At the other extreme Italy has high percentages in the "higher percentage" bonus ategories - particularly in Banking and Insurance.
4. Scheduled Working Week

Here the most noticeable features across all sectors for both male and female employees are the percentages appearing in the shortest scheduled working week category in the United Kingdom and Ireland. These are noticeably higher than elsewhere.

CHAPTER 7: TECHNICAL ANNEX

This short chapter consists of extracts from Volume 1 regarding definitions and terms used in the various sectors. A brief description of the problem of cut-off values for the earnings data is also given.

4-1

NACE Activity

61	Wholesale distribution (except dealing in scrap and waste materials)
611	Wholesale dealing in agricultural raw materials, live animals, textile raw materials and semi-finished goods
612	Wholesale dealing in fuels, ores, metals and industrial chemicals
613	Wholesale dealing in timber and in building materials
614	Wholesale dealing in machinery, industrial equipment and vehicles (including ships and aircraft)
615	Wholesale dealing in furniture, household goods, hardware and ironmongery
616	Wholesale dealing in textiles, clothing, footwear and leather goods
617	Wholesale dealing in food, drink and tobacco
617.8	Wholesale dealing in tobacco products
618	Wholesale dealing in pharmaceutical and medical goods, cosmetics and cleaning materials
618.1	Wholesale dealing in pharmaceutical goods
619	Other specialized wholesale distribution and wholesale dealing in a variety of goods (without any particular specialization)
619.1/2	Wholesale dealing in paper, board and stationery and wholesale dealing in books, periodicals and newspapers
64/653- 656	Retail distribution, except retail distribution of motor vehicles and cycles and retail distribution of motor fuels and lubricating oils
641/642	Retail distribution of food, drink and tobacco
643	Dispensing chemists
644	Retail distribution of medical goods, cosmetics and cleaning materials
645	Retail distribution of clothing
646	Retail distribution of footwear and leather goods
647	Retail distribution of furnishing fabrics and other household textiles
648/649	Retail distribution of household equipment, fittings and appliances, hardware and ironmongery
653	Retail distribution of books, newspapers, stationery and office supplies
654/655	Other retail distribution
65 6	Retail distribution of a wide range of goods, with non food goods predominating
812/813	Credit institutions other than central banking authorities
	Establishments whose employees have a status similar to that of civil servants
	Establishments whose employees benefit from the national collective agreement for banks
	Other banking establishments

82 Insurance, except for compulsory social insurance

8.8 Occupational qualifications

Classifications of employees according to qualifications differ considerably between economic activities and between countries, and it was therefore left to the national statistical institutes to classify employees in broad groups on the basis of qualifications which were defined by common agreement for the separate sectors of wholesale and retail distribution, banking and insurance.

Every employee in these sectors was assigned to one of the following qualification groups, depending on the tasks performed, the degree of difficulty involved, the inherent responsibilities and the level of training or experience required for doing the job.

Qualification groups in the sectors of wholesale and retail distribution

For ease of classification, separate qualifications subgroups were defined for shop staff (retail distribution) and warehouse and dispatching personnel (wholesale distribution).

1. Management personnel

- 1A: Top management personnel with *powers of decision* and responsibilities with regard to the general running of the whole undertaking.
- 1B: Management personnel and senior executives directly responsible to the persons described under 1A, with *powers and responsibilities* in respect of *large sub-divisions of the undertaking*, e.g. the principal establishments and their closest collaborators.

This category includes managers of local or regional units with a considerable degree of autonomy in their management function. It also covers employees in *study, research and advisory functions* with a certain degree of independence and power of decision.

In general, the *knowledge* required for such functions will be at *university, college* or at least *higher vocational training* level. This level of knowledge may, by its very nature, have been acquired by experience or by private study. 2. Executives and managerial staff

This category covers employees who are directly responsible to or are given their instructions by the persons described under 1B.

In general, within the limits of their instructions, employees in this category possess wide scope for initiative and responsibility. *Knowledga of at* least *higher secondary school leval* (advanced-level GCE, commercial and technical college diplomas, etc.) is generally required for this type of job. This knowledge may be acquired by general secondary school education, secondary vocational training, experience or private study.

This category includes branch managers, heads of division within a local unit (retail trade) and representatives and sales staff (wholesale trade).

3. Highly qualified junior personnel

This qualification group classifies separately personnel with duties specific to the wholesale and retail trades (sub-group 3A) and other highlyskilled junior personnel (sub-group 3B):

- 3A: This sub-group includes
 - for the retail trade, sales staff who work largely on their own, and give instructions on a limited scale. This category must have had a good specialized training and possess a wide knowledge of the particular trade. This refers particularly to the head salesman/saleslady, chief cashier.
 - for the wholesale trade, warehouse and dispatching personnel who work largely on their own and give instructions on a limited scale, which necessitates a certain amount of specialized training. This refers, for instance, to the working foreman, dispatch manager, assistant storekeeper.
- 3B: Other junior personnel performing specialist functions for which a general vocational training or at least knowledge at advanced primary educational level is required. These employees generally carry out the instructions given by the persons described in category 2 and sometimes category 1B. A certain degree of initiative and responsibility is required for these functions. In some cases, employees in this category may directly supervise the work of personnel in lower categories.

4. Skilled junior personnel

This qualification group classifies separately personnel with duties specific to the wholesale and retail trades (sub-group 4A) and other skilled junior personnel (sub-group 4B):

4A: This sub-group includes:

- for the retail trade, other sales staff (not included in sub-group 3A), whose job mainly consists of serving customers, stores work and looking after the cash.
- for the wholesale trade, warehouse and dispatch personnel (not included in sub-group 3A) whose functions include work demanding accuracy, a certain degree of special training and/or wide practical experience, as well as a certain aptitude for working unsupervised. This refers for instance to the storeman.
- 4B: This sub-group includes other skilled personnel working to precise instructions from their immediate superiors and whose tasks, as a rule, only call for a small degree of initiative and responsibility.

5. Unskilled junior personnel

This group covers all wage-earners performing work of a simple nature, which is normally repetitive and as a rule only requires knowledge of elementary educational level. This category includes, for instance, the packer and the person making up the orders (wholesale trade).

9.2.6 Adjusted gross monthly earnings

In addition to annual earnings, monthly earnings were also recorded, the reference period being the month of October 1974, for the reasons already stated (cf. Section 4). In fact, partly because the survey was conducted on a sample basis, and partly because the data used for the calculation of annual earnings related only to those employees who were paid in full for 1974, it was decided to publish the detailed results in terms of monthly (instead of annual) earnings (cf. Section 11.1). However, with a view to ensuring improved comparability with annual earnings which include bonuses and premiums not paid on a monthly basis, the figures for gross monthly earnings were adjusted in proportion to the estimated rates of bonus (cf. Section 9.2.5), according to the formula

$$\mathbf{x}_{i} = \mathbf{R}\mathbf{m}_{i} \left(1 + \frac{\mathbf{P}\mathbf{r}_{i}}{\mathbf{R}\mathbf{a}_{i} - \mathbf{P}\mathbf{r}_{i}}\right)$$

where Rmi = monthly remuneration

Pri = total bonuses

Rai = annual earnings.

This calculation only included the sample employees paid in full for the month of October 1974, and a distinction was drawn between part-time and full-time employees. In addition, employees whose remuneration exceeded a specific ceiling (cf. footnote to Section 8.14) were excluded from the calculations.

9.3 Quantiles

Because the results of the calculations of earnings were affected by the exclusion of employees whose earnings exceeded the ceiling, particularly in the case of highly qualified staff, various quantiles were also calculated. Thus, on the basis of the frequency distributions of full-time (and fully-paid) employees according to annual earnings and adjusted gross monthly earnings, the first and last decile have been given (Ω (0.10) and Ω (0.90)), together with the first and third quartiles (Ω (0.25) and Ω (0.75)) and the median (Ω (0.50)).⁽¹⁾

 (1) These quantiles provide characteristic statistical averages (median: Q (0.50)) and dispersions (interquartile spread: (0.75) = Q (0.25)

 $Q = \frac{(0.75) - Q (0.25)}{Q (0.50)}$ homologous to the arithmetic mean and

coefficient of variation previously referred to.

Occupational qualification groups in the banking sector

1. Directors, top management

Senior management personnel with powers of decision and responsibility in relation to the general running of the whole undertaking.

2. Senior executives

Managers and senior executives with powers and responsibilities for subdivisions of the business, e.g. departments and main establishments. This category includes managers of local or regional units having a considerable degree of autonomy in their management function.

It also covers employees with study, research and advisory functions having a certain degree of independence and authority to take decisions.

In general, the knowledge required for such functions will be of university, college or at least higher vocational training level. This knowledge may of course have been acquired by experience or private study.

3. Executives (junior management)

This group includes those working directly with the senior executives (group 2 above) who may exercise authority over the personnel of specific sectors.

4. Highly-qualified clerical staff

This group includes highly-qualified clerical staff, who are required to have knowledge of secondary school level, working with members of the category above (group 3) and directly responsible to them and carrying out their instructions. Their duties may require – within the limits of the instructions received – a certain amount of initiative and responsibility. In some cases such staff may coordinate the work of lower categories.

5. Qualified clerical staff

This group includes qualified staff working to precise instructions from their immediate superiors and whose duties as a rule call for only a small amount of initiative and responsibility.

6. Other employees

This group includes all unqualified wage earners not included in the five categories above.

281

I. Managers - Senior management executives

These are the wage earning and salaried staff who because of a very wide degree of delegation of authority, are able to make decisions within the undertaking affecting the undertaking as a whole, or one or more of its main departments.

II. Middle management executives

These are wage earning and salaried staff with wide decision-making powers and a high level of responsibility:

- either at the level of the undertaking, department or section which they organize or supervise,
- or in the planning research or advisory duties, assigned to them.
- III. Junior executives and personnel with equivalent qualifications

These are wage earning and salaried staff who by virtue of working directly or indirectly with one or more of the executives in groups I or II above:

- either supervise, explain or check the work of a group, section or team of employees,
- or are entrusted, by virtue of their ability and high occupational qualifications, with planning tasks involving special responsibilities.

IV. Highly skilled employees (1)

These are the wage earning and salaried staff who perform work requiring proven knowledge of one or more branches of insurance or an equivalent qualification in a field outside insurance and who are able to exercise some initiative in carrying out these duties, within the limits of instructions received. In some cases these duties may include coordination and checking of the work of other employees.

V. Skilled employees (1)

These are the wage earning and salaried staff performing work requiring only good occupational knowledge within the insurance field or outside it. In some cases, these duties may include coordination and checking of the work of other employees.

VI. Other wage-earning employees (1)

These are the wage earning and salaried staff performing tasks requiring only elementary occupational knowledge and only a very short period of initial training.

⁽¹⁾ Non-management workers were assigned to groups IV, V or VI, depending on their qualifications.

CUT-OFF VALUES

Most member countries did not record the exact remuneration of employees whose earnings exceeded a certain ceiling. For monthly earnings the following countries imposed the following ceilings;

DM 5500
FF 20000
L 2,000,000
F1 7300
FB 100,000
Flux 80,000

Denmark imposed a ceiling of 225,000Kr on annual earnings. In the volume for each country there is a preliminary table which gives the percentages in each occupational qualification group exceeding the ceiling. As a percentage of all full-time employées in each sector the figures are very small. However, most of these 'high earners' tend to be concentrated in the top two categories ... and when taken as a percentage of all employees in these categories the percentages are somewhat larger. According to sections 9.2.6 and 9.3 of Volume 1 these employees were excluded from the calculations and it would therefore seem that the mean values obtained for the highest occupational qualification categories will have been affected in different ways in different countries. Some comments on particular problems in the various sectors are given below.

WHOLESALE DISTFIBUTION NACE 61

25 in each category falling above the coiling volue.

MEN					•				
•	CATFOORY	GERMANY	FRANCE	ITALY	HEPHS	BELG IUM	INX	DENMARK	Ϋ́,
	1A 1B		- 5.5 0.4	0.7	9 .] 1.2	<u>18,</u> 4 2,6	23,2	26.9	
MEN AND	WOHEN	•				•••		in the second	
• - •	1A 1B	22.3 14.8	5.2 0.4	0.4	8.9 1.1	18.0 2.5	22.6 6.5	26.7 2.6	
								•	

* ANNUAL FIGURE

According to sections 9.2.6 and 9.4 of Volume 1 these employees were excluded from the calculations and it would therefore seem that the mean values for category 1A in Germany, Belgium and Luxembourg would be on the low side. In this connection, it is also interesting to note the figures from table C7 (Quantiles of Distribution of Full-time Employees, according to Gross Monthly pay corrected). For Wholesale Distribution for Germany, Netherlands, Belgium and Luxembourg the value which appears under the highest decile heading for males in category 1A, and for males and females in category 1A is higher than the ceiling value in each case - see below (figures in national currencies)

•	Category	Hig Male	hest Decile Male and Female	Ceiling
GERMANY	AL	5968	5922	5500
NETHERLANDS	٦A	7582	7541	7300
BEIGIUM	14	106211	1059147	100000
LUXEMBOURG	1A	84250	831.25	00003

The figures for categor, 1B for Germany are also interesting in particular the value for females

	High	est De	cile
Category	M ·	F	T
1B	5527	6221	5531

The fact that these ceiling values are exceeded in this way seems unusual. RETAIL DISTRIBUTION NACE 64/653-656

										•
	 						-		-	

GERMANY

e ¶ ₃	in	each	category	falling	above	the	ceiling va	lue
1.							and the second	C.C. statistics

MEN	· · · ·	· · · · · · · ·	· · · · ·	. . <i></i> -				
	CATEGORY	GERMANY	FRANCE	ITALY	NETHS	BELG.	LUX	DENMARK*
• `	1A 1 B	17.9 15.6	3•3 0•2	•	5.6	24.4	15.7 1.3	18.7 0.9
MEN	AND WOMEN		a		:	·	•	· · ·
	1A- 1B	15 .1 13.8	3.2 0.1	•	5.1 1.8	23.7 1.8	13.6 2.1	18.9 0.5
CEI	LING VALUE	5500	20,00Ċ	2,000,000	7300	100,000	80,000	225,000
(na Cur	TIONAL RENCIES)	*The fi	gure for	Denmark is	for a	mual pay	•	•

From Table C7 (Quantiles of Distribution of Full-Time Employees according to Gross Monthly pay corrected) for the volume for Belgium we also have that the highest decile values for men (104682), and for men and women (104364) are above the ceiling value.

Some of these considerations may help to explain the situations in Germany and Denmark for males where the index figure for category 1A is lower than that for 1B. A similar situation occurs for females in these categories in Germany and Eelgium.

BANKING 812/813

(NATIONAL CURRENCIES)

% in each category falling above the cailing value

MEN	,	·	•••		••••		
CATEGORY	GIERMANT-	FRANCE	ITALY	NETHS.	BELG.	LUX.	DIMMARY
1 2 2	25.5 19.2	2.9 0.1	4.8 0.3	15 .2 3 .8	50.0 11,5	47.2 20.4	53.2 1.4
KEN AND HOLEN OLEVELORY		•					
1	25.5 10.7	2.8 0.1	4.6 0.3	14.7 3.7	42.7 11,5	47.2 20.1	53.2 1,4
CTITIC VALUE	5.500	20.000 2.	000.000	7,300	100,000	80,000	225,000

*The figure for Denmark is for annual pay.

It is also interesting to note from Table C7 (Quantiles of distribution of full-time employees according to gross monthly pay (corrected), by activity, Fex and qualification) that for category 1 for men, and for men and wemen in Cormeny, Balgium and Innembourg the upper quantile value and the highest decide value are both above the ceiling value in each case. This also applies to the median value for category 1 men in inxembourg (median value = 61250 F lux), and the highest decide for category 2 men (82750) there also.

Some of the above might help to explain why the figures for France and Italy for category 1 males are much higher than elsewhere (fewer of the top carmers have been excluded from the calculations?) (On the other hand Cermany is lower than might have been expected(?) even allowing for the problems of cut-off values/.

-285

CATEGORY	GERMANY	FRANCE	TTALY	NETHS	BELG.	- IUX	DENMARK*
I	23.2 7.6	5.9 1.7	5.0 1.8	23.0 1,9	22.4 2.0	13.8 5.2	23.0
				٦.	•		
EN AND NOMEN	•				·		
CATEGORY	· · · ·		. •		· · · · ·		• **
I II	22.7	5.6 1.5	4.4 1.9	22.6 1.9	22, 2 1.9	13.8 5.1	23.0
EILING ALDE	5,500	20,000	2,000,000	7,300	100,010	80,000	225,000
NATIONAL TIPRENCIES)		×	The figure	for Der	mark is for	· annual p	87

lull-time employees accordin pay (corrected), by activity, sex and qualification) that in Germany, Netherlands, Belgium, Luxembourg (and in Denmark if the monthly figure is converted to an annual one) that the highest decile value for nen (and for men and women) in category 1 exceeds the ceiling value. In Germany the upper quartile value /for man (and for man and women) in category 1 also exceeds the ceiling value.

Highest Decile Values (National Currencies)

	MEN	MEN AND WOMEN
GERMANY	6,346	6,326
NETHERLANDS	8,030	8,003
BELGIUM	100,775	. 100,525
LUXEMBOURG	86,000	86,000

Dpper quartile value in Garmany 5,613 (Mon); 5,583 (Men and Women)7

Some of the above considerations may help to explain why the index value for mon in category 1 in France and Italy are much higher than elsewhere (fewer of the highest carners excluded from the calculations?).

· · · · .: DE THE SAME AGE AND SEX - PROFESSIONAL FIGURE 5

	nu dum	SLUTE.	
		%	BELGIQUE/BELGIË
		· •	 Fighter participation and press and statements
Forces de traveller rapport à la pepulation	'n		a philippe philippe and a statistical statistical statistics of the second
totale de monar agre et sexe		-6)	
,		. N	
Forze di lavous la capporto alla populazione	3	60 .	
totale della sistema elle e susso	-	. 50	
	••••	40	
Asodool upp the subministrachten in do total	e hevelling		ta a construction of the second s
van ne like he Diri an notaele	o beronang) 30	
, And Replan to real and Populatif		20	n - mara an an
•	•	10	an a
		0	un an
R.B. DEHTSCHLAND			LUXEMBOLIEG
	· · · · · · · · · · · · · · · · · · ·	103	
الارجميس إدام أأوحفو معهم وبربد المروجونة وبالرابر بأباد	المستحدية وتعاولا	···· ·· 90	المحتجان المراجب السابيك الربيان فالمتهونية وروماتهم أتناف ومؤرجة ومنافر والمراجع
دي. هوما انداري ماري محمد الارتسانية الملك منها منها منها مراجع المراجع المراجع المراجع المراجع المراجع المراجع		80	ին է է է է է է է է է է է է է է է է է է է
· · · · · · · · · · · · · · · · · · ·	1		
4 € 4 € 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			
·····	•• •• • • • • •••	{0	المستقدية الاستهمة بالتي كالاستاذ منصلة أسافته السوار بالم أرادة ومستقد السبار السبار. المستقد والا
		40	n an anna an an an Anna
and the second		30	
		20	· · · · · · · · · · · · · · · · · · ·
	÷ • .		
•	******		
n i i i i i i i i i i i i i i i i i i i		0	
FRANCE			UNITED EULGDOM
y gaterie v name. 	• • • • • • • • • •	100	
والأبيان المستعمل والمناج وأجرب والمستجد والمراجع والمتاجي والمناجي			a a a a a a a a a a a a a a a a a a a

And the second s •• 80 • 70 63 • . • • • • • • • 50

ITALIA IRELAND الالالا مانان معاملين منظنات من الأربان من معامل من من من من معامل معامل الالاري ال

NEGERLAND

. 27 70 ••' segurine and a standard a 50 . .. <u>.</u>. : 40 · ••

50 រាំ រំា 50 ÷. · · · . . .

1.101 1 . .

· · · · · · · · · · · ·

. .

..... an an an an raise à la straite de la straite,**.** · ° 4 684 84 8468 88 6 • C ·· · · ·· ••••• ** 20 * #^{*} 2729 8 6 0 0 0 0

---- 160

37

8

10

DANMARK ·..

់ត

A30.1 6.... 10. Lard of

A-10

· .i